# Review of the Delivery of Restorative Justice in Family Violence Cases by Providers funded by the Ministry of Justice

by

### Venezia Kingi, Judy Paulin and Laurie Porima

Crime and Justice Research Centre Victoria University of Wellington

May 2008



New Zealand Government

# Contents

		gements	
Execu	ıtive s	summary	i
	Metho	odology	
	Natior	nal survey of restorative justice providers	i
		rogrammes at five sites	
,	Views	of victims and offenders	iv
		of key informants	
		nary and discussion	V
Chapt	er 1	Introduction	11
	1.1	Background	
	1.2	Definition of family violence	
	1.3	The delivery of restorative justice in family violence cases	
	1.4	The review	
	1.5	The structure of this report	
Chapt	er 2	Methodology	
	2.1	Objectives	13
	2.2	Ethical issues	
		Safety procedures	13
		Cultural safety	14
		The effect of the research on those being studied	14
	2.3	Research methodology	14
	2.4	Data sources	
	2.5	Interview schedules and information sheets	
	2.6	Sample selection	15
		The five sites	15
		National survey of restorative justice providers	
		Volume of family violence cases	16
		Victim and offender interviews	
		Key informant interviews	21
		Observations	21
	2.7	Data Analysis	22
	2.8	Limitations of the research	
Chapt	er 3	National survey of restorative justice providers	23
	3.1	Introduction	23
	3.2	Survey findings	23
		Percentage of total referrals involving family violence offences	24
		Referral sources and stages	25
		Consent processes	26
		Suitability of referrals	26
		Participants' safety	27
		Facilitators' qualifications, training and experience	28
		Cultural issues	29
		Recommendations or plans	29
		Monitoring compliance	30
		Feedback from participants	30
		Other comments	30
	3.3	Summary	30
Chapt	er 4	The programmes at five sites	32

4.1	Introduction	32
4.2	Nga Whare Waatea	32
	Demographic characteristics of the Manukau population	32
	Description of the Crossing Bridges programme	33
4.3	Horowhenua Restorative Justice Programme	36
	Demographic characteristics of the Levin population	36
	Description of the Horowhenua Restorative Justice programme	36
4.4	Nelson Restorative Justice Services	39
	Demographic characteristics of the Nelson population	39
	Description of the Just Right programme	39
4.5	Christchurch Restorative Justice Services	42
	Demographic characteristics of the Christchurch population	42
	Description of the Restorative Justice Services programme	42
4.6	Project Turnaround, Timaru	45
	Demographic characteristics of the Timaru population	45
	Description of the Project Turnaround programme	46
4.7	Summary	49
Chapter 5	The views of victims and offenders	52
5.1	Introduction	52
5.2	Choosing to have a restorative justice meeting	52
5.3	Preparation	55
5.4	The meeting	57
	First impressions	57
	Participation	58
	Impressions at the end of the meeting	60
	Good and bad features of the restorative justice meetings	61
	Meeting cultural needs through processes	62
5.5	The agreement	63
	Reaching agreements	63
	Good and bad features of the agreements	65
5.6	Sentencing	65
5.7	After the meeting	
	Satisfaction with the restorative justice process	66
	Taking responsibility for the offending	67
	Appropriateness of restorative justice for family violence cases	68
	Offenders' behaviour subsequently	69
	Nature of relationship between victims and offenders subsequently	70
5.8	Views of matched victims and offenders interviewed	70
5.9	Summary	72
Chapter 6	The views of key informants	74
6.1	Introduction	74
6.2	Key informants' perspectives	74
	Views in general	74
	Offences/offenders for which it is particularly appropriate	75
	Offences/offenders for which it is particularly inappropriate	76
	Their ratings of local programmes' processes	76
	Their ratings of local programmes' agreements	79
	Effectiveness of the local programme	79
	Effectiveness in meeting the needs of Judges, victims, offenders & the community.	81
	Effectiveness in meeting the needs of different cultural groups	82

	Improvements in programme processes	83
	Quality of relationships	
	Impact on agency/service	
	Final comments	
6.3	Summary	
Chapter 7	Summary and discussion	
7.1	Meeting the objectives of the review	
	Selection of cases	
	Assessment of appropriateness of each referral	
	Consent of participants	
	Detail of the restorative justice processes	88
	Safety of participants	
	Agreements reached at meetings	
	Experiences of participants	
7.2	In conclusion	
References	s 94	
Appendix A	ASupplementary tables	96
	BInformation sheets, consent forms, and interview schedules	
B1	Information sheets and consent forms	
B1	Ministry of Justice survey	
В3	Interview schedules	
	Questionnaire: Victim attending conference	
	Questionnaire: Victim not attending conference	
	Questionnaire: Offender	
B4	Areas to discuss in family violence cases - Restorative Justice Services Christchurch	
	·	
Tables		
Table 2.1	Demographic characteristics of victims and offenders interviewed: numbers	18
Table 2.2	Nature of the relationship between victim and offender: numbers	19
Table 2.3	Nature of relationship between offender and victim: numbers	20
Table 2.4	Time since restorative justice meeting and interviews with victms' and offenders: number	ers20
Table 2.5	Roles of key informants interviewed: numbers	21
Table 3.1	Percentage of total cases referred to providers (n=21) involving family violence offence	s 24
Table 3.2	Types of family violence offending referred to providers (n=21) during the previous 12 months	24
Table 3.3	Percentage of family violence referrals to providers (n=21) which resulted in a conferent panel meeting	
Table 5.1	Preparation of victms (n=19) and offenders (n=18) for the restorative justice meeting: numbers saying 'yes'	55
Table 5.2	Victms' (n=19) and offenders (n=19) views of restorative justice meeting process: numbers aying 'yes'	
Table 5.3	Victims' (n=19) and offenders' (n=19) views of restorative justice meeting outcomes: numbers	64
Table 5.4	Victms' (n=19) and offenders' (n=19) views of restorative justice process: numbers sayi 'yes'	
Table 5.5	Matched victms' and offenders' views (n=11) of restorative justice process: numbers	
Table 6.1	Key informants' (N=24) ratings on programme effectiveness in family violence cases; numbers	80
Table 6.2	Key informants' (N=24) ratings on programme effectiveness in family violence cases in meeting needs of stakeholders; numbers	81

Supplementary tables
----------------------

Supplemer	ntary tables	
Table A2.1:	Some characteristics of restorative justice meetings involving family violence cases, and estimates of number of such meetings	9

# Acknowledgements

The Crime and Justice Research Centre at Victoria University of Wellington was commissioned by the Ministry of Justice in 2006 to undertake a review of the delivery of restorative justice in family violence cases by providers funded by the Ministry. The members of the review team and their roles were as follows: Dr Venezia Kingi, was the project manager and was involved primarily in interviewing victims and offenders. Judy Paulin was involved primarily in interviewing programme providers and key informants. Laurie Porima was the Māori cultural consultant and provided cultural advice to the team and undertook interviews with key informants and offenders.

Our thanks go to our colleagues, Dr Elaine Mossman and Dr Lesley MacGibbon, at the Crime and Justice Research Centre for their help and support in reading various drafts of the report and providing insightful feedback.

We would like to acknowledge and thank all those who were involved in this review. We would like to thank the members of the Ministry of Justice Review Advisory Group for their support and advice. We would also like to thank the restorative justice providers at the five sites who were generous with their time and information sharing and who facilitated our observations of meetings and contact with participants. Their support was crucial to this review. In particular our thanks to: Mike Hinton of Nga Whare Waatea Trust, Eric Rooke of the former Horowhenua Restorative Justice Programme, Mark Rutledge of Nelson Restorative Justice Services, Jennifer Cato of Restorative Justice Services in Christchurch and Linda Gaskin of Project Turnaround in Timaru. Our thanks also to key informants who made time in their busy working days to talk to us.

Finally, we would like to express our appreciation to the victims and offenders who agreed to take part in this research. Without their willingness to share details of traumatic events in their lives with us, this research would not have been possible. We hope we have been able to give them a voice.

Published by the Ministry of Justice PO Box 180 Wellington New Zealand

© Crown Copyright

ISBN 978-0-478-29054-3

The views, opinions, findings and conclusions or recommendations expressed in this publication are strictly those of the author/s. They do not necessarily reflect the views of the Ministry of Justice. The Ministry of Justice takes no responsibility for any errors or omissions in, or for the correctness of, the information contained in the publication.

# **Executive summary**

As described in Chapter 1, in June 2006, the Ministry of Justice contracted researchers from the Crime and Justice Research Centre, Victoria University, to undertake a review of the delivery of restorative justice in family violence cases.

The objectives of the review were to describe the nature and extent of the delivery of restorative justice in family violence cases across the restorative justice programmes operating at five sites including the following:

- how selection of cases occurs;
- assessment of appropriateness of each referral;
- how the consent of participants is obtained;
- detail of the restorative justice processes;
- safety of participants;
- the outcomes for individuals (both victim and offender) from the processes e.g. participation in therapeutic programmes, sentencing outcomes, etc); and

#### Methodology

A mixed methods approach was utilised for the review (see Chapter 2) that included analysing a national survey of restorative justice providers, reviewing and analysing reports from providers, interviewing key informants and victims and offenders, observing restorative justice processes, and collating information on referrals. Qualitative and quantitative data were triangulated to assess the appropriateness of the use of restorative justice processes in family violence cases.

#### National survey of restorative justice providers

Chapter 3 presents an analysis of a national survey of restorative justice providers carried out by the Ministry of Justice in 2006. Twenty-four of the 30 providers surveyed responded about their use and practice of restorative justice processes in family violence cases. The main findings were:

- Twenty one (or 88%) of these providers accepted referrals for restorative justice cases in family violence cases. Fifty-two percent used a victim-offender process, 29% used a community panel process, and 10% used a combination of victim-offender and community panel processes.
- Family violence did not make up the majority of their caseloads. Approximately one half estimated that 1-10 percent of the cases referred to them involved family violence and one quarter estimated that 11-25 percent of the cases referred to them involved family violence.
- Providers who had accepted restorative justice cases in family violence cases had received referrals which included sexual assaults/violation (including against children), physical assaults, threats, and wilful damage.
- Most commonly the Court (n=17), defence lawyers (n=15), and the Police (n=9) referred family violence cases to restorative justice providers.

- Most restorative justice providers (86%) required victim consent before a case went to a conference or panel.
- Providers thought that particular care needed to be taken with the screening of referrals, preconference preparation, and conference facilitation in order for a restorative justice meeting to happen safely in family violence cases.
- Providers said they worked appropriately cross culturally through recruitment of facilitators, community panel members, support people and Board members from a range of ethnic backgrounds, including from tangata whenua, Pacific peoples and from peoples of other ethnic groups.
- Most providers (81%) indicated that their conferences or plans usually recommended that offenders be referred to other programmes or services.
- Most commonly the coordinator took responsibility for monitoring offenders' compliance with their plans, including attendance at programmes or services and payment of agreed reparation to the victim.
- Two-thirds of providers who accepted referrals for family violence cases sought feedback from conference participants as to their satisfaction with processes.

#### The programmes at five sites

The restorative justice programmes at the five sites selected for the review are described in Chapter 4.

The programme providers at the five sites chosen for the review were: Nga Whare Waatea Trust in South Auckland, the Horowhenua Restorative Justice Programme in Levin, Nelson Restorative Justice Services located in Nelson, Restorative Justice Services in Christchurch and Project Turnaround in Timaru.

The following is a brief overview of the number of meetings held at each of the five sites during 2005/2006 and the stage at which the restorative justice process took place

- Nga Whare Waatea: six meetings; five pre-sentence and one post-sentence to inform a
  parole board hearing;
- Horowhenua Restorative Justice Programme: 15 meetings; all pre-sentence;
- Nelson Restorative Justice Services: 14 meetings; all pre-sentence;
- Restorative Justice Services Christchurch: 15 meetings; 12 pre-sentence, three
  community referrals two for historic sexual abuse (no charges laid) and one to arrange the
  division of property after a relationship break-up; and
- Project Turnaround: 19 meetings; all pre-trial community panel diversion.

The processes followed at the five sites are briefly described below.

At two of the sites, Levin and Nelson, the restorative justice process used is a community
panel pre-sentence model which involves two panel members and a facilitator. At two other
sites, Nga Whare Waatea and Christchurch, the restorative justice process is a victimoffender meeting with a facilitator and a co-facilitator. At Nga Whare Waatea the process
takes place either pre or post-sentencing. In the case of Christchurch the restorative justice

process can take place pre or post sentencing or where there is no formal charge. At the fifth site, Timaru, the process is one of community panel diversion and involves the Police, panel members and a facilitator.

- Referrals come to programmes from a variety of sources. Referral sources include lawyers, Judges, Community Probation Officers and Victim Advisers. Christchurch accepts referrals from the community.
- All five programmes approach family violence referrals with caution and assess referrals on a case-by-case basis. Nelson and Horowhenua will not accept referrals where there has been a breach of a protection order.
- All programmes require the victim's consent before the restorative justice process can go
  ahead and encourage the participation of victims. Horowhenua and Nelson require the
  victim to be present at the meeting.
- All programmes address cultural issues by endeavouring to match the ethnicity of facilitator
  and or panel members with the ethnicity of participants and referring participants to culturally
  appropriate violence prevention programmes where possible.
- Facilitators consider safety at referral time, where they assess referrals on a case by case basis. Christchurch has a specific protocol setting out the areas for the facilitators to discuss with family violence victims and offenders in preparation for the meeting.
- All programmes encourage the victim and offender to bring support to the meeting.
   However, Nelson requires both victim and offender to have support at the meeting for the restorative justice process to go ahead.
- All five programmes advise victims and offenders of the rules for the meeting and what
  would happen if these were breached. At Christchurch, participants are required to sign an
  agreement which sets out the rules for the meeting. In all cases the meeting would be
  stopped if anyone felt unsafe and did not want to carry on with the process.
- During 2005/2006 all five programmes used restorative justice processes for the offences of
  male assaults female and assault/common assault. Other offences for which restorative
  justice processes were used at specific sites included: arson, grievous bodily harm; assault
  with intent to injure; assault on a child; an historical sexual offence, conspiracy to murder,
  assault with a weapon and dangerous driving causing injury.
- The restorative justice meetings held for family violence cases in Horowhenua, Nelson and Timaru resulted in agreements which offenders are required to complete before the disposition of their case.
- At Nga Whare Waatea agreements take the form of suggestions of what the offender needs
  to do to 'heal the harm' within the whānau or family. At Christchurch agreements reached in
  family violence cases take the form of suggestions or ideas as to what 'should happen' for
  the best outcomes. In both of these sites, if cases have been referred from the Court and
  take place pre-sentencing, the outcomes or agreements are included in a report to the
  sentencing Judge.

- Agreements arrived at with which offenders had to comply, commonly included attendance
  at a violence intervention programme and sometimes community work or donations.
   Monitoring of tasks was predominantly carried out by the restorative justice facilitator.
- Where recommendations for programmes are taken up by the sentencing Judge the
  monitoring of programme completion takes part through the Court process (ie either through
  delaying sentencing until the offender has complied with requirements or making programme
  attendance part of the sentence to be monitored by the Department of Corrections.)
- Sentences for offenders in family violence cases where restorative justice process had been used commonly included discharge without conviction or charges being withdrawn by Police Prosecutions for those cases referred as part of the community panel diversion programme.

#### Views of victims and offenders

A small number of victims and offenders who had participated in a restorative justice meeting for family violence at one of the five sites were interviewed. These interviews are described in Chapter 5. The main findings were

- Most victims and offenders said that they took part in restorative justice meetings voluntarily
  for a variety of reasons which included to: sort out the issues that had led to the offending;
  be able to discuss their differences in a safe environment; and to get closure if they were no
  longer in a relationship
- In particular, victims wanted the offender to get help to change their behaviour and offenders wanted to take responsibility for their actions and make amends to the victim.
- Most victims and offenders found the restorative justice meeting was a positive experience.
   They generally understood what was happening and felt involved; felt that they had been treated with respect; had had the opportunity to express their views; and were not too scared to say what they wanted to.
- Where participants had support at the meeting (50% of offenders and 37% of victims) they found this useful.
- Few victims or offenders had anything bad to say about the meeting.
- The good features of the restorative justice meeting most commonly reported by victims were: the open dialogue and the healing process and being able to meet the offender in a safe supportive environment.
- The good features of the meeting most often referred to by offenders included: being able to put things right, and feeling supported and being treated with respect.
- A few victims (21%) and offenders (16%) felt that it was important that the restorative justice
  meeting meet their cultural needs. All bar one of this group reported that this had
  happened.
- Agreements reached at meetings usually included some form or counselling or programme attendance and sometimes community work or a fun social activity. Most participants said that they were satisfied with the agreement reached. However, offenders were more likely to report this than victims (93% compared with 79%).

- Offenders (68%) were more likely than victims (32%) to say that they had understood how the victim felt about the offending
- Most victims (63%) felt that the offender had been made accountable for their actions.
   However, approximately one half of victims felt that the offender had not been able to 'repair the harm' or make up for what they had done and a few felt that the offender had got off too lightly.
- Most victims and offenders were pleased that they had taken part in the restorative justice meeting. All offenders and most victims (84%) said that they would recommend restorative justice processes to other victims and offenders to deal with family violence. Almost four fifths (79%) of victims said that they would take part in the process again should they need to.
- Victims were slightly more likely to think that offenders still needed help to address their violent behaviour, than offenders themselves. Most (84%) offenders said that they had got help or were involved in the process of getting it
- Only two victims said they felt safer as a result of taking part in the restorative justice
  meeting. Most victims said that this was not an issue for them: they had either ended the
  relationship or they had never felt unsafe.
- Victims' views were mixed on whether or not the violence or abusive behaviour had stopped since the meeting. One third said it had whereas another third said that the violence had just changed in nature: from physical to psychological violence.
- More than one half of offenders said that they had learnt something about their behaviour and its effects on others and that they needed to change.
- Victims and offenders generally agreed that restorative justice meetings were a good way to deal with family violence, although some qualified this by saying it depended on those involved and the circumstances relating to the violence.
- Offenders were slightly more likely than victims to think that their relationship with the other party had improved since the restorative justice meeting (63% compared with 53%).

#### Views of key informants

A number of key informants from stakeholder groups including violence intervention programme providers, Victim Advisers, Judges, Police and lawyers across the five sites were interviewed. Their views are presented in Chapter 6. The main findings were:

- Key informants were divided in their views as to the appropriateness of using restorative justice processes with victims and offenders in family violence cases: 33% expressed unconditional support; 38% expressed conditional support, and 29% were against the use of restorative justice for family violence cases
- Key informants felt that the use of restorative justice for family violence cases was appropriate
  for: less serious offences; offenders who showed remorse; offenders who accepted
  responsibility for their offending; and victims and offenders in ongoing relationships.

- Key informants felt that the use of restorative justice for family violence cases was inappropriate
  for: rape and other sexual offences; serious assaults; and cases where victim was coerced
  into taking part by offender.
- More than one third (38%) of key informants considered that they did not know enough about the restorative justice processes used by their local programme when dealing with family violence cases to be able to comment on their performance.
- Key informants who were familiar with their local programme gave them the following ratings for their work with family violence cases: 46% working well or very well; 4% neither well nor poorly; and 12% poorly or very poorly
- Opponents of the use of restorative justice processes in family violence cases continually referred to the futility of a one-off meeting.
- Supporters of the use of restorative justice processes in family violence cases felt that these
  could provide some victims and offenders with an opportunity to come together in a less
  pressured environment than the Court, allow for victims to have a voice and for offenders to
  start to address their offending.

#### Summary and discussion

The objectives of the review were to describe the nature and extent of the delivery of restorative justice in family violence cases. These are discussed in turn in Chapter 7.

#### Selection of cases

The selection of family violence cases for restorative justice processes occurs at several levels. At the referral level, Judges, lawyers, Police, and some other criminal justice representatives, may suggest to the Judge that the offender might be a suitable candidate for restorative justice. At the restorative justice provider level, it is usually the coordinator who decides whether to proceed with the use of restorative processes by typically considering the types of offences the provider accepts, the suitability of the victim and the offender to participate and the facilitator's ability to hold a meeting in a supportive and safe environment.

Eighty-three percent of offenders reported that their first contact about restorative justice was from the restorative justice provider. Court Victim Advisers were usually the first to contact victims about restorative justice. Fifty-eight percent of victims reported this.

Almost three-quarters (71%) of key informants interviewed were in favour of the use of restorative justice processes in family violence cases, although over half of this number added caveats to their support such as the importance of facilitators skilled in the dynamics of family violence relationships and the need for clear safety protocols/guidelines.

#### Assessment of appropriateness of each referral

The coordinator/facilitator assesses, on a case by case basis, the appropriateness of the offender and the victim to participate in restorative justice processes. Restorative justice meetings were more likely to be held for family violence cases in circumstances where the victim wanted to continue in the relationship with the offender.

Victims and offenders said they had decided to take part in the restorative justice process because they wanted to sort out the issues that had led to the offending and they wanted to be able to discuss their differences in a safe environment; to get help for the offender and those who were no longer in the relationship, wanted closure.

#### **Consent of participants**

The offender's verbal consent to participate in restorative justice processes is usually obtained at a pre-meeting, as is that of the victim once the provider has explained their processes. Written consents are not usually obtained by coordinators/facilitators at the five sites.

Most victims and offenders said that they were comfortable being contacted about their potential participation in the restorative justice process.

#### Detail of the restorative justice processes

The restorative justice processes used by providers for family violence cases did not appear to differ significantly from the processes they would usually use to deal with other types of cases.

At some sites, such as Nelson, the victim must agree to be present at the meeting whereas at others, such as Timaru, this is not a requirement.

It was a common theme for offenders to offer an apology at the meeting, although they were not formally required to do so

Most victims and offenders had expectations about what would come out of the restorative justice process. These included: for the offender to realise the impact of what he had done and be made accountable and to help both parties so that this would not happen again.

Victims and offenders generally understood what was happening and felt involved, they felt they had been treated with respect and had had the opportunity to say what they wanted. Offenders had been given the opportunity to explain their actions and victims had been able to make clear the effects of the offending on them. Victims and offenders had been able to meet in a safe, non judgemental environment

In general, levels of satisfaction with the process and agreements reached were high, although offenders were consistently more likely than victims to report this. A few victims and offenders said that it was important that the restorative justice process meet their cultural needs. Only one offender said his/her needs had not been met.

Key informants also generally felt that the local restorative justice programme was effective: in involving the victim; in holding the offender accountable and in addressing the needs of the offender in relation to his/her violent behaviour.

#### Safety of participants

The safety of participants at the restorative justice meetings is of paramount concern to the coordinators/facilitators. The providers' consideration of safety issues begins with their assessment of the suitability to take part in a restorative justice meeting of the offenders and victims who cases are referred. If the coordinator/facilitator has any safety concerns the meeting will not go ahead.

At the meeting, additional safety elements at Timaru include the presence of a Police representative, and at Nelson include support people for the victim and offender. In Christchurch all participants are required to sign an agreement saying that they will abide by the rules set down for the meeting and there is a specific protocol setting out the areas to discuss with family violence victims and offenders prior to the restorative justice process taking place The facilitator's experience and ability in dealing with these sorts of cases is also key. If the victim says she or he feels unsafe, the facilitator will stop the meeting, and the offender will be returned to the court to be dealt with in the usual way.

In preparation for the restorative justice meeting, approximately a third of victims and a small number of offenders said that they had been asked by the coordinator/facilitator whether or not they had any concerns for their safety if they participated. More than two-fifths of offenders said that they had been informed of the safety rules under which the meeting would be conducted - zero tolerance of abuse of any type.

A few victims said there had been a direct reference to how they would be kept safe during the meeting and how they could take a break or leave at any time if they felt things were getting out of hand. Not all victims appeared to be concerned about their safety – most assumed that they would be safe during the restorative justice process due to the general rules set down for the meeting.

Key informants generally felt that their local programme was effective in keeping the victim safe during the meeting. No key informants specifically mentioned having any concerns about the safety of participants when asked to identify the strengths and weaknesses of their local restorative justice programme.

#### Agreements reached at meetings)

During 2005/2006 all five programmes used restorative justice processes in family violence cases. The restorative justice meetings held for such cases in Horowhenua, Nelson and Timaru resulted in agreements which offenders are required to complete before the disposition of their case. In the case of Horowhenua and Nelson a report is provided for the sentencing Judge, who may choose to incorporate all or part of the agreement into the offender's sentence. In community panel diversion cases- ie Timaru – a report goes to Police Prosecutions who may withdraw charges if the agreement is fulfilled by the offender.

Typically, these agreements commonly included the offender's attendance at a violence intervention programme. Written apologies or community work featured less frequently. Although at Levin community work was invariably a feature of the agreement. An additional feature of Nelson agreements was that they usually included a fun element or social activity.

Although most participants interviewed said that they were satisfied with the agreement reached at their meeting, offenders were more likely to say this than victims. Most victims felt that the offender had been made accountable for their actions. Despite this almost one half of victims felt that the offender had not been able to 'repair the harm' or make up for what they had done and a few felt that the offender had got off too lightly.

Approximately two-fifths of key informants had an opinion on the nature of agreements reached at the meetings held in their area. These opinions were fairly evenly divided. Three thought that the agreements were soft; four thought they were about right and three thought they were harsh.

Sentences for family violence offenders who have participated in restorative justice processes at the five sites and who have completed their agreements, have included discharges without conviction,

convictions and discharges and orders to come up for sentence if called upon. Most of the offenders who had been sentenced said that the Judge had referred positively to the reports from the restorative justice meetings provided by programme facilitators.

Twelve key informants rated their local restorative justice programme in terms of meeting the needs of the Judges. Three quarters rated them as being effective. Judges interviewed generally concurred with this view.

#### **Experiences of participants**

The good features of the restorative justice meeting most commonly reported by victims were the open dialogue and the healing process that occurred in a non-judgemental environment and being able to meet the offender in a safe supportive environment. Good features of the meeting most often referred to by offenders included being able to put things right, feeling supported and being treated with respect. Few victims or offenders had anything bad to say about the meeting. Offenders were more likely than victims to be satisfied overall with the meeting.

Victims' views were mixed on whether or not the violence or abusive behaviour had stopped since the restorative justice meeting. Although a third said it had another third said that the nature of the violence had just changed. However, more than one half of offenders said that they had learnt something about their behaviour and its effects on others and thought that they needed to change. Some victims thought that offenders still needed help with their violent behaviour.

All offenders and most victims said that they would recommend restorative justice processes to other victims and offenders to deal with family violence. And most victims said that they would take part in the process again should they need to. All participants generally agreed that restorative justice meetings were a good way to deal with family violence, although some qualified this by saying it depended on those involved and the circumstances relating to the violence.

#### In conclusion

Family violence offences span the continuum from less serious offences (such as wilful damage) to the most serious offences (such as rape and murder). Based on the findings of this review we think that the use of restorative justice processes in family violence cases is appropriate for the types of cases currently being accepted by providers at the five sites. That is, mostly offences in the less serious to medium seriousness range such as male assaults female offences that have not required the victim's admission to hospital.

The findings of this review indicate that restorative justice meetings for couples who want to continue their relationships, combined with offenders' participation in violence intervention programmes, may lead to positive change for some couples. Similarly, restorative justice meetings may enable couples who are no longer together gain some resolution and end the relationship on amicable terms. This may be particularly relevant where there are children involved.

Only one of the five sites included in this review had specific guidelines on issues that should be discussed for the use of restorative justice in family violence cases. The other four providers felt that their generic restorative justice protocols based on the Ministry's principles of best practice were sufficient for them to accept referrals of family violence cases on a case-by-case basis. Particular care was taken by providers at all sites in the preparation of the victim and offender, the selection of appropriate panel members, the safety of participants and the development of appropriate agreements.

We consider that restorative justice processes, as one of a suite of potential responses to family violence, need to continue to be offered. Therefore, there may be value in the Ministry of Justice developing specific guidelines for the use of restorative justice processes in family violence cases to assist providers operating in an area characterised by complex relationships and often entrenched patterns of violent behaviour.

## Chapter 1 Introduction

#### 1.1 Background

In conventional criminal justice systems, professionals representing the State make the decisions on how to respond to an offender's behaviour. Restorative justice processes, in contrast, aim to involve victims, offenders and their 'communities of care' in the decisions.<sup>1</sup>

Among common law jurisdictions, New Zealand has been at the forefront of delivering and expanding restorative justice processes. Expansion of restorative justice processes among community-based providers has continued to the point where, at the time of this review, the Ministry of Justice funded 32 such providers.<sup>2</sup> These offer community panel or victim-offender conferences (i.e. facilitated meetings) in a wide range of cases. Some of these providers deliver restorative justice processes in family, domestic and sexual violence cases. Twelve have contracts that do not allow them to deliver restorative justice processes in cases involving family violence or sexual offending. They may, nonetheless, accept such cases outside of that contract.

#### 1.2 Definition of family violence

The Domestic Violence Act 1995 provides protection for people in a wide range of relationship types. These encompass married and de-facto couples, same sex couples, children, extended family or whānau and anyone in a close personal relationship, including flatmates or caregivers.

Family violence is defined by the third section of the Domestic Violence Act 1995 as 'physical abuse, sexual abuse, and psychological abuse, including, but not limited to intimidation, harassment, damage to property, threats, and causing or allowing a child to see or hear physical, sexual or psychological abuse'.

#### 1.3 The delivery of restorative justice in family violence cases

'The use of restorative justice processes in cases of family violence and sexual violence must be very carefully considered' (Ministry of Justice, 2004; p. 19). One of the Principles of Best Practice for Restorative Justice in Criminal Cases is the use of experienced skilled facilitators and stringent safety and support mechanisms. Processes of restorative justice should be underpinned by voluntariness and place an emphasis on safety for participants at all times.

There is a much contested debate around the use of restorative justice processes in partner, family and sexual violence.<sup>3</sup> Some advocates envisage them as having the potential to increase women's choices, provide women with a voice, and draw on the support of family/whānau and friends in a way that may increase their safety.<sup>4</sup> Opponents draw attention to the unequal power relationships

<sup>1</sup> See, for example, Braithwaite (1989) and Hudson (2003).

<sup>2</sup> Twenty are funded through the Crime Prevention Unit (CPU) and 12 through the Courts.

<sup>3</sup> See, for example, Braithwaite & Daly (1994), Carbonatto (1995), Morris & Gelsthorpe (2000), and contributors to Strang & Braithwaite (2002).

See, for example, Morris & Gelsthorpe (2000). Daly (2006) has also recently argued, based on a study of victims of young sexual offenders, that restorative justice processes may be less victimising than the court process for victims.

between victims and perpetrators, and raise concerns that restorative justice processes may compromise women's safety and expose them to further victimisation.<sup>5</sup> In any event, firm research evidence on the effectiveness of restorative justice processes in family violence cases is scant, and such pointers as exist are somewhat contested.<sup>6</sup>

#### 1.4 The review

In June 2006, the Ministry of Justice contracted researchers from the Crime and Justice Research Centre (CJRC), Victoria University, to undertake a review of the delivery of restorative justice in family violence cases. The review is part of a larger project being undertaken by the Ministry and aims to further investigate the nature and extent of the use of restorative justice processes in family violence cases by focusing on five selected sites/providers. The results will contribute to a Ministry of Justice consultation process concerning the nature, extent and effectiveness of use of restorative justice processes in family violence cases.

#### 1.5 The structure of this report

This report presents the findings of the review. Chapter 2 provides a detailed account of the methods used in this study and their limitations, ethical and safety issues, and the demographics and other characteristics of the sample. Chapter 3 provides an analysis of the national survey of Ministry of Justice funded restorative justice providers in relation to family violence cases. Chapter 4 describes the five sites that were the focus of the review. Chapter 5 presents the views of victims and offenders who have taken part in a restorative justice meeting for family violence offending at the sites. Their opinions of their experiences of the use of restorative justice as a means of helping address an incident of family violence, and their general views of such use are provided. Chapter 6 describes the views of key informants on the use of restorative justice processes with family violence cases. Chapter 7 summarises the findings of the review and provides some concluding observations.

Appendix A contains tables of additional data on. Appendix B contains the interview schedules, information sheets and consent forms for the interviewees.

<sup>5</sup> See for example, Lewis et al. (2001), Busch (2002) and Stubbs (2002).

<sup>6</sup> See Stubbs (2004) for example.

# Chapter 2 Methodology

#### 2.1 Objectives

The objectives of the review were to describe the nature and extent of the delivery of restorative justice in family violence cases including the following:

- how selection of cases occurs;
- assessment of appropriateness of each referral;
- how the consent of participants is obtained;
- detail of the restorative justice processes;
- safety of participants;
- the outcomes for individuals (both victim and offender) from the processes e.g. participation in therapeutic programmes, sentencing outcomes, etc); and
- the experience of individuals who have participated in restorative justice processes in family violence or domestic violence cases.

#### 2.2 Ethical issues

An application for ethical approval for the review was submitted to the Victoria University of Wellington Human Ethics Committee detailing procedures for fully informing those being asked to take part in interviews about the research, for obtaining their consent, for providing feedback at the conclusion of the study, and for procedures for storing and maintaining the confidentiality of information. Ethics approval was granted in August 2006.

The provisions of the Privacy Act 1993 with respect to confidentiality and methods of obtaining, storing and destroying information were adhered to in this study.

#### Safety procedures

7

CJRC researchers have had considerable experience in interviewing related to sensitive topics such as family violence and have established procedures for contacting and interviewing victims and offenders. These procedures have been designed to ensure researcher and interviewee safety.

Protocols for interviewees included:

- fully informing them as to who would have access to their data;
- 'protecting' them and/or supporting them during the interview process; and
- informing them of the issues surrounding confidentiality: that they did not have to answer any questions they did not want to and that no information would be included in the report which would enable them to be identified in any way.<sup>7</sup>

Such protocols acknowledge the importance of respecting a participant's right to privacy and her / his right to refuse to answer certain questions, or indeed to be interviewed at all (see for example, Oppenheim, 1992).

Interviewees were also advised that confidentiality would only be breached if the interviewer received information that suggested the research participant or another person was at risk of harm.

In addition, the safety of interviewers was addressed by:

- putting in place debriefing procedures;
- safety protocols that required researchers to take a cell phone with them when conducting fieldwork; and
- putting in place safety measures during the interview process for example requiring them to always inform someone of their whereabouts.

#### Cultural safety

The CJRC recognises the importance of undertaking research in a culturally appropriate manner, and appropriate Māori and Pacific cultural protocols and processes guided this research.<sup>8</sup> The aim was to treat all participants courteously and with respect and to respond sensitively so as to neither offend those who are more culturally conservative nor embarrass those are less so.

Our approach was one that acknowledges and is sensitive to the needs of Māori as primary participants – programme providers, key informants and victims - in the review. Our review team included a highly respected and experienced Māori researcher. The review team also included Māori and Pacific consultants in a quality assurance role.

The effect of the research on those being studied

As researchers we are always aware of the effect the interview process may have on research participants, especially where questions involve personal or sensitive issues, and put in place processes to mediate any unintended outcomes. In general, in previous research carried out by the CJRC, participants have reported that their involvement in the research has been positive; it has given them an opportunity to reflect on their experiences and to express their feelings – both positive and negative (for example, Maxwell et al, 2004 and Kingi & Poppelwell, 2005). Steps were taken to ensure, as far as possible, that there were no negative effects from participating in the research. Interview guidelines address issues such as: reminding participants that they need not answer any particular question and how to respond should a participant become distressed. Informants were also able to have support people present if they wished.

#### 2.3 Research methodology

A mixed methods approach was utilised for the review that included analysing the national survey, reviewing and analysing reports from restorative justice providers, interviewing key informants and victims and offenders, observing restorative justice processes, and collating information on referrals. Qualitative and quantitative data were triangulated to assess the appropriateness of the use of restorative justice processes in family violence cases.

#### 2.4 Data sources

Data sources for the review included:

- the Ministry of Justice survey of Crime Prevention Unit (CPU)and Courts' contracted restorative justice providers;
- interviews with victims and offenders
- interviews with key informants;
- provider documentation on policies and procedures;
- provider information on panels/meetings held for family violence cases; and
- observations of restorative justice meetings;

#### 2.5 Interview schedules and information sheets

Interview schedules and information sheets were based on those used by the CJRC in previous restorative justice evaluations.<sup>9</sup> These were adapted for this review in consultation with the review advisory group.

#### 2.6 Sample selection

#### The five sites

The selection of the five sites was informed by the Ministry's national survey and a stock take of Ministry of Justice Restorative Justice Programmes undertaken in 2005/2006 (Wilde & Baker, 2006). Taken together these information sources provided a rough estimation of the potential number of restorative justice meetings undertaken over a six month period, and over the proposed three month observation period for the review.<sup>10</sup>

The criteria for selection was decided in consultation with the Review Advisory Group and were primarily based on the volume of family violence cases handled by providers. Secondary considerations were geographical spread and the inclusion of Māori provider groups.

The five restorative justice providers chosen were as follows: 11:

- Nga Whare Waatea Trust Mangere;
- Horowhenua Restorative Justice Programme Levin;
- Nelson Restorative Justice Services Nelson;
- Restorative Justice Services Christchurch; and
- Project Turnaround Timaru.

<sup>9</sup> See Crime and Justice Research Centre ( 2005) and Paulin, Kingi and Huirama (2005).

<sup>10</sup> See Table A2.1 in Appendix A.

One site originally chosen was withdrawn from the review due to the unavailability of information on family violence cases dealt with during the period January 2005 to June 2006. A replacement site/provider was proposed by the Ministry in mid 2007.

Initial visits were made by the review team to the five sites to facilitate a collaborative working relationship with the programme providers and to ensure that the data collection requirements of the review had minimal impact on the day-to-day workings of the programme. <sup>12</sup> The review team also drafted a letter to all the restorative justice providers who had been involved in the Ministry's national survey, informing them of the review taking place and the sites that had been selected for this.

#### National survey of restorative justice providers

In March 2006, the Ministry surveyed 30 restorative justice providers contracted through either the CPU or Courts about their use and practice of restorative justice process in family violence cases. A standard questionnaire was used for the survey (see Appendix B1). Providers were asked whether or not they accepted referrals for family violence cases and about the type of restorative justice process used (i.e. community panel or victim and offender). They were also asked about the types of family violence referred, source of referrals, consent processes, assessment of suitability of referrals, and how participants' safety was addressed. Information was sought on the facilitators' qualifications, training and experience relevant to dealing with family violence cases and in working cross-culturally. Providers were also asked to describe the recommendations or plans arising from restorative justice meetings, how compliance with these was monitored and whether or not they had protocols to seek feedback from participants.

Responses were received from 24 (80%) out of the 30, 16 were based in the North Island and eight in the South Island Twenty one of these providers reported accepting referrals for cases involving family violence. The analysis of this information is reported further in Chapter 3.

#### Volume of family violence cases

We extracted information on meetings held for family violence cases in all five sites from available programme records. In four sites (i.e. Mangere, Levin, Christchurch, and Timaru) the period covered January 2005 to June 2006. In Nelson, however, the period differed; it was October 2005 to June 2006 – due to changes in staff/practice prior to October 2005.

The following is a brief overview of the number of meetings held at each of the five sites during this period and the stage at which the restorative justice process took place. These data are described in more detail in Chapter 4.

- Nga Whare Waatea: six meetings; five pre-sentence and one post-sentence to inform a
  parole board hearing;
- Horowhenua Restorative Justice Programme: 15 meetings; all pre-sentence;
- Nelson Restorative Justice Services: 14 meetings; all pre-sentence;
- Restorative Justice Services Christchurch: 15 meetings; 12 pre-sentence, three
  community referrals two for historic sexual abuse (no charges laid) and one to arrange the
  division of property after a relationship break-up; and
- **Project Turnaround**: 19 meetings; all pre-trial community panel diversion;

#### Victim and offender interviews

The providers identified victims and offenders who had participated in a restorative justice meeting relating to a family violence offence who they considered would be suitable to interview. The process for contacting victims and offenders was to obtain their names and contact details (address and phone numbers) from the relevant provider, and post them information sheets telling them about the research. They were invited to participate, asking that they return a slip in a reply paid envelope if they did not wish to take part.

In three areas (Levin, Timaru and Nelson) the providers made the initial contact with potential interviewees by posting out the information letters. At this stage, the researchers were provided with the contact details. In the other two areas (Auckland and Christchurch) providers were happy for the researchers to identify potential interviewees from providers' records and make this initial contact by mail. In all cases, the researchers then contacted victims and offenders (by phone or in person) who did not indicate that they wished to decline from taking part, to explain more fully what their participation in the review would involve.

Attempts were made to contact 40 offenders in total. Nineteen (or 48%) were interviewed, 14 (or 35%) could not be contacted<sup>14</sup> and seven (or 18%) declined to take part, either in person or by returning the 'opt out' reply slip. Attempts were made to contact a similar number of victims (n=37). Twenty (or 54%) were interviewed,<sup>15</sup> 11 (or 30%) could not be contacted and four (or 11%) declined to take part, either in person or by 'opting out' by mail.

Informed consent was obtained from all interviewees (see Appendix B2). They were told about the aims of the review, that participation was voluntary and that they could refuse to answer questions or withdraw from the research at any time. They were informed about what participation would involve and about confidentiality, including who would have access to their information. They were told that they would not be able to be identified in any subsequent research reports.

Victims were asked about the restorative justice process, including why they decided to attend (or not), preparation for the conference, the procedures followed, the extent of participants' involvement, the attitude of the offender, the nature of the agreements reached, and whether or not their needs (including their safety) were met. The questions offenders were asked broadly mirrored those victims were asked (see Appendix B3).

Interviews with victims and offenders progressed slowly. The main problem was the difficulty in locating people rather than refusals. Victims interviewed came from across the five sites: Nelson (n=5), Timaru (n=5), Christchurch (n=4), Levin (n=4) and Auckland (n=2). Offenders interviewed also came from across the five sites: Nelson (n=5), Timaru (n=4), Christchurch (n=4), Levin (n=3) and Auckland (n=3). The demographic characteristics interviewees are presented in Table 2.1.

This was predominantly for the period from January 2005 through to June 2006. However, in some cases more recent restorative justice meeting participants were identified to boost potential interviewee numbers.

Where we were unable to contact offenders or victims this was usually because the contact details were out of date and we did not have any other method of tracing them.

All except one of the victims interviewed had attended the restorative justice meeting. The one who had not was a teenager who had been represented by a social worker.

Table 2.1 Demographic characteristics of victims and offenders interviewed: numbers

Characteristics	Victims(n=20)	Offenders(n=19)
Gender		
Female	16	5
Male	4	14
Ethnicity		
NZ Māori	7	8
NZ European	9	6
Pacific	1	3
Other <sup>1</sup>	3	2
Age		
<20	1	0
20-29	5	6
30-39	7	7
40-49	3	3
50-59	3	3
60 or over	1	0
Employment status		
Employed	92	<b>7</b> <sup>3</sup>
Beneficiary <sup>4</sup>	6	8
Other <sup>5</sup>	5	4

#### Notes:

- 1 This included those who identified for example, as Australian and Irish.
- This included a sales representative, a farmer, two factory process workers, two education professionals, two clerical/administrative workers, and an agriculture/fishing industry worker.<sup>16</sup>
- This included a health professional, a construction trades worker, a carer, an engineering worker, a stores person, an agriculture/fishing industry worker and a hospitality industry worker.
- 4 This included those on the Domestic Purposes Benefit, Unemployment, Invalids and Sickness benefits.
- 5 This included a student, a retiree, a prison inmate, a 'pensioner' and those supported by a partner.

Victims were predominantly female (16 or 80%) and offenders were predominantly male (14 or 74%). Similar numbers of victims and offenders were Māori whereas more victims than offenders were New Zealand European.<sup>17</sup> The two groups also did not differ markedly in terms of age or employment status. However, slightly more victims were employed and conversely, slightly more offenders were beneficiaries. Ten (or 50%) of the victims had dependent children (whose ages ranged from 7 months to 14 years) and half (n=5) of this number still lived with the offender.<sup>18</sup>

These descriptions are based on the Australian and New Zealand Standard Classification of Occupations (2005).

<sup>17</sup> Interviewees who identified as Māori or as Māori and one other ethnicity have been described as Māori.

This included one couple who had been jointly charged with common assault and who were both interviewed as a victim and an offender.

The nature of the relationships between victims interviewed and 'their' offenders was often that of a current (10 or 50%) or ex-partner (5 or 25%). Other relationships included siblings, children, parents and in-laws. These relationships are described in Table 2.2. The nature of the most serious offence included: Conspiracy to commit murder/injures with intent to injure (n=1), male assaults female (n=10), of common assault (n=5), grievous bodily harm (n=1), threatening act/behaviour (n=2), wilful damage (n=1).

Table 2.2 Nature of the relationship between victim and offender: numbers

Characteristics of offender	Victims (n=20)
Relationship	
Brother	1
Son/Daughter	2
Mother	1
In-law	1
Current partner	10
Ex-partner	5
Living together	
At time of offending	11
At time of interview	7
Protection order	
At time of offending	4
At time of interview	2

The nature of relationships between offenders interviewed and 'their' victims was similar and is described in Table 2.3. Again, most were either current (7 or 39%) or ex-partners (6 or 33%). The nature of the most serious offence included: arson (n=1), assault with intent to injure (n=1), male assaults female (n=9),<sup>21</sup> common assault (n=6), threatening act/behaviour (n=2).<sup>22</sup>

In two cases of 'male assaults female' there were additional charges. One was breach of a protection order and the other was unlawful possession of a firearm.

<sup>20</sup> In one case involving a 'threatening act' there was also an additional charge of intentional damage.

In two cases of 'male assaults female' there were additional charges. One was breach of a protection order and the other was unlawful possession of a firearm.

<sup>22</sup> In one case involving a 'threatening act' there was also an additional charge of intentional damage.

Table 2.3 Nature of relationship between offender and victim: numbers

Characteristics of victim	Offenders (n=19)
Relationship	
Sister <sup>1</sup>	3
Son/daughter	2
In-law	1
Current partner	7
Ex-partner	6
Living together	
At time of the offending	10
At time of interview	7
Protection order	
At time of the offending	4
At time of interview	1

Note:

The sample of victims and offenders interviewed included 11 matched 'sets', that is, they had been involved in the same incident/offfending. Most were current (n=5) or ex-partners (n=4) others were siblings (n=1) and 'in-laws' (n=1). Most victims were female (10 out of 11) and most offenders were male (10 out of 11).

Interviews with victims and offenders were undertaken between October 2006 and August 2007. The time between their interview and their attendance at a restorative justice meeting varied and is described in Table 2.4.

Table 2.4 Time since restorative justice meeting and interviews with victms' and offenders: numbers

Time since meeting	Victims (n=20)	Offenders (n=19)
Less than 6 months	4	7
6 – 12 months	12	8
More than 12 months	4	4

Twelve (or 60%) victims were interviewed six to 12 months after their meeting. Similar numbers of offenders were interviewed less than six months (7 or 39%) or six to 12 months (8 or 44%) afterwards. A small number of victims and offenders were interviewed more than a year after their restorative justice meeting.

Most interviews were conducted face-to-face (two victims were interviewed by phone) and took place at a time and venue that suited the interviewee. The majority of interviews took place during the day (16 out of 20 victims and 17 out of 19 offenders), usually in the interviewee's home (17 out of 20 victims and 14 out of 19 offenders). Interviews took on average between 30 and 45 minutes. Interviewees were given a \$20 koha and after the interview asked if they would like a summary of the review findings.

<sup>1</sup> This included one offender whose victims were his sister and mother.

#### Key informant interviews

We identified key informants at each site in consultation with programme providers. Those identified included: District Court Judges, Police prosecutors, Police Family Violence Coordinators, Police Iwi Liaison Officers, Court Victim Advisers, lawyers, and family violence intervention programme providers.

As mentioned previously, interview questions (see Appendix B3) were adapted from those we had developed previously and used successfully for similar work. Topics covered varied depending on the role that the particular informant had in relation to the restorative justice programme. Core questions included:

- their views of the suitability of the use of restorative justice processes for family violence cases;
- their views on the provider's practices and their effectiveness for victims and offenders of family violence; and
- an assessment of the provider's strengths and weaknesses; and critical factors relating to success or failure in family violence cases.

In addition to the core questions Judges were asked how they decide which family violence cases to refer and how they decide whether to accept all, part, or none of the plans/agreements reached in a restorative justice process.

Interviews were undertaken with 24 key informants during the period November 2006 to August 2007. These took place over the phone (n=8) or face-to-face (n=16) at a time and place suitable for them. Six of those interviewed were from Timaru, five from Nelson, five from Levin, four from Manukau and four from Auckland. The roles of the various key informants interviewed are presented in Table 2.5.

Table 2.5 Roles of key informants interviewed: numbers

Roles	Number (n=25)
Family violence intervention programme providers	5
Judges	4
Victim Advisers	4
Police prosecutors	4
Police Family Violence Coordinators	2
Lawyers	2
Other	2
Probation Officer	1

#### Observations

The aim was to observe at least two restorative justice meetings at each of the five sites. However, this proved to be an ongoing challenge despite the best attempts of the providers. This was due to two factors: the shortage of suitable cases or the lack of any cases involving family violence. However, we managed to observe two panel meetings at both Project Turnaround in Timaru and the Horowhenua Restorative Justice Programme and one at Nelson Restorative Justice Services, with the consent of all parties involved. This consent was obtained by the providers prior to the meeting.

The researchers explained the nature of the review to those present at the meeting and undertook to leave at any stage if anyone wished them to do so. All of the charges dealt with at the meetings observed were 'male assaults female'.

#### 2.7 Data Analysis

Interview schedules were structured and included a mixture of closed and open-ended questions which were analysed qualitatively and quantitatively. Quantitative data were entered into a Microsoft Excel 2003 spreadsheet for checking and analysis. Frequencies were produced for numerical data and qualitative data (e.g. the responses to open-ended questions) were coded around key themes. Quotes are used to illustrate these themes.

Data presented in tables are usually self explanatory, however, at times these data will be summarised in the body of the report. In general, throughout the report data have been presented as proportions or percentages – raw data have only been reported where the numbers involved are very small and it is misleading to report percentages.

#### 2.8 Limitations of the research

Some caveats should be noted in relation to the findings from this research. Firstly the five sites varied in terms of the procedures and types of agreements reached. Secondly, we were reliant on restorative justice providers to provide us with the contact details of victims and offenders. Therefore, the experiences of victims and offenders that we describe may not be typical of all those who participate in restorative justice processes in relation to family violence offending. Those who agreed to participate in the research might have been influenced in one way or another by the degree and nature of their experiences – good or bad. Thirdly, only a small number of victims and offenders interviewed were able to be interviewed. For these reasons, the extent to which the findings from this research can be generalised is limited - to all participants at a particular site, across all five sites or to other restorative justice providers accepting family violence cases.

# Chapter 3 National survey of restorative justice providers

#### 3.1 Introduction

Ministry of Justice funded restorative justice providers offer community panel or victim-offender conferences (i.e. facilitated meetings) in a wide range of cases, including family violence offending.

Victim-offender conferencing is the process used in the court-referred restorative justice pilot and in a number of the CPU-funded community-managed programmes. Following a guilty plea offenders are referred to facilitators from community-based restorative justice provider groups. The facilitators meet with the victim and the offender, and convene a conference if the offender appears suitable, and both victim and offender are willing to participate. A report on the interactions of the conference, and any agreements reached, is provided to the Court at sentencing. Agreements may include reparative and reintegrative elements, such as offers of payment or attendance at programmes designed to address the underlying causes of the offending. The Court will take the report and any agreement into account when sentencing.

Community panel processes differ in that the offender meets with a facilitator, a panel of community members, and where possible the direct victim (if there is one), to discuss the impact of their offending. Usually a plan will be agreed to address the impact of the offending, through for example, reparation or community service. Plans frequently also incorporate rehabilitative activities. These processes are sometimes used as diversion programmes, where compliance with the plan may result in a withdrawal of the charges.

The Ministry of Justice surveyed 30 restorative justice providers in March 2006 to assess their use and practice of restorative justice processes in family violence cases using a standard questionnaire.

Twenty four restorative justice providers (or 80%) responded to the survey. Six were based in Auckland, 10 were based in other parts of the North Island (Gisborne, Hastings, Levin, New Plymouth, Rotorua, Tauranga, Thames, Wanganui, Wellington, and Whangarei) and eight were based in the South Island (Ashburton, Blenheim, Christchurch, Dunedin, Greymouth, Nelson, Oamaru, and Timaru).

#### 3.2 Survey findings

Twenty one (or 88%) of the restorative justice providers who responded to the survey accepted referrals for restorative justice cases involving family violence. Three (or 13%) did not.

Of the 21 providers surveyed who accepted referrals for restorative justice cases involving family violence, 11 (or 52%) used a victim-offender process, six (or 29%) used a panel process, two (or 10%) used a combination of victim-offender and panel processes, and two (or 10%) responded 'not applicable' (NA).<sup>24</sup>

Percentage of total referrals involving family violence offences

Restorative justice providers were asked what percentage of the total number of cases referred to them involved family violence offences. Their responses are set out in Table 3.1.

Table 3.1 Percentage of total cases referred to providers (n=21) involving family violence offences

Percentage of cases	No of Providers
1-10	10
11-25	5
26-50	2
51-75	2
NA <sup>1</sup>	2

Note:

Ten (or 48%) restorative justice providers who accepted restorative justice cases involving family violence estimated that 1-10 percent of the cases referred to them involved family violence. Five (24%) estimated that 11-25 percent of the cases referred to them involved family violence. Two (or 10%) providers estimated that 26-50 percent of the cases referred to them involved family violence and another two (or 10%) estimated that 51-75 percent of the cases referred to them involved family violence; Two (or 10%) providers responded NA.<sup>25</sup>

#### Types of family violence

In the year prior to the survey, restorative justice providers who accepted restorative justice cases involving family violence received referrals for a range of types of family violence. These are described in Table 3.2 below.

Table 3.2 Types of family violence offending referred to providers (n=21) during the previous 12 months

Types of family violence	No of Providers
Physical assaults	
No medical attention needed	16
Causing injury treated by doctor or nurse	16
Requiring admission to hospital	5
Property damage	13
Threats	12
Sexual assaults	
Child	4
Adult	3
Sexual violation	
Adult	3
Child	2

<sup>1</sup> Two providers had not yet had any family violence referrals.

Sixteen (or 76%) providers had been referred cases involving physical assaults where no medical attention was needed. Sixteen (or 76%) had been referred cases involving physical assaults causing injury treated by a doctor or nurse, five (or 24%) had been referred cases involving physical assaults requiring admission to hospital and thirteen (or 62%) providers had been referred cases involving damage to property. Twelve (or 57%) providers had been referred cases involving threats. Four (or 19%) had been referred cases involving sexual assaults against children and three (or 14%) had been referred cases involving sexual assaults against adults. Three (or 14%) had been referred cases involving sexual violation against an adult, and two (or 10%) had been referred cases involving sexual violation against a child.

#### Referral sources and stages

Restorative justice providers reported that most family violence referrals came to them via the Court (n=17), defence lawyers (n=15), and the Police (n=9).<sup>26</sup> Other referral sources included Victim Support, Victim Advisers, Youth Justice, Community Probation, the Child and Adolescent Mental Health Service, Legal Aid, and Child, Youth and Family.

In general, cases referred for a restorative justice process can be categorised into two main groups:

- Police Diversion cases: cases that meet Police Diversion criteria;
- Pre-sentence cases (two types): lower<sup>27</sup> and higher seriousness<sup>28</sup> cases that do not meet the criteria for Police Diversion.

Family violence referrals from the Court were generally made at the pre-sentence stage. Nineteen (or 90%) restorative justice providers indicated this, and two (10%) indicated NA.<sup>29</sup>

Not all family violence referrals made to providers resulted in a conference or panel being held (see Table 3.3).

The numbers in brackets refer to the number of providers who reported particular referral sources.

<sup>27</sup> These are low level seriousness offences where the maximum penalty is less than 2 years' imprisonment.

These are more serious offences where the maximum penalty is 2 or more years' imprisonment.

<sup>29</sup> These were the two providers who had not yet had any referrals.

Table 3.3 Percentage of family violence referrals to providers (n=21) which resulted in a conference or panel meeting

Types of family violence	No of Providers
General	
50% or less	4
51-99%	8
100%	7
NA <sup>1</sup>	2
Sexual offending	
50% or less	5
51-99%	3
100%	0
NA <sup>2</sup>	13

#### Notes:

- 1 Two providers had not yet had any family violence referrals.
- 2 Thirteen providers had not had any family violence referrals involving sexual offending

Four (or 19%) restorative justice providers estimated that 50% or less of family violence referrals would result in a conference or panel meeting, eight (or 38%) estimated that 51-99% of family violence referrals would result in a conference or panel meeting, and seven (or 33%) estimated 100% of family violence referrals would result in a conference or panel meeting. Two (10%) responded NA.

Five (or 24%) restorative justice providers estimated that 50% or less of family violence referrals involving sexual offending would result in a conference or panel meeting, and three (or 14%) estimated that 100% of such referrals would result in a conference or panel meeting. Thirteen (or 62%) responded NA.

#### Consent processes

Eighteen (or 86%) restorative justice providers required victim consent before a case went to a conference or panel. One did not require victim consent, and two indicated NA. Providers described a variety of ways in which they obtained victims' consent. Most involved providers obtaining victims' consent face-to-face. Other ways included providers obtaining victims' consent directly by phone, or indirectly through Court Victim Advisers, Victim Support or through the conference or panel meeting facilitator.

#### Suitability of referrals

In assessing the victim and offender's suitability for use of restorative justice processes in family violence cases, providers (or facilitators on their behalf) described various screening criteria. In relation to victims, providers mentioned:

- the victim's consent to participate in a conference (or their genuine desire for a conference to proceed in their absence);
- the victim's views about what led to the offending and the victim's role (if any) in it;
- the victim's attitude towards the offender;

- what the victim hoped to achieve through the conference;
- the level of contact the victim had with the offender;
- whether the victim wanted to continue in the relationship with the offender;
- whether the victim was continuing to be offended against;
- the victim's support structures;
- the extent of the victim's vulnerability and fear of the offender; and
- the extent of the impact of the offence on the victim.

Providers' considerations in relation to the offender included:

- whether or not the offender admitted the charges;
- the offender's consent to participate in a conference;
- the offender's attitude (e.g. judged to show genuine remorse or insight into their own behaviour, took responsibility for their offending behaviour, wanted to put things right or repair the harm, wanted to change expressed through a desire to participate in programmes);
- what the offender hoped to achieve through the conference;
- the offender's history of use and level of violence;
- any history of attempts at restorative justice to resolve the violence;
- what the offender had done to take responsibility for his/her violent behaviour;
- offender's ongoing needs such as mental health, drug and alcohol issues; and
- reasons the offender gives as to why they should not go to prison.

Providers had to make judgements based not only on considerations in relation to victims' and offenders' suitability for restorative justice conferences, but also whether they as facilitators had:

- the skills to facilitate the particular case;
- the ability to provide support for the victim and the offender at the conference; and
- the ability to ensure the victim's safety from re-victimisation or threats of violence at the conference and following the conference.

#### Participants' safety

Providers who accepted referrals for family violence cases were asked to identify the mechanisms or processes they had in place to ensure victims' and offenders' safety at conferences. They tended not to make a distinction between the processes they used to ensure the victims' safety and those they used to ensure the offenders' safety. Rather, for both parties they stressed the need for particular care in the screening of referrals, pre-conference preparation, and conference facilitation. For some providers the presence of support people at the conferences was mandatory, while for others their presence was encouraged but discretionary. The mechanisms or processes providers identified included:

- rigorous screening at the referral level to assess the offender's suitability (e.g. through discussions with the Court Victim Advisers or through referrals of cases to other agencies working in the family violence area) and the victim's suitability and circumstances;
- extensive pre-conference preparation with the victim and offender, and sometimes with support people. Preparation may include the coordinator/facilitator<sup>30</sup> explaining the conference aims and processes and gaining their in-depth informed consent to proceed, informing the victim and offender of who else will be at the conference, preparing the victim and offender for how to handle strong emotion safely (including taking time out), and reassuring them of what will happen should their feelings look to be getting out of control;
- ensuring the facilitator is qualified and specifically trained in facilitation of family violence cases, and has access to professional supervision and ongoing training. (Some providers indicated they used an additional co-facilitator for conferences of family violence cases.);
- facilitating the conference in line with best practice principles (Ministry of Justice, 2004) for facilitation of restorative justice meetings;
- careful selection of the conference venue and seating arrangements. (One provider
  mentioned that they held conferences in a Court building around a table with a Police officer
  present. Another provider mentioned that conferences were held in surroundings where
  other staff/people were in close proximity.) Easy access to a phone was also mentioned as
  being a consideration;
- a panel member being specifically assigned the role of ensuring the victim's safety at the conference and their arrival and departure to and from the conference;
- the presence of support people and a balanced number of participants at the conference;
   and
- extensive post conference follow-up with the offender and victim (including their linking with support agencies).

#### Facilitators' qualifications, training and experience

30

Providers were asked to identify the qualifications, training and experience their facilitators had of family violence cases. In terms of qualifications, some providers responded by identifying the tertiary qualifications (including in counselling and law) held by their facilitators. One responded that their organisation was approved under the Domestic Violence Act to provide programmes for Māori women and children.

In terms of training, providers mentioned specific training related to family violence delivered by a range of social service organisations (e.g. Women's Refuge, Rape Crisis, Victim Support, Dove in Hawke's Bay, Safer Community Violence Intervention, STOP) together with more general training in restorative justice principles and practice (e.g. former Department for Courts training).

All providers identified their facilitators' often extensive experience in dealing with family violence. Some providers also mentioned that their community panel members invariably had training and experience in the family violence area. Indeed, that was the reason they had recruited them in the first place.

The coordinator is primarily the manager of the restorative justice programme. Sometimes coordinators also act as facilitators of restorative justice meetings – these two terms are used throughout the report.

None of the 21 providers who accepted referrals for restorative justice cases involving family violence responded that they were accredited family violence service providers to other organisations.

### Cultural issues

Providers were also asked to identify the skills and knowledge they had to work appropriately cross culturally. The main way in which providers said they did this was by attempting to recruit facilitators, community panel members, support people and Board members from a range of ethnic backgrounds, including from tangata whenua,<sup>31</sup> Pacific peoples and from peoples of other ethnic groups.

Some providers had found it difficult to recruit facilitators from diverse ethnic backgrounds. However, facilitators regardless of their ethnicity were expected to have an understanding of different cultures. As one provider put it:

An understanding of the tikanga<sup>32</sup> of whatever ethnic groups are involved in the violence is part of a facilitator's preparation and planning for the conference.

Some providers mentioned that their facilitators and community panellists were provided with some ongoing training to develop their cultural knowledge (e.g. in Māori language and culture, Treaty of Waitangi courses).

Others mentioned that they canvassed victims and offenders regarding their preferences for cultural support at the conference and accessed the support of local kaumatua<sup>33</sup> or support people from other cultures where this was the desire of particular victims or offenders. Victims and offenders might also be asked whether there were particular practices that would make the conference better for them, in which case they would try to accommodate these practices.

## Recommendations or plans<sup>34</sup>

Seventeen (or 81%) of the 21 providers indicated that their conferences or plans usually recommended that offenders be referred to other programmes or services, two (or about 10%) did not, and two (or about 10%) responded NA.<sup>35</sup>

Providers listed a range of different types of programmes or services that were usually recommended in the plans of family violence cases. Types of programmes or services most commonly mentioned were:

• living without violence programmes (e.g. stopping violence programmes for male offenders, women in safe environments programmes for female victims);

32 These are customs.

33 That is, the elders of the community.

35 These recommendations are included in a report to the Court to inform sentencing.

<sup>31</sup> That is, Māori.

The terms 'recommendations' 'agreement', 'outcome' and 'plan' are often used interchangeably to refer to agreements reached between the parties at restorative justice conferences for some actions to be undertaken. For simplicity, all outcomes of restorative justice meetings at the five sites have been referred to throughout this report as 'agreements'.

- free relationship counselling through the Family Court (6 sessions);
- individual counselling;
- anger management programmes; and
- drug and alcohol assessments, counselling and treatment programmes.

In some areas, programmes were available for offenders from specific ethnic groups (e.g. Living Without Violence for Pacific Island Men in Nelson). Mental health services, gambling addiction programmes, literacy assistance and employment related assistance were also mentioned as elements of some plans.

## Monitoring compliance

Usually the coordinator takes responsibility for monitoring offenders' compliance with their plans, including attendance at programmes or services, payment of agreed reparation to the victim or other aspects decided upon. The Court may adjourn the case for sentencing until the completion of the plan. The monitoring is usually done in liaison with the providers of the programmes or services to which offenders have been referred. Towards the end of the plan, the coordinator will usually phone the providers and any other relevant people to check the offenders' compliance, and provide a report for the Court prior to sentencing.

Where the offender fails to comply with his/her plan, the case is referred back to Court (along with a report from the restorative justice provider detailing the offender's non-compliance) and processed in the usual way. In some instances, an offender may be given a second chance to complete a programme or service prior to this happening.

#### Feedback from participants

At least 13 of the 21 providers who accepted referrals for family violence cases sought feedback from conference participants. Feedback took various forms, but most usually involved victims and offenders being posted an evaluation form with a freepost return. A minority of coordinators also sought feedback on a regular basis from Victim Advisers and other court staff, Judges, lawyers, Police, programme providers and their community panellists.

## Other comments

Of the 24 restorative justice providers who responded to the survey, 21 who currently accept referrals for family violence cases indicated they intend to continue doing so and one provider indicated they intended to accept referrals in the future. Two providers indicated they would continue to not accept referrals for family violence cases in the future.

Providers generally used the space provided for 'Other comments' to expand or qualify the responses they had given earlier in the questionnaire. No particular themes emerged from this section of the questionnaire.

# 3.3 Summary

In 2006, 24 restorative justice providers responded to a Ministry of Justice national survey about their use and practice of restorative justice processes in family violence cases.

Twenty one (or 88%) of these providers accepted referrals for restorative justice cases in family violence cases. Eleven (or 52%) used a victim-offender process, six (or 29%) used a panel process, two (or 10%) used a combination of victim-offender and panel processes, and two (or 10%) responded 'not applicable'.

Most commonly the Court (n=17), defence lawyers (n=15), and the Police (n=9) referred family violence cases to restorative justice providers. Ten (or 48%) estimated that 1-10 percent of the cases referred to them involved family violence and five (24%) estimated that 11-25 percent of the cases referred to them involved family violence.

Providers who had accepted restorative justice cases in family violence cases had received referrals for a range of family violence cases, including sexual assaults/violation (including against children), physical assaults, threats, and wilful damage.

Eighteen (or 86%) restorative justice providers required victim consent before a case went to a conference or panel. Providers used a wide range of factors in the assessment of the victim and offender's suitability for use of restorative justice processes in family violence cases.

These providers thought that particular care needed to be taken with the screening of referrals, preconference preparation, and conference facilitation in order for a restorative justice meeting to happen safely in family violence cases. None who accepted referrals for restorative justice cases in family violence cases responded that they were accredited family violence service providers to other organisations.

The main way in which providers said they worked appropriately cross culturally was through recruitment of facilitators, community panel members, support people and Board members from a range of ethnic backgrounds, including from tangata whenua, Pacific peoples and from peoples of other ethnic groups.

Seventeen (or 81%) of the 21 providers indicated that their conferences or plans usually recommended that offenders be referred to other programmes or services. Most commonly the coordinator took responsibility for monitoring offenders' compliance with their plans, including attendance at programmes or services, payment of agreed reparation to the victim or other aspects decided upon.

At least 13 of the 21 providers who accepted referrals for family violence cases sought feedback from conference participants as to their satisfaction with processes.

# Chapter 4 The programmes at five sites

## 4.1 Introduction

Programme providers at five sites were chosen for the review of the use of restorative justice processes in family violence cases. Under the terms of their contracts with the Ministry of Justice, all were required to deliver services in a manner consistent with the *Principles of Best Practice for Restorative Justice in Criminal Cases* and to have a code of ethics and a complaints procedure. Within these parameters each programme operated in a somewhat different manner, in terms of referral processes, at what stage the restorative justice process took place, and in whether or not agreement was reached at the meeting.

This Chapter provides a brief overview of each programme. This includes a description of the demographic characteristics of the population in the area where the programme operates, information on the history, infrastructure and operation of the programme, and information on family violence cases processed during 2005/2006. None of the programmes were accredited family violence service providers to any other organisation under the Domestic Violence Programmes Regulations 1996.

The following is a brief overview of the number of meetings held at each of the five sites during this period and the stage at which the restorative justice process took place. The five sites are described in more detail below.

- Nga Whare Waatea: six meetings; five pre-sentence and one post-sentence to inform a
  parole board hearing;
- Horowhenua Restorative Justice Programme: 15 meetings; all pre-sentence;
- Nelson Restorative Justice Services: 14 meetings; all pre-sentence;
- Restorative Justice Services Christchurch: 15 meetings; 12 pre-sentence, three
  community referrals two for historic sexual abuse (no charges laid) and one to arrange the
  division of property after a relationship break-up; and
- Project Turnaround: 19 meetings; all pre-trial community panel diversion.

# 4.2 Nga Whare Waatea

Demographic characteristics of the Manukau population

Nga Whare Waatea Marae<sup>36</sup> is an urban marae located in the heart of Mangere. One of the social services provided is a Restorative Justice programme. Mangere is in the densely populated Auckland region in the north of the North Island of New Zealand and is part of the Manukau Territorial Local Authority. At the time of the 2006 population census 328,968 people were living in Manukau. Thirty-five percent of the people were aged under 20 and 57% aged from 20 to 64. Only 8% were aged 65 and over.<sup>37</sup> The people who live in Manukau are ethnically diverse. Thirty-eight

A marae is a Māori community facility that usually consists of a carved meeting house, a dining hall and cooking area as well as the marae ātea (sacred space in front of the meeting house).

Of the total estimated resident population 29% were aged under 20, 59% aged 20 to 64 and 12% aged 65 and over (Statistics New Zealand Demographic Trends 2007 see <a href="https://www.stats.govtl.nz">www.stats.govtl.nz</a>)

percent of residents self-identified as being of NZ European/Pakeha ethnicity, 26% as Pacific Peoples, 20% as Asian, 14% as Māori and 2% as being of Other ethnicities.<sup>38</sup> Forty-four percent of residents of working age earned less than NZ\$20,000 per year.

**Description of the Crossing Bridges programme** 

#### History

The Nga Whare Waatea Trust is part of an urban Māori organisation<sup>39</sup> and operates the *Crossing Bridges* restorative justice programme which accepts some referrals for family violence. The programme has been operating since 2003.

#### Infrastructure

The Trust employs a part-time coordinator/manager (who is also the Operations Manager of the marae) to run the *Crossing Bridges* programme. The Trust is funded by the Ministry of Justice to run 36 restorative justice meetings annually.

The coordinator/manager<sup>40</sup> acts as the facilitator for the *Crossing Bridges* programme. The coordinator/manager is an experienced facilitator and has access to a number of similarly experienced co-facilitators including violence prevention programme tutors.

The programme operates from an office on the Nga Whare Waatea complex and is supported to some extent by the infrastructure of the marae. Most restorative justice meetings are held on site in an adjacent meeting room.

#### Type of restorative justice process

The Crossing Bridges programme uses a victim-offender meeting process.

The victim must agree to the process going ahead before a restorative justice meeting can take place. The focus of the restorative justice meetings and agreements is on 'healing the harm' within whānau. Consequently, these may be held at any stage of the criminal justice process – pre or post sentence.

Participants at the restorative justice meeting are typically a facilitator and co-facilitator, the victim and offender and any support people.

The facilitator meets again with the victim and offender two weeks after the restorative justice process has taken place. Feedback on the process is sought from victims and offenders either verbally or in written form.

Of the total estimated residential population 15% were Māori, 9% were Asian and 7% were Pacific Peoples. Seventy-nine percent identified as New Zealand European or a New Zealander. Categories are overlapping. (Statistics New Zealand 2006 Census see <a href="https://www.stats.govtl.nz">www.stats.govtl.nz</a>)

<sup>39</sup> This is Manukau Urban Māori Authority.

The coordinator is primarily the manager of the restorative justice programme. Sometimes coordinators also act as facilitators of restorative justice meetings – these two terms are used throughout the report.

#### **Cultural** issues

The programme is located within an urban Māori organisation and located on a marae. Special consideration is given to cultural issues. Gaining an understanding of the tikanga<sup>41</sup> of whatever ethnic groups are involved in the violence is a part of the facilitator's preparation and planning for the meeting.

Culturally relevant protocols (e.g. karakia<sup>42</sup> at the beginning and end of the meeting) are incorporated into processes associated with the restorative justice meeting.

## Types of family violence cases accepted

The decision whether or not to accept family violence referrals is made on a case-by-case basis. No offence types are excluded. To date the numbers dealt with have been small and encompass a variety of charges including arson, male assaults female and common assault.

Over an 18 month period, January 2005 to June 2006, *Crossing Bridges* used restorative justice processes in six family violence cases. Five were pre-sentence and one post sentence. The most serious offences for which restorative justice processes were used were common assault (n=2) arson (n=1), male assaults female (n=1), assault with intent to injure (n=1) and grievous bodily harm (n=1).

The offenders comprised five men and one woman. They were aged from 21 to 56 years. Their self-identified ethnicities were Māori (n=4), NZ European (n=1) and Samoan (n=1). Their victims were partners (n=2), sisters (n=2), <sup>43</sup> a son (n=1) and whānau member/friend (n=1). Victims were present at five out of the six restorative justice meetings.

#### Referral processes

The *Crossing Bridges* programme services referrals from the Manukau District Court. Referrals of offenders to the programme come via lawyers, the Community Probation Service or Judges. A common scenario is that a referral form is faxed through from a lawyer along with a copy of the Summary of Facts (SOF).<sup>44</sup> Referrals can take place pre and post sentencing.

A facilitator assesses the offender's suitability for the process based on an admission of guilt, a willingness to participate in the process and motivation to seek change or a solution.

## Consent processes

After the referral form has been received the facilitator will meet face-to-face with the victim and offender separately. The voluntary nature of the procedure is stressed throughout. The victim's details are usually obtained from the referral source or Police Prosecution. Victim consent is required before the process goes ahead.

The meeting with the victim takes place first. The victim is asked whether or not he/she is prepared to meet the offender and what he/she wants from the meeting/offender. They are also informed

That is customs, rules or principles.

<sup>42</sup> A prayer.

In one case the victims were the offender's sister and mother.

This is the police report on facts relating to the incident/offence.

about the restorative justice process and safety protocols procedures for the conduct of the conference are explained. This process is then repeated with the offender. A meeting will not take place if there is any indication that the victim may be re victimised or under threat.

## Participants' safety

As far as we could ascertain, the programme does not have a written safety protocol for dealing with family violence cases. However, the facilitator considers safety at referral time, where s/he assesses each referral on a case by case basis, and at the restorative justice meeting itself.

The facilitator takes precautions in relation to the safety of the victim and offender. There are a balanced number of participants and all are given a clear indication of their responsibilities and the conduct expected of them during the meeting. An experienced facilitator maintains control over the process and will adjourn or halt the meeting at any sign of trouble.

## Restorative justice agreements

Agreements were reached in five out of six cases held between January 2005 and June 2006. These agreements consisted of from two to six tasks or 'suggestions'. Attendance at programmes (n=5) was the most common suggestion. Verbal apologies (n=3) and reparation (n=1) featured less frequently in agreements.

## Monitoring compliance with agreements<sup>45</sup>

Tasks included in agreements such as apologies and reparation are usually completed before the offender reappears in Court. Little monitoring is, therefore, required by the facilitator.

#### **Outcomes**

A report of the meeting is provided to the Court 48 hours before sentencing detailing a summary of the pre-meeting discussions with the victim and offender, what happened at the meeting and the outcomes and recommendations. Recommendations for programmes can be approved by the sentencing Judge. If this is the case, the monitoring of programme completion takes part through the Court process. In such cases the offender's programme attendance is monitored by the Department of Corrections.

The outcomes were varied for the six family violence offenders whose cases were held between January 2005 and June 2006. In the post sentence case the meeting was held to aid a parole application. In one of the pre-sentence meetings the offender had been remanded in custody and a term of imprisonment was inevitable. No information on sentencing outcomes was available for the other four offenders.

As mentioned in Chapter 3, the terms 'recommendations' 'agreement', 'outcome' and 'plan' are often used interchangeably in the literature to refer to agreements reached between the parties at restorative justice conferences for some actions to be undertaken. For simplicity, all outcomes of restorative justice meetings at the five sites have been referred to throughout this report as 'agreements'.

# 4.3 Horowhenua Restorative Justice Programme

Demographic characteristics of the Levin population

The Horowhenua Restorative Justice programme is in Levin. Levin is in the south-west of North Island of New Zealand and is part of the Horowhenua district which is one of the smaller territorial authorities in the country. At the time of the 2006 population census 29,866 people were living there. Twenty-eight percent of the people were aged under 20 and 52% aged 20 to 64 years. Twenty percent were aged 65 and over. Seventy-two percent of Horowhenua residents self-identified as being of NZ European/Pakeha ethnicity, 20% as Māori. Pacific Peoples, Asian, and Other ethnicities make up the remaining 8%. Fifty-four percent of residents of working age earned less than NZ\$20,000 per year.

Description of the Horowhenua Restorative Justice programme

#### History

The Horowhenua Restorative Justice programme operates under the Horowhenua District Council. The programme began operation in 1999 as part of the then local Safer Community Council. 46

#### Infrastructure

The programme employs one full-time coordinator/facilitator who is the only staff member and who has been in the position for the length of time the programme has been in operation. The programme is funded by the Ministry of Justice to run a specified number<sup>47</sup> of restorative justice conferences annually.

The coordinator facilitates all panels. His background includes 30 years in the Police where he participated in domestic violence training courses.

The coordinator is supported by a pool of 10-12 volunteer panel members whose backgrounds and occupations reflect those of the local community and include some who work for Victim Support.

The programme operates from a 'rent free' office (originally set aside for the use of the Police Community Constable) in the local shopping mall. Most restorative justice meetings are held in the adjacent board-room or in local community rooms. The office equipment is limited, consisting of a computer and printer, some filing cabinets, and a phone.

#### Type of restorative justice process

The Horowhenua Restorative Justice programme uses a community panel process.

Towards the end of 2006 the Horowhenua District Council decided that when the current contract with the CPU expired, it no longer wanted to be involved in the delivery of restorative justice programmes in the area. Consequently, the Horowhenua Restorative Justice programme ceased to operate at the end of June 2007. Restorative justice services in the wider area will now be managed from the Palmerston North District Court. In July 2007 the Ministry of Justice issued a request for proposals from restorative justice providers in the area to provide these services.

<sup>47</sup> The programme's funding was \$50,000 per annum to cover all costs.

A restorative justice meeting only takes place if the victim consents to the process and agrees to be present, although this latter condition had not always been the case.<sup>48</sup> The victim is encouraged to bring a support person along to the meeting.

Participants at the restorative justice meeting are typically the victim, the victim's support person, the offender, the offender's support person, two community panel members (one male and one female) and one facilitator (i.e. the programme coordinator).

Feedback is obtained verbally from participants at the end of the restorative justice meeting.

#### Cultural issues

Cultural issues are addressed by making attempts to include panellists of the same ethnicity as those involved in the violence. Where appropriate, meetings may be held on a local marae.

The programme has the support of local kaumatua<sup>49</sup> and panel members include representatives from the local iwi.<sup>50</sup>

#### Types of family violence cases accepted

Except for breaches of protection orders, which are excluded, the Horowhenua Restorative Justice programme has no strict rules as to which types of offences it will accept. The programme is available for those offenders charged with a criminal offence who acknowledge their guilt, and have few recent convictions. The facilitator must also feel that the offender exhibits genuine remorse. The aim is to prevent re-offending particularly in the area of family violence.

Over an 18 month period, January 2005 to June 2006, the Horowhenua Restorative Justice programme used restorative justice processes in 15 family violence cases. Victims were present in nine (or 60%) restorative justice meetings, and agreements were reached in all cases. The most serious offences for which restorative justice processes were used were male assaults female (n=6), common assault (n=7), assault on a child (n=1) and wilful damage (n=1).

The offenders comprised 11 men and four women. They were aged from 20 to 52 years. Their self-identified ethnicities were NZ European (n=9), Pacific (n=4) and Māori (n=2). The victims involved included partners (n=7), ex-partners (n=3), daughter/step-daughters (n=3), a step-son (n=1) and a mother (n=1). Victims were present in nine (or 60%) restorative justice meetings.

## Referral processes

Referrals to the programme are made via the Levin District Court. Lawyers (including duty solicitors), Police and Victim Advisers may suggest to the Judge that the offender might be an appropriate candidate for restorative justice. Referrals take place at the pre-sentencing stage.

The facilitator assesses the offender's suitability for a restorative justice meeting based on his or her genuine remorse and a lack of recent convictions.

<sup>48</sup> At one stage restorative justice meetings for family violence could take place without the victim present. Two examples include charges of assault where the victims were teenagers and the offenders their parents. This changed when the Court expressed disapproval of this practice.

<sup>49</sup> That is, elders.

<sup>50</sup> This is the local Māori tribal group.

#### Consent processes

The facilitator meets with the offender, usually at the Court, and obtains the offender's verbal consent to participation in restorative justice processes. The case is then remanded for two weeks for contact to be made with the victim. The Victim Adviser is always the first point of contact for the victim. Victims are asked if they agree for their contact details to be released to the restorative justice provider. If they agree the facilitator then contacts and meets face-to-face with the victim to explain the process.

The facilitator makes contact by phone or in person with the victim and offender separately in preparation for the meeting. Both are encouraged to bring support people to the restorative justice meeting (although this rarely happens) and briefed on what will happen there. It is stressed that the meeting is to find a resolution to the issues between victim and offender, not to decide on guilt as a guilty plea has already been entered at Court and the facts of the incident admitted.

#### Participants' safety

As far as we could ascertain, the programme does not have a written safety protocol for dealing with family violence cases. However, the facilitator considers safety issues on a case-by-case basis, and at the restorative justice meeting itself. The process would not go ahead if the facilitator felt that the victim had been coerced or bullied into participating.

The facilitator's precautions to ensure the safety of victim and offender include setting out the ground rules for participation at the beginning of the meeting. The victim will be asked if s/he has any concerns for his/her safety at this point. If at any stage it becomes apparent that any participant is feeling unsafe the meeting will be adjourned to ask them if they want to continue. If a participant decides that s/he wishes to stop the meeting the facilitator will make sure that s/he can leave safely. Although the coordinator had initial concerns about accepting family violence referrals, there has never been a situation where safety has been an issue.

#### Restorative justice agreements

All participants sign the agreement reached at the restorative justice meeting. Usually offenders agree to undertake a stopping violence programme and/or some community work. The local stopping violence programme is subsidised by the community and runs on a koha<sup>51</sup> basis for offenders.

Agreements for the 15 family violence cases for which restorative justice meetings were held between January 2005 and June 2006, consisted of from one to four tasks each. Community work (n=8) and promises not to be violent (n=8) were the two most common tasks, followed by attendance at programmes (n=7). Reparation (n=2) and donations (n=2) featured less frequently in agreements.

Offenders charged with violence against a partner are required to attend the Horowhenua Family Violence Intervention programme. The programme will only accept those who are currently living with their partner. If appropriate, offenders are also referred to relationship counselling through the Family Court, alcohol and drug services or gambling anonymous.

#### Monitoring compliance with agreements

The facilitator usually undertakes most of the monitoring of the offender's compliance with the tasks in the agreement.

Where an offender is required to attend a programme as part of his/her agreement the case is usually adjourned for about 23 weeks to allow the offender to complete the programme. However, the offender may be required to make an appearance in court during this period to report to the Judge as to progress.

The facilitator will liaise with programme providers and any failure to attend by the offender will result in the case being returned to court.

#### Sentencing at Court

On completion of the agreement the facilitator advises the Judge by way of a written report. The offender must return to Court for sentencing regardless of whether or not s/he completes his/her agreement. All participants are informed of the final Court outcome.

Thirteen of the fifteen family violence offenders whose cases were held between January 2005 and June 2006 completed all the tasks in their agreements. All of these thirteen offenders received a discharge without conviction at Court.

## 4.4 Nelson Restorative Justice Services

Demographic characteristics of the Nelson population

The Nelson Restorative Justice Service operates in the city of Nelson, in the north of the South Island of New Zealand. At the time of the 2006 population census, 42,888 people were living there. Twenty six percent of the people were aged under 20, 59% were aged 20 to 64, and 14% were aged 65 years and over. Seventy eight percent of Nelson residents self-identified as being of NZ European/Pakeha ethnicity, 8% as Māori, 3% as Asian, and 11% as being of 'Other' ethnicities. Forty four percent of residents of working age earned less than NZ\$20,000 per year.

Description of the Just Right programme

#### **History**

The Nelson Restorative Justice Trust operates two programmes. One of these, *Just Right*, accepts some referrals for family violence. The *Just Right* programme has been operating since late 2005.

#### Infrastructure

The Trust employs two staff members to run the *Just Right* programme, along with a community panel of volunteers. The Trust is funded by the Ministry of Justice to run 110 restorative justice conferences annually.

The coordinator/facilitator is funded to work on the programme for 35 hours/week and a second facilitator is funded to work for 20 hours/week. Both staff members have been in their positions since early 2006.

One of the facilitators has a counselling diploma and both facilitators are experienced in the community sector in dealing with family violence.

The facilitators are supported by 28 volunteer panel members whose backgrounds and occupations are wide ranging.

The Trust rents some office space from a local church. Most restorative justice meetings are held in this venue or in local community rooms. The office equipment is basic, consisting of a couple of laptops and printers, some filing cabinets, and a phone.

## Type of restorative justice process

The Just Right programme uses a community panel process.

A restorative justice meeting only happens if the victim agrees to be present. The victim is encouraged to bring a support person along to the meeting. If the victim has no support person in mind, the facilitator will assist in organising an appropriate person (e.g. a person from the local Women's Centre).

Participants at the restorative justice meeting are typically the victim, the victim's support person, the offender, the offender's support person, two community panel members and one facilitator.

*Just Right* seeks written feedback from participants as to their satisfaction with the restorative justice meeting and the processes.

#### **Cultural** issues

Panel members come from a range of cultural backgrounds. The facilitator may consider the cultural background of a support person or panel member as a factor in the selection of volunteers to participate in a particular restorative justice meeting. Where appropriate, meetings may be held on a local marae and participants are referred to culturally appropriate stopping violence agencies wherever possible.

## Types of family violence cases accepted

Except for breaches of protection orders, which the coordinator excludes, *Just Right* has no blanket rules as to which types of offences it will accept. Staff do, however, take the offender's recent history of offending into account.<sup>52</sup>

Over a nine month period, October 2005 to June 2006, *Just Right* used restorative justice processes in 14 family violence cases. The most serious offences for which restorative justice processes were used were male assaults female (n=6), common assault (n=5), threatening act (n=1), threatening behaviour (n=1), wilful damage (n=1).

The offenders comprised 12 men and two women. They were aged from 25 to 55 years. Their self-identified ethnicities were NZ European (n=9), Māori (n=2), Samoan (n=2) and 'Other' ethnicity (n=1). Their relationships with the victims were that of partner (n=12) or ex-partner (n=2). Victims were present at all restorative justice meetings.

#### Referral processes

All referrals of offenders to *Just Right* go through the Court. The offender must plead guilty. Lawyers (including duty solicitors) and prosecutors may suggest to the Judge that the offender might be a suitable candidate for restorative justice. Referrals take place at the pre-sentence stage.

A *Just Right* facilitator assesses the offender's suitability to take part in a restorative justice meeting on the basis of whether the offender displays remorse, wants to put things right and accepts fault.

#### **Consent processes**

A facilitator meets with the offender, usually at the Court, and obtains the offender's verbal consent to participation in restorative justice processes. A facilitator then contacts and meets face-to-face with the victim to explain *Just Right* processes and assess the victim's suitability. Considerations include the victim's safety, level of vulnerability, fear, and support structures. At this pre-meeting the victim may provide verbal consent to participation. If so, the facilitator will also ask the victim who s/he might bring along to the restorative justice meeting to support her/him. If the victim has no support person in mind, the staff member will assist in organising an appropriate person (e.g. a person from the local Women's Centre).

The same facilitator will also have a pre-meeting with the offender to ensure the offender has a support person to accompany him/her, and to set the scene for the restorative justice meeting.

## Participants' safety

As far as we could ascertain, the programme does not have a written safety protocol for dealing with family violence cases. However, the facilitator considers safety at referral time, where s/he assesses each referral a case by case basis, and at the restorative justice meeting itself. Support persons are seen as crucial in this process and the meeting will not go ahead if either the victim or the offender does not have support.

The victim and offender may choose their own seats at the meeting. The facilitator and the community panel members are primed for body language or innuendo that may alert them to any safety risks for the victim and offender. If the victim says they feel unsafe at any time in the restorative justice meeting, the facilitator would call for time out, and after speaking further with the victim, either abandon the meeting or arrange to re-convene it.

#### Restorative justice agreements

The victim and offender sign the agreement reached at the restorative justice meeting. Usually offenders agree to organising a fun social activity (e.g. a mid week social outing with the victim, or taking the children to a fast foods outlet) and attend a violence intervention programme as part of the agreement. Offenders must pay personally for participation in such an intervention programme.

Agreements for the 14 family violence cases for which restorative justice meetings were held between October 2005 and June 2006 consisted of from two to nine tasks each. A fun social activity (n=13) and attendance at programmes (n=12) were the two most common tasks. Reparation (n=5), work for the victim (n=3), community work (n=2), apologies (n=1) and donations (n=1) featured less frequently in agreements.

Programmes/agencies that offenders are referred to include relationship counselling through the Family Court, and violence prevention programmes such as - Living without Violence (Nelson), Get

Safe (Motueka), Te Puawaitanga o Te Aroha (Nelson) and Living without Violence for Pacific Island Men.

## Monitoring compliance with agreements

The facilitator usually undertakes most of the monitoring of the offender's compliance with the tasks in the agreement. A referral is sent to the appropriate agency, where appropriate, setting out the task required.

Where an offender is required to attend a programme as part of his/her agreement, the case is usually adjourned for 18 to 21 weeks to allow the offender to complete the programme. However, the offender may be required to appear in court at some point during the 18 to 21 weeks to report to the Judge as to progress.

The facilitator will liaise with the programme provider or agency and request feedback on progress and at completion. Any failure by the offender to complete a task/attend a programme may result in an extension of time being negotiated or in the case being returned to court.

### Sentencing at Court

The offender must return to Court for sentencing regardless of whether or not s/he completes his/her agreement.

Ten of the fourteen family violence offenders whose cases were held between October 2005 and June 2006 completed all tasks in their agreements. Of these 10, four offenders subsequently had their charges withdrawn, four offenders were convicted, <sup>53</sup> and for two offenders the sentence was unknown.

## 4.5 Christchurch Restorative Justice Services

Demographic characteristics of the Christchurch population

Restorative Justice Services are located in the city of Christchurch. Christchurch is situated at the mid point of the South Island of New Zealand, on the east coast. At the time of the 2006 population census, 348,435 people were living there. Twenty-six percent of the people were under the age of 20, 60% aged 20 to 64 years, and 14% were aged 65 and over. Seventy three percent of Christchurch residents self-identified as being of NZ European/Pakeha ethnicity, 8% as Māori, 8% as Asian, 3% as Pacific Peoples, and 8% as being of 'Other' ethnicities. Forty-four percent of Christchurch residents of working age earned less than NZ\$20,000 per year.

Description of the Restorative Justice Services programme

## History

Restorative Justice Services is a Charitable Trust and has been in operation since 1998. The Trust provides restorative justice victim-offender conferences both within the criminal justice system and from community referrals (such as historic sexual abuse).

Two were convicted and discharged, one was convicted and to come up for sentence if called upon in the following 12 months, and one was fined \$400. Two were given name suppression.

#### Infrastructure

The Trust is funded by the Ministry of Justice to run 35 pre-sentence restorative justice conferences annually. They also receive funding from other sources including Christchurch City Council, Canterbury Community Trust, New Zealand Post and Telstra Clear.

The Trust employs a part-time coordinator who works 30 hours a week and a part-time administrative support person who works 12 hours a week.

The role of the coordinator is to manage and coordinate the restorative justice services provided by the Trust. The current coordinator has been in this position for the last four years.

There is a pool of 13 facilitators who preside over restorative justice meetings. All have had training in restorative justice practice and principles and receive professional supervision outside the Trust. They include school teachers, mediators, lawyers and counsellors. They hold monthly peer review meetings.

The Trust is also supported by a small number of volunteers who, along with the coordinator, provide a 'presence' on behalf of the Trust in the Christchurch District Court several times a week.

The Trust rents an office in Christchurch Community <sup>54</sup> where most restorative justice meetings are held in an adjacent board room. The rent is subsidised. The Trust owns basic office equipment including computers, printers, a fax machine and filing cabinets.

## Type of restorative justice process

Restorative Justice Services use a victim-offender meeting process.

A restorative justice meeting will only take place if the victim consents to it taking place. The process is victim focussed and participant led. The aim is to get victims and offenders together and to conduct a process whereby they can communicate and come up with an agreement of some sort. Facilitators are there to guide this process and will not contribute ideas unless specifically asked.

Participants at the restorative justice meeting are typically the victim, the offender, any support people, a facilitator and a co-facilitator/recorder.

Facilitators make contact with victims and offenders a few days after the meeting to see how they are and answer any questions they may have about the process or outcomes.

#### **Cultural** issues

The Trust has Board members and facilitators who are Māori, Pacific and of other cultures. Cultural issues are addressed by ensuring that there is a cultural match wherever possible between facilitators and those involved in the violence and referring participants to culturally appropriate stopping violence agencies.

There is a commitment by the Board to ongoing education and the structuring of policies and practice to ensure that services are delivered appropriately cross culturally.

#### Types of family violence cases accepted

Restorative Justice Services accepts referrals for family violence offences but only proceeds with these after a vetting process (see below).

Over an 18 month period, January 2005 to June 2006, Restorative Justice Services used restorative justice processes in 15 family violence cases. Twelve were pre-sentence, and three were community referrals (no charges laid). The most serious offences for which restorative justice processes were used in pre-sentence cases were sexual violation (n=1), conspiracy to murder (n=1), male assaults female (n=5), assault with a weapon (n=1) assault with intent to injure (n=1), dangerous driving causing injury (n=1), and common assault (n=2).

The offenders comprised 10 men and five women. They were aged from 17 to 56 years. Their self-identified ethnicities were NZ European (n=7), Māori (n=6), Pacific (n=1) and other ethnicity (n=1). Their victims included partners (n=4), sisters (n=3), aunts (n=2) mothers (n=2), cousins (n=2), a father (n=1) and an ex-partner (n=1). Victims were present at all restorative justice meetings.

## Referral processes

The coordinator receives referrals in three main categories: from the community where there is no formal charge; pre-sentence; and post-sentence. Referral sources include: Judges, lawyers, and community probation officers. Once a referral is received the coordinator contacts the victim and offender by mail outlining what the service offers and advising them that a facilitator will be in contact. If the case has been referred at the pre-sentence stage via the Christchurch District Court there will be a strict timeline for the convening of the meeting.

Facilitators assess the offender's suitability for a restorative justice on a case-by-case basis. Particular caution is used when assessing family violence referrals. Although the offender need not have entered a guilty plea at Court s/he must accept responsibility for the offending, be willing to take part and be able to meet safely with the victim.

#### Consent processes

Facilitators' practice is guided by a policy and procedure manual based on material sourced from the Court-referred Restorative Justice Pilot Facilitator Training Manual.<sup>55</sup> In addition, a specific protocol setting out the areas to discuss with family violence victims and offenders is included (see Appendix B4).

The first meetings with the victim and offender are crucial. The aim of these is to provide the potential participants with a clear indication of what will happen if the restorative justice process goes ahead. The facilitator first meets with the offender. Support people can also be present. If the offender is willing to participate and is able to meet safely with the victim the facilitator will then contact the victim. The facilitator then meets with the victim and any support people that are available. During this meeting the facilitator explains the process, seeks the victim's view of the offending, including whether the abuse is still going on. The victim's needs and views of what led to the offending are also sought. If the victim is willing to meet with the offender and can do so safely, the meeting will go ahead.

55

#### Participants' safety

As far as we could ascertain, the programme does not have a written safety protocol for dealing with family violence cases. However, the facilitator considers safety at selection time, where s/he selects cases on a case by case basis, and at the restorative justice meeting itself.

At the beginning of the meeting participants are required to sign an agreement which sets out the rules for the meeting. Facilitators take precautions to ensure the safety of the victim and offender. If any participant is perceived to be at risk the meeting will be terminated and the victim allowed to leave before the offender.

## Restorative justice agreements

Consensus was reached in 13 out of the 15 family violence cases for which restorative justice meetings were held between January 2005 and June 2006. Apologies were frequently given (n=10), victims and offenders offered support to each other or to work together to solve their differences (n=6) and programme attendance was often talked about (n=4). No outcomes were formalised and no monitoring was required.

Restorative Justice Services has links with culturally appropriate stopping violence agencies and can provide offenders with this information if required.

## Monitoring compliance with agreements

Agreements reached in family violence cases take the form of suggestions or ideas as to what 'should happen' for the best outcomes. The facilitator writes a report which includes a summary of the pre-meeting discussions with the victim and offender, what happened at the meeting and any agreement/consensus reached. This is sent to all participants and the Court if applicable. Suggestions that programme attendance might benefit the offender can be taken up by the sentencing Judge and made part of the sentence. If this is the case, the monitoring of programme completion takes part through the Court process and is carried out by the Department of Corrections.

#### Sentencing at Court

56

Sentencing information was available for 10 of the 12 family violence offenders who had taken part in pre-sentence meetings between January 2005 and June 2006. Eight were convicted at Court, <sup>56</sup> one had his/her charges withdrawn and one was discharged without conviction.

# 4.6 Project Turnaround, Timaru

Demographic characteristics of the Timaru population

Project Turnaround is located in Timaru, in the more populous east coast of the South Island of New Zealand. At the time of the 2006 population census, 42,870 people were living there. Twenty six percent of the people were aged under 20, 56% were aged 20 to 64 years, and 18% were aged 65 years and over. Eighty percent of Timaru residents self-identified as being of NZ European/Pakeha

Two were sentenced to supervision, two were sentenced to supervision and community work, one was sentenced to 15 months imprisonment and one was fined \$450. One was ordered to come up for sentence if called upon and one was convicted and discharged.

ethnicity, 6% as Maori, and 14% as being of Other ethnicities. Forty eight percent of residents of working age earned less than NZ\$20,000 per year.

Description of the Project Turnaround programme

#### History

*Project Turnaround* consists of two programmes. The first, a community panel adult pre-trial diversion programme, was formally evaluated in 1999.<sup>57</sup> The evaluation findings were very positive in that offenders who participated in this programme were found to have reduced their re-offending over one year compared to offenders dealt with by conventional Court processes. In addition, for those who were reconvicted, the seriousness of the major offence was not as great among offenders in the programme as it was among comparison offenders. The second, a newer programme used for more serious cases, is modelled on New Zealand Court-referred restorative justice pilot processes.<sup>58</sup> *Project Turnaround* accepts referrals for family violence into both programmes, although at the time of the data collection for this evaluation (August 2006), it had accepted referrals for family violence into its first but not its second programme.

#### Infrastructure

A Trust oversees the operation of *Project Turnaround*. The Trust is funded by the Ministry of Justice to run 80 community-panel meetings annually.

It employs one full-time coordinator/facilitator, a part-time facilitator and a part-time administrative support officer. The coordinator/facilitator has been in this position for over ten years. The Trust also employs a contract case worker (who is also a case worker for Women's Refuge) to support clients where needed.

The staff members are supported by a pool of approximately 20 volunteer panel members. The panel has a good retention rate.

Training of staff and panel members has included a session where representatives from South Canterbury Violence Intervention and South Canterbury Women's Refuge spoke about issues from both the offenders' and victims' perspectives. Many of the facilitators and community panel members have relevant vocational skills (e.g. counsellor, Women's Refuge case worker, Stopping Violence facilitator).

The Trust is based at the Timaru District Council Safer Communities office; the use of basic office equipment (e.g. computers, photocopier, printers) is included in this arrangement. Most restorative justice meetings are held at this venue or in local community rooms if there are a large number of participants.

Maxwell G, Morris A, Anderson T. (1999). Community Panel Adult Pre-Trial Diversion: Supplementary Evaluation. Wellington: Crime Prevention Unit, Department of Prime Minister and Cabinet.

<sup>58</sup> New Zealand Court-Referred Restorative Justice Pilot: Evaluation (Crime and Justice Research Centre, 2005).

#### Type of restorative justice processes

As mentioned above, *Project Turnaround* consists of two programmes: a community panel adult pretrial diversion programme, and a programme modelled on New Zealand Court-referred restorative justice pilot processes.

Participants at a community panel pre-trial diversion restorative justice meeting are typically, the offender, two community panel members, the Police Family Violence Coordinator, and one facilitator. In about half the cases the victim is also present.

At a restorative justice meeting based on Court-referred restorative justice processes and occurring post the offender's conviction are typically the victim, offender, and two facilitators. While victims and offenders are encouraged to bring support people, it is *Project Turnaround's* experience that victims and offenders of family violence cases are more likely to wish to attend by themselves.

At the conclusion of the restorative justice process, victims and offenders are invited to provide feedback on their experiences of restorative justice processes via a questionnaire.

#### **Cultural** issues

Issues of culture are addressed by the inclusion of Māori panel members where appropriate or seeking guidance and support from the particular cultural community as relevant. *Project Turnaround* has three Māori panel members.

Culturally relevant protocols (e.g. karakia at the beginning and end of the meeting) may be incorporated into processes associated with the restorative justice meeting. Where appropriate, meetings may be held on a local marae.

The facilitators and community panel members have undertaken a four week training course covering basic Māori language and cultural issues.

#### Types of family violence cases accepted

The facilitator decides whether to accept family violence cases on a case by case basis. To August 2006, *Project Turnaround* has dealt mainly with male assaults female offences where the assaults have not resulted in the victims being admitted to hospital.

Over an 18 month period, January 2005 to June 2006, *Project Turnaround* used restorative justice processes in 19 family violence cases. The most serious offences for which restorative justice processes were used were male assaults female (n=14), assault on a child (n=1) and common assault (n=4).

The offenders comprised 17 men and three women. They were aged from 18 to 50 years. Their self-identified ethnicities were NZ European (n=17) and Māori (n=3). Their victims included partners (n=14) and ex-partners (n=2), children (n=2), a father (n=1), a friend (1), and a sister (n=1). Victims were present at 10 (or 50%) of the restorative justice meetings.

## **Referral processes**

Lawyers (including duty solicitors), Police, Victim Support, and Victim Advisers may suggest to the Judge that an offender might be a suitable candidate for restorative justice. An offender must plead guilty. Referrals take place at the pre-trial stage.

A *Project Turnaround* facilitator assesses the offender's suitability for a restorative justice meeting based on his or her motivation to make positive changes, for example by demonstrating a willingness to participate in programmes such as stopping violence courses.

## **Consent processes**

The facilitator meets with the offender and obtains the offender's verbal consent to participation in restorative justice processes.

The facilitator then usually contacts the victim by phone to explain *Project Turnaround* processes. The facilitator obtains the victim's verbal consent to either, the restorative justice meeting proceeding in her/his absence, or to her/his participation in it. The facilitator assesses the victim's suitability to participate taking first and foremost into consideration the victim's wishes, plus whether s/he wants to continue in the relationship, <sup>59</sup> and whether s/he is owning any of the issues surrounding the violence. The facilitator will often also have a pre-meeting with the victim at which she will encourage the victim, if the victim agrees, to bring along a support person to the restorative justice meeting.

## Participants' safety

As far as we could ascertain, the programme does not have a written safety protocol for dealing with family violence cases. However, the facilitator considers safety at referral time, where s/he selects cases on a case by case basis, and at the restorative justice meeting itself.

The presence of a Police person at the community panel pre-trial meetings helps create feelings of safety.

The facilitator gives special consideration to the selection of the panel members (usually one woman and one man).

## **Agreements**

The victim and offender sign the agreement reached at the restorative justice meeting. Usually offenders agree to undertake some counselling and/or stopping violence programmes. Some counselling (e.g. relationship counselling through the Family Court) and programmes (e.g. Men's stopping violence programme) are free to offenders, while others the offender must fund. Less frequently agreements contain donations or community work since these sorts of sanctions are thought to be less relevant in family violence cases.

Agreements of the 20 family violence cases dealt with between January 2005 and June 2006 consisted of from one to six tasks each. Attendance at programmes (n=18) was the most common task. A written apology (n=5), community work (n=4), reparation (n=3), and donations (n=2) featured less frequently in agreements.

Programmes/agencies that offenders are referred to include: relationship counselling through the Family Court, Stopping Violence programme, personal counselling, alcohol and drug counselling, parenting programmes.

#### Monitoring compliance with agreements

The offender's case is usually adjourned while the offender is carrying out the tasks in the agreement.

The facilitator usually undertakes the monitoring of the offender's compliance with his/her agreement by liaising with programme providers.

#### **Outcomes**

Providing the offender completes his/her agreement, the outcome for the offender who has participated in a community panel meeting, is usually that the case is withdrawn. In this way, the process operates similarly to Police Diversion.<sup>60</sup>

# 4.7 Summary

The programme providers at the five sites chosen for the review were spread throughout the country: three in the North Island and two in the South Island.

At two of the sites, Levin and Nelson, the restorative justice process used is a community panel presentence model which involves two panel members and a coordinator/facilitator. At two other sites, Nga Whare Waatea and Christchurch, the restorative justice process is a victim-offender meeting with a facilitator and a co-facilitator. At Nga Whare Waatea the process takes place either pre or post-sentencing. In the case of Christchurch the restorative justice process can take place pre or post sentencing or where there is no formal charge. At the fifth site, Timaru, the process used to date is one of community panel pre-trial diversion and involves the Police, panel members and a facilitator.

Referrals come to all programmes via the Court. Referral sources include lawyers, the Judge, Community Probation Officers and Victim Advisers. Christchurch accepts referrals from the community.

All five programmes approach family violence referrals with caution and assess referrals on a caseby-case basis. Nelson and Horowhenua will not accept referrals where there has been a breach of a protection order. All programmes require the victim's consent before the restorative justice process can go ahead and encourage the participation of victims. Horowhenua and Nelson require the victim to be present at the meeting.

All programmes address cultural issues by endeavouring to match the ethnicity of facilitator and or panel members with the ethnicity of participants and referring offenders to culturally appropriate violence prevention programmes where possible. Facilitators inform themselves of the tikanga of the particular cultural group of the participants involved as part of the preparation for the meeting.

At the time this research was undertaken, the process undertaken at Project Turnaround was that cases were referred back to the Police Prosecution Service for charges to be withdrawn upon successful completion of the agreed conditions, in the same way that successful completion of conditions of the Police Adult Diversion Scheme results in a withdrawal of a charge. As part of the work on the review of the Police's diversion policy a process was clarified whereby only rare cases which qualify under the diversion policy will result in a charge being withdrawn. All cases that do not qualify under the diversion policy will be referred to the court for sentencing, with all options under the Sentencing Act 2002 available, but not for the charge to be withdrawn. The Police Adult Diversion Scheme policy, as revised from November 2007, can be obtained from the Police's website at the following link: <a href="https://www.police.govt.nz/service/diversion/policy.html">https://www.police.govt.nz/service/diversion/policy.html</a>

As far as we could ascertain, none of the programmes had a written safety protocol for dealing with family violence cases. However, the facilitators consider safety at selection time where they select cases on a case by case basis, and at the restorative justice meeting itself. Christchurch has a specific protocol setting out the areas for the facilitators to discuss with family violence victims and offenders in preparation for the meeting.

All programmes encourage the victim and offender to bring support people to the meeting. However, Nelson requires both victim and offender to have support at the meeting for the restorative justice process to go ahead.

All five programmes advise victims and offenders of the ground rules for the meeting and what would happen if these were breached. At Christchurch, at the beginning of the meeting, participants are required to sign an agreement which sets out the rules for the meeting. In all cases the meeting would be stopped if anyone felt unsafe and did not want to carry on with the process.

Over the 18 month period from January 2005 to June 2006 Timaru used restorative justice processes in 20 family violence cases, Horowhenua and Christchurch each used restorative justice processes in 15 family violence cases and Nga Whare Waatea used restorative justice processes in six family violence cases. Over a nine month period, October 2005 to June 2006, Nelson used restorative justice processes in 14 family violence cases.

Over the period January 2005 to June 2006 all five programmes used restorative justice processes for the offences of male assaults female and common assault. Other offences for which restorative justice processes were used across the five sites were: arson, and grievous bodily harm (Nga Whare Waatea); assault with intent to injure (Christchurch and Nga Whare Waatea); assault on a child (Levin and Timaru); an historical sexual offence, conspiracy to murder, assault with a weapon and dangerous driving causing injury (Christchurch); threatening behaviour/act (Nelson); and wilful damage (Nelson and Levin).

The outcomes of the restorative justice meetings held for family violence cases resulted in agreements at three out of the five sites – Horowhenua, Nelson and Timaru – which offenders are required to complete before the disposition of their case. At Nga Whare Waatea agreements take the form of suggestions of what the offender needs to do to 'heal the harm' within the whānau or family. At Christchurch agreements reached in family violence cases take the form of suggestions or ideas as to what 'should happen' for the best outcomes.

Agreements which required offenders to undertake specific actions commonly included attendance at a violence intervention programme. At Nelson a fun activity was often included and at Horowhenua the offender was frequently required to undertake community work and not to behave in a violent manner as part of the agreement. Monitoring of tasks was predominantly carried out by the restorative justice facilitator. Where recommendations for programmes are taken up by the sentencing Judge the monitoring of programme completion takes part through the Court process. That is, the Court may adjourn the case for sentencing until the completion of the plan or an offender is sentenced and his/her programme attendance is monitored by the Department of Corrections.



<sup>61</sup> Charges being withdrawn is an outcome used in the rare cases where it considered appropriate that an offender qualifies for the Police Adult Diversion Scheme. The charge is withdrawn after the offender completes the agreed conditions.

# Chapter 5 The views of victims and offenders

## 5.1 Introduction

Two of the objectives of the review were to describe the experience of individuals who have participated in restorative justice processes in family violence cases and the outcomes for these individuals (both victim and offender) from the processes. This chapter describes these experiences and outcomes from the point of view of the victims and offenders interviewed.

By asking offenders about family violence it is possible to assess the extent to which they recognise the harm caused by their actions and whether or not they are committed to changing their behaviours as the result of the restorative justice process. Seeking the views of victims, on the other hand, enables us to gauge whether or not they perceived this to have happened.

The primary focus of the chapter is to describe how victims (n=19) and offenders (n=19) experienced restorative justice panels/meetings for family violence cases and their views on the suitability of such processes in relation to dealing with family violence. We consider each step of the restorative justice process for victims and offenders, starting from the preparatory stages, through the meeting process to the outcomes or agreement decided on at the meeting. Restorative justice processes provide a forum for victims, offenders and their supporters to enable them to reach agreements about how best to deal with the offending and its consequences. The hope is that the agreement will heal the harm caused by the offending and take steps to prevent its recurrence. Victims' and offenders' views on whether or not this actually happened in 'their' meeting are described in this chapter.

# 5.2 Choosing to have a restorative justice meeting

Offenders (18 or 95%) were more likely than victims (15 or 75%) to remember whose idea it was to deal with the offending by restorative justice processes. The suggestions often came from the offender's lawyer (ten offenders and four victims reported this). In some cases the victim made the suggestion (two offenders and one victim reported this). Two victims said the Court Victim Adviser had suggested restorative justice. A further two offenders said that the restorative justice provider was at Court and had been the one to suggest that their case be dealt with in this fashion. Three victims and one offender said that they thought that the Judge had suggested restorative justice.

It was usually the restorative justice provider who had first contact with offenders. Fifteen out of the 18 reported this. Another three said that their first point of contact was their lawyer. One could not remember.

Court Victim Advisers usually had the first contact with victims. Eleven out of the 19 victims who took part in restorative justice panels/meetings reported this. The other eight said they had initially been contacted by the restorative justice provider. All said that they were comfortable being contacted about their potential participation in the process. As one commented:

I was quite happy about that. Really good. It was a good way to resolve the issue we had. Restorative justice helps to resolve the issues between the people. (female victim)

However, two victims and one offender (all women) said that they felt somewhat pressured to take part, mainly in that they were unsure of their options. One commented:

I didn't feel like I could refuse to go. I didn't know what my options were if I refused. (female victim)

One victim remarked she was made to feel guilty if she chose not to participate and this resulted in adverse consequences for the offender.

I liked the idea of the whole process. But there was a certain amount of pressure because I knew what he was like and what happened. I was being made to feel guilty if he got a conviction and that it was my fault. (female victim)

Victims and offenders were asked why they had decided to take part in the restorative justice process. <sup>63</sup> Some common themes that emerged from what they said.

They wanted to sort out the issues that had led to the offending.

So that we could address the anger thing and not have an ongoing physical [violence] thing. To make it easier for us to work it out. To address matters properly not just community work or a fine. (female victim)

They offered help. We were going through a shitty patch. We'd be as good as gold then things would just erupt. (male victim)

Because we wanted to get things right for us to carry on being together. (male offender)

It was domestic and we run a home-based business. We have to live and work together. A court sentence creates some problems, so you're going to go for the best solutions. (female victim)

• They wanted to be able to discuss their differences in a safe environment.

It was a good idea for us to sit down and discuss what had happened with the safety of others around. (female victim)

I felt comfortable that we could both express our views on what happened and would like to happen without getting into an argument. (male offender)

• Some victims wanted the offender to get help to change.

63

I wanted a solution so that [offender] would not do this again to me or to any other person. (female victim)

I wanted anger management. They said if I pressed charges it would be dealt with with RJ and he could get that. That's all I wanted. (female victim)

With [offender] assaulting me I wanted him to get help rather go to jail. I'd never heard of restorative justice before but was prepared to try anything. (female victim)

Offenders wanted to take responsibility, help themselves and/or make amends to the victim.

The victim who did not attend said that he had been asked if it was alright if the meeting went ahead without him. He agreed. He was represented by a Child, Youth and Family social worker and was happy with this.

To take responsibility for my offending. (male offender)

To help myself and the victim - my partner- we thought it was a good idea. (male offender)

To meet with victim, to make amends. (female offender)

• Others, who were no longer in the relationship, wanted closure.

Just to get my feelings out and get a bit of closure. (female victim)

To get closure. It had been drawn out over a long time. I couldn't have contact with my daughter. (male offender).

Others had more pragmatic reasons for participating. Four offenders (two female and two male) said that they took part to 'Get a lower sentence.' The concern here was that of getting a criminal record. One victim said similarly she did not want her partner 'To have a criminal record.'

Despite the existence of bail conditions that sometimes prohibited contact between the victim and offender, it was not unusual for them to have some form of contact before restorative justice meeting. This was reported by 13 (or 68%) victims and 12 (63%) offenders. 'We went together. We both agreed.' 'We just discussed what would happen.' A common scenario is illustrated by the following quote:

We were still living together. One of the bail conditions was going to be we weren't supposed to see each other but someone said we couldn't go through the restorative justice process without seeing each other so they waived that. (male offender)

Only one victim said that she felt that she had felt pressured to take part in the process at this stage. <sup>64</sup>

[Offender] pressured me to go. He thought he was becoming the victim. I would've preferred just to go to the Court. It was hard to live together with waiting for all this to come to a conclusion.

In fact two female victims said they had been the ones applying the pressure.

I pressured [offender] probably. The whole scene put huge amounts of pressure on both of us and we were already under pressure.

However, when asked to rate the impact of the offence at the time it had happened 18 (or 95%) victims said that it had had a considerable negative impact on them. In fact most said that the impact had been 'high' (n=7) or 'very high' (n=8). One stated succinctly 'It was shocking.'

In this case the victim had also been charged with willful damage.

Victims were asked to rate the impact of the offending on a scale from 1=no impact to 7 = very high impact.

The victim who did not attend the conference rated the impact of the offending as 'little'.

## 5.3 Preparation

Effective participation in a restorative justice process requires that victims and offenders are well prepared and well informed. As described previously, the first point of contact for potential participants varied, but the main point of contact for both offenders and victims was the restorative justice provider offering the service. They spoke with them on the telephone; and often sent out written information and/or met with potential participants in person.

Victims and offenders were asked about what they had been consulted on or told in relation to the preparation for the restorative justice meeting. In some cases this process had taken place some time ago and so they could not remember clearly but others could. Their responses are summarised in Table 5.1.

Table 5.1 Preparation of victms (n=19) and offenders (n=18) for the restorative justice meeting: numbers saying 'yes'

Preparation	Victims saying 'yes'	Offenders saying 'yes'
Asked about:		
Who should be there	19	19
When meeting should be held	18	18
Where meeting should be held	3	8
Whether or not any safety concerns	6	3
Told:		
What would happen at meeting	19	19
Who else would be there	19	19
In plenty of time about when meeting would be held	19	17
What they would have to do there	18	18
Any safety 'ground rules' for meeting	NA	8
How they would be kept safe at meeting	3	NA

All victims and offenders said that they had been asked who they would like to be invited to the meeting. Similarly, most said they had been consulted on the timing of the meeting. Approximately a third of victims (n=6) said that they had been asked whether or not they had any concerns for their safety if they attended a meeting. Even fewer (n=3) offenders said they had been asked if they had concerns for their safety. Similarly, few victims (n=3) and offenders (n=8) said that they had been consulted about the venue for the meeting. However, it was not unusual for providers to have access to a suitable neutral venue such as community rooms where they commonly held the meetings and so this was not frequently an issue for discussion.

All victims and offenders said that they had been told what would happen at the meeting and who would be there. Similarly all victims and all but two offenders said that they had been given plenty of notice as to when the meeting would be held. Most participants (all bar one victim and one offender)

Another three (two women and one man) said that this was not an issue for them and five could not remember whether this had been discussed.

Five (three women and two men) said that it was not an issue for them and two said they could not remember.

One offender remembered 'They told me I had to have support.'

agreed that they had been given some idea as to what would be expected of them at the meeting. Offenders were asked if they had been told about any safety 'ground rules' governing the meeting, eight said that they had. These were basically: 'Zero tolerance of any abuse of any type' and 'If it got out of hand it was over. '68 Victims were asked if they had been informed how their safety would be addressed during the meeting. Only three said that there had been a direct reference to this. What was discussed related to the conditions under which the meeting could be curtailed: 'At any point if I felt it was getting out of hand I could take a break or leave', or having support: 'Someone with you when you arrive for protection or as a means of support - up to me.'69 However, this did not mean that that all victims had concerns; most assumed that they would be safe. The following quotes are illustrative:

[Facilitator] was very good I felt very confident with him. He didn't pull any punches at all - it was his job to keep me safe.

[Facilitator] asked me if I was going to be okay to be around (offender) and if I would be okay if his family were there. Didn't want me to feel threatened.

Only two victims remembered any discussion of their safety arriving at or after the meeting. One had been asked 'Do you think there will be any trouble afterwards?' Six (four women and two men) said that in their case there was 'no reason' to discuss this.'

Victims and offenders were also asked if there had been any discussion during the preparation for the restorative justice meeting about bail conditions and/or the existence of protection orders. A few victims remembered being asked about protection orders (n=7) and bail conditions (n=6) and three said that the programme providers already had that information. The rest could not remember if they had been asked. The number of offenders reporting this was similar. Seven said they were asked about protection orders and five about bail conditions. Three said the providers already had this information. Ten offenders and six victims remembered discussing with the facilitator at this stage what the offender could do to 'repair the harm'. This was described as 'Very briefly about the types of things people can do.'

One male offender made a comment which summed up what most victims and offenders remembered being told in preparation for the meeting.

Basically that if it got out of hand it was over. No physical or verbal threats. Not to have an argument; to find a way through alternatives to deal with our problems.

On a scale from 1 ('not at all prepared') to 7 ('very well prepared') more than one third of victims (7 or 37%) and offenders (6 or 33%) said that they had felt 'very well prepared' by what they were told in preparation for the meeting. Most participants said that they felt adequately prepared (that is giving a rating of 5, 6 or 7) for the restorative justice conference although there was a noteworthy difference between offenders 83% (n=15) and victims 63% (n=12) reporting this. And one male victim said he was 'not at all prepared'. He commented:

I didn't know what the hell I was going to go in there for. I was understanding that we would talk things out there. I thought it would be like counselling.

Three offenders (two female and one male) said that this was not an issue in their case, three said there had been no direct reference to safety rules and four could not remember being told.

Three victims (two female and one male) said that safety was not an issue for them and five could not remember if they had been asked.

One of the female offenders commented on the speed of the process 'It wasn't very informative. Everything was real fast.' And a female victim explained 'It was more the emotional content - technically I was prepared.'

# 5.4 The meeting

## First impressions

Victims and offenders were asked what they thought the restorative justice meeting was trying to achieve. Common themes emerged from what they said.

To talk about what had happened incorporating both points of view.

A close to that situation. Giving her a chance to say something to me and me to her. A time to listen and hear what each other had to say and move on. It was good. We got a whole lot off our chests to each other. (male offender)

It was for both of us - giving us the chance to express how we felt at the time and talk about what happened. And what he felt and why he did what he did. (female victim)

For the offender to realise the impact of what he had done and be made accountable.

What I wanted was to scare the lights out of him. Give him something to do to get him back on track so he wouldn't do it again. (female victim)

Taking responsibility for my actions, recognising who the real victim is. Because I felt like the victim for a while. I accepted that she was the victim. (male offender)

To come to some resolution over the offending – repair the harm.

For me to apologise, acknowledge guilt, to say sorry and to try and make it right with [victim]. (male offender)

We talk about what happened on that day and how he can change the way he reacts and his temper. (female victim)

To help the offender and victim so that this would not happen again.

To sit down and be able to each discuss our views of what happened and our views on why it happened and different ways it could have been resolved. (female victim)

To come to an amicable solution and things put in place so that this wouldn't happen again. (female offender)

One offender's comments exemplify most views:

A chance for me to talk to [victim] about what happened and ... a chance to say sorry and offer to do something for her to repair the harm. Gave us both a chance to talk about what happened. It was good for someone else to hear what happened. (male offender)

Victims and offenders were also asked how they felt when they first went into the meeting. Approximately half of all victims (n=13) and offenders (n=11) reported feeling anxious, nervous or

apprehensive. However, a few (five victims and four offenders) said that they had felt positive about what was going to take place.

Good - really good - we came together. There was a little bit of tension between us before we went in but we worked it out. (female victim)

Understandably some admitted to mixed feelings. 'Apprehensive. Anxious. Hopeful. I didn't know what the outcome would be.'

## **Participation**

Victims and offenders were asked various questions about their meeting. Their views are summarised in Table 5.2.70

Table 5.2 Victms' (n=19) and offenders (n=19) views of restorative justice meeting process: numbers saying 'yes'

Views	Victims saying 'yes'	Offenders saying 'yes'
Understood what going on	19	16 <sup>1</sup>
Felt involved	18	16 <sup>2</sup>
Treated with respect	17	18
Treated fairly	NA	17
Opportunity to say what wanted	16	18
Chance to explain why offence happened	NA	17
Chance to explain effect of offence	14 <sup>3</sup>	NA
Felt ashamed of what done	NA	144
Made to feel bad	NA	2
Too scared to say what really felt	4	3
Feel unsafe	2	NA
Anyone say too much	1	1

#### Notes:

- 1 Three further offenders said they 'partly' understood.
- One other said 'partly' and two said 'no'.
- 3 Two others said 'partly', one could not remember and two said 'no' but did not comment further.
- Four others said 'partly' and the other said 'no'.

It is clear from Table 5.2 that the restorative justice meeting for most victims and most offenders was a positive experience. Victims and offenders generally understood what was happening and felt involved. For example one victim said 'Oh yes I think it really does aim at the victim.' However, another victim who said that she had only felt 'partly' involved was angry at how she had been treated.

There were times when I felt totally not involved. [They were] talking about [offender] rationalising and minimising his behaviour. It made me really angry. I began to think who is this focussed on? It was more to do with [offender's] recovery more than mine and I was just the witness.

This victim also reported that she had only felt 'partly' treated with respect. However, she acknowledged that she had an opportunity to say what she wanted and had not felt too scared to express her thoughts despite the fact that she had felt 'emotionally' unsafe at times. This had made her angry rather than afraid.

However, victims and offenders, in the main, felt that they had been treated with respect. Only one female victim had anything negative to say 'I wasn't treated as the victim not at any stage.' Similarly, victims and offenders had had the opportunity to say what they wanted. Most offenders also thought that they had been given a chance to explain their actions. One who did not said:

They had a preconceived idea I'm negative, a bad person. Nothing I could say made a difference. (male offender)

Most victims could make clear the effects of the offence on them.

I was able to say what I wanted to say and how I felt without getting too stressed out. (female victim)

Most offenders felt they were treated fairly and most said that, at the conference, they had felt ashamed of what they had done. One who did not, commented 'It never would have happened if he hadn't put his hands on me.' Another offender made a poignant comment.

Watching my mother cry was hard to handle. It made me feel ashamed for what I had done. (male offender)

Only two felt that they were dealt with at the conference as if they were 'a bad person'. On the other hand, if those saying that this had happened 'to some extent' are added, more than a quarter (5 or 28%) of offenders (four male and one female) felt that they had been treated in this way. The following are two examples of what they said.

In some bits I did. I felt guilty when I was explaining my actions. It felt like I was being made out to be a caveman type of thing - bop her and take her back to the cave. (male offender)

They did actually. I felt very small. When I got home I was quite angry with them. (male offender)

Only one victim and one offender said that someone dominated the discussion. Few victims or offenders said that they were too scared to say what they really felt. In addition, only two victims said that they felt unsafe emotionally. They explained:

You've been frightened, your senses are heightened you can misconstrue what is said. (female victim)

Anger for me is emotional. I got angry because I felt unsafe. (female victim).

One of these women also reported feeling too scared to say what she really felt. She explained 'You feel subdued you have to be polite and keep it civil.'

Some offenders (9 or 50%) and victims (7 or 37%) also said that they had support with them at the restorative justice meeting conference and almost all (all victims and 8 out of 9 offenders) said that this had been helpful. Only four victims and three offenders said that they would have liked more support. Similar numbers of victims (11 or 58%) and offenders (11 or 61%) said that someone spoke up for them in the meeting. These were commonly supporters (n=8) the facilitators (n=7) or panel

members (n=7). One offender said the victim had spoken up for him. The lack of support was not always a negative thing.

It wasn't needed. The panel were supportive in that they heard what we had to say and talked to us about [our] emotions ... I appreciated that. (female victim)

I felt okay because everyone that was there – apart from the facilitators – was whānau. (male offender)

Victims were asked to describe the behaviour of the offender during the meeting. They described offenders' behaviour as 'defensive', 'quiet', 'humble', 'willing to take part', 'subdued', 'sorry', or 'angry'. Two victims commented:

He was very sad he still is very sad. He was quite emotional - was drinking a lot at that time. (female victim)

Very erratic. He got quite angry a couple of times. (female victim)

Similarly offenders were asked to describe the behaviour of the victim. In two cases the victim was not present and so offenders could not comment. Victims were described variously as being 'angry', 'hurt', 'calm', 'emotional', 'reasonable', 'quiet' or 'involved'. Two offenders offered the following descriptions.

It was a bit of everything - happy, sad, crying - it was quite an emotional thing. (male offender)

Dominating, erratic, rude. He got in a huff when I suggested he needed anger management and called it to a halt. (female offender)

## Impressions at the end of the meeting

As mentioned previously, most victims and offenders reported being apprehensive and feeling some anxiety when they first went into the meeting. However, when asked how they felt by the end of the meeting most reported feeling relieved and positive about the experience. The following quotes provide examples of the sorts of things they said.

I left thinking man I've got a lot to do! Let's get into it; I want to get it done. You've got to want to acknowledge yourself as the offender and that you have done wrong and who the victim is and you need to make serious changes. (male offender)

I was quite happy and relieved that we'd gone through that. It had been dragging on. It had happened about two months before. (female victim)

Absolutely fine - smiles on both sides a bit of laughter, banter, very relaxed. (female victim)

However, some victims had been dissatisfied with the process (n=3) and others (n=3) had concerns about what would happened next.

I felt bewildered – thought what was that about?. It was expecting some back and forth and talking things through. It didn't happen. (female victim)

Unsure uncertain of whether [offender] was genuine or not. I was waiting to see what would happen. (female victim)

In one case where the meeting broke down the victim said 'I felt furious.' And the offender commented:

I was left flabbergasted. Feeling as if things weren't fully resolved. We were made aware that if we wanted to stop the meeting that was fine but it was just the way it was done. (female offender)

Whether or not the meeting had had a productive outcome, victims' feelings towards the offender had rarely changed. Several said they were 'still angry' and even those who had forgiven the offender to some extent had not forgotten what had happened, there was a lack of trust. The remarks of two sum up how some felt.

I'd forgiven him in the first week he'd come home. I never forgave what he'd done. (female victim)

I had a really horrible taste in mouth. But I play the game he was really happy but I stewed on it. I'm still not over it. I felt that was over the top and he got away with it as usual. (female victim)

Offenders viewed things somewhat differently, as is possibly to be expected.

I didn't have as much animosity towards her as I did previous. I got to see it from her perspective. (male offender)

I don't think her feelings had changed. I was appreciative she had supported this instead of me going to jail. (male offender)

One young man, who had committed a serious offence against members of his whānau, had this to say.

I felt sad for them (victims) because of the pain I'd caused. I realised I needed to change if I wanted a chance of reconciliation.

Good and bad features of the restorative justice meetings

During their interview, victims and offenders were asked to identify any 'good' or 'bad' features of the meeting. Their opinions were, in the main, positive. Sixty-three percent (n=12) of victims said that there were no 'bad' features. The most commonly mentioned 'bad' features were victims feeling unsupported or being made to feel that they were to blame for the offending. The next most commonly mentioned 'bad' feature was that the meeting did not achieve anything, there was no resolution. One victim did not like the venue and another felt pressured to contribute. She remarked:

You shouldn't feel pressurised to respond if you weren't confident responding. It put you on the spot sort of. I had a little bit to say but it was like they were expecting more.

Only four victims said that there were no 'good' features. By far the most common 'good' features identified were the open dialogue and the healing process that occurred in a non-judgemental environment. The following quotes illustrate the sort of comments made.

It helped the healing process - you're angry with what's happened but you've got no control over it. Time heals they say, don't they - but this process helped for sure. (female victim)

It was relaxed. I was tense. When we went in we were embarrassed by our actions and why we were there. We were frightened we might get judged. But it was okay. It was nice to have people there from the community that I knew. (female victim)

Being able to meet the offender in a safe supportive environment was also mentioned by victims as a good thing. They appreciated that people were willing to offer help and that the offender could be made accountable without going to prison. Some victims also spoke positively of facilitators or panel members.

As with the victims, offenders mentioned 'bad' features of the meeting less frequently than 'good' features. Seventy-nine percent (n=15) of offenders said that there were no 'bad' features. However, four offenders reported things that they were unhappy with. These included: the meeting not resolving anything, not being given a chance to speak without interruption and being made to feel bad. One was angry about the process and commented:

I was not supported, made to feel bad, not treated with respect. (male offender)

This was also the only offender who said that there were no 'good' features of the meeting. By far the most common 'good' features identified were that the meeting had enabled them to put things right or provided a starting point for this by allowing them to meet with the victim. The next most commonly mentioned were 'feeling supported', being 'treated with respect' and the fact that the process had been 'well organised'.

On a scale from 1 ('very dissatisfied') to 7 ('very satisfied') approximately one half of victims (9 or 47%) and offenders (9 or 50%) said that overall they were 'very satisfied' with the meeting. Most participants said that they were satisfied (that is giving a rating of 5, 6 or 7) although offenders (16 or 89%) were more likely than victims (14 or 74%) to report this. In fact two victims (one male and one female) said that that were 'very dissatisfied' with the meeting. As one remarked 'It made me feel worse.'

#### Meeting cultural needs through processes

Victims and offenders of all cultures were asked if it was important to them that the meeting took account of their culture. This was not an issue for the majority. However, four victims (one Pacific and three Māori women) said that this was very important to them. Only one of them said that her needs had not been met. She commented:

It would have been good to have someone who spoke Te Reo who I could express my feelings to. People who speak Te Reo know how important the 'baby makers' [women] are.

In addition, three offenders (one Pacific male and two Māori one of whom was female) also said that it was 'very important' that the meeting take account of their culture. Only one said that his needs had not been met, and that an acknowledgement of his culture by the facilitator and panel members 'Would have been a good thing.'

# 5.5 The agreement

## Reaching agreements

Fourteen victims and fourteen offenders had taken part in restorative justice processes which resulted in 'formal' agreements.<sup>71</sup> An agreement or resolution had been reached at all of these meetings. In most cases the scenario when deciding on the details of the agreement had been 'Panel and facilitator. They came up with ideas as we went along. We had the chance to agree.' All victims and offenders said that they could remember at least some of what was in the agreement and that they had understood what needed to be done. However, the remarks made by two female victims signal that in some cases such 'collaborative' processes may be unintentionally coercive.

I felt a bit swept along I had to go along with it - trying to come up with all the answers to turn the situation around. I was probably a little self-sacrificing for the greater good. To heal him and us. It was not what I needed. He was not made accountable; I put my own needs aside.

Mainly panel members and facilitator all chucking ideas out there I didn't say much I didn't have any idea of what he could do. It was like its Christmas time what do you want?

Agreements (n=20)<sup>72</sup> reached at these restorative justice meetings consisted of a variety of tasks ranging from one to eight. Attendance at programmes (n=12) and relationship counselling (n=10) were the two most common tasks. Community work (n=6), a fun social activity (n=5), reparation (n=4) and apologies (n=4), featured less frequently in agreements. Donations (n=2), not to offer violence (n=2) and work for the victim (n=1) were also included in a few agreements. The facilitators undertook most of the monitoring of the offender's compliance with the tasks in their agreement. None of the offenders had failed to undertake the required components of the agreement, although some were still involved in programmes or community work. However, sometimes the 'fun social activities' that were suggested as a way of 'repairing the harm' were impossible to undertake when those involved were on limited incomes.

On a scale from 1 ('very dissatisfied') to 7 ('very satisfied') approximately one half of victims (7 out of 14) and offenders (8 out of 14) said that overall they were 'very satisfied' with the agreement reached at their meeting. Most participants said that they were satisfied (that is giving a rating of 5, 6 or 7) although offenders (13 or 93%) were more likely than victims (11 or 79%) to report this. In fact one victim said that she was 'very dissatisfied' with the agreement. She did not comment further.

Five victims and five offenders took part in meetings which, although they often reached consensus (in the case of two offenders and two victims), did not result in a formal agreement to be monitored and taken into account by the court. In one, for example, a victim had taken part in a post-sentence restorative justice meeting to aid a parole application for the offender.

Fight out of the 14 victims and 14 offenders who had attended meetings that resulted in formal agreements attended the same meeting. This resulted in eight agreements in total for this group plus the agreements for the meetings that the other victims (n=6) and offenders (n=6) attended – a total of 20 agreements.

## Views of agreements and outcomes

All victims (n=19) and offenders (n=19) were also asked various questions about outcomes of their restorative justice meeting and possible effects of these on them. Table 5.3 summarises their views.<sup>73</sup>

Table 5.3 Victims' (n=19) and offenders' (n=19) views of restorative justice meeting outcomes: numbers

Views	Victims saying 'yes'	Offenders saying 'yes'
Offender apologised	13	17
Victim accepted apology	11 <sup>1</sup>	15
Offender understood how victim felt	6	13
Offender made accountable for offending	12	NA
Offender made up for what did	5 <sup>3</sup>	10

#### Notes:

- 1 Two further victims said they had 'partly' accepted the apology
- 2 Four others said that they felt 'partly' ashamed.
- 3 Three further victims said that the offender had 'partly' made up for what they had done.

More than two thirds of victims said that the offender had apologised to them at the meeting. Similarly most offenders said that they had apologised at this stage. Apologies were usually accepted. Offenders (n=13) were more likely than victims (n=6) to say that they had understood how the victim felt about the offending. However, one offender who admitted he had not been able to understand made the following comments:

To be honest no I didn't. I know what had happened - she did say to me, that because she feels safe around me it hadn't damaged her. But she was quite upset. It's hard to put yourself in someone's place.

However, despite this fact almost two thirds of victims said that they felt the offender had been made accountable for what s/he had done. Although five (two males and three females) categorically stated that this was not the case.

Not at all. And the worst part about it was after that meeting they withdrew the charges and I never got told why. It had gone on for 9 months. (male victim)

Because he played the system - there was too much change required and his way of dealing with things is to withdraw emotionally. (female victim)

Although some victims felt that the offender had made up for what s/he had done others were less sure. And almost half (seven women and two men) stated unequivocally 'no' when asked this question.

I don't think you can ever make up for it. You can learn how to live better and get on with life. (female victim)

An apology is a start. It takes a lot to make up for something like that. (female victim)

#### Good and bad features of the agreements

Victims (n=14) and offenders (n=14) were specifically asked to identify the 'good' and 'bad' features of the agreement. All victims and all but one of the offenders had something to say about 'good' features. That most often mentioned by victims (n=10) and offenders (n=7) was the attendance at a programme or involvement in counselling. Reaching and agreement was referred to by three victims and one offender. Receiving apologies was mentioned by three victims and being able to apologise by one offender. One victim said it was good 'that the offender did not go to prison', one offender said that he and his ex-partner 'were friends again' and as another offender remarked 'It brought closure and created a starting point to move on.'

Only a few victims (n=5) commented on the existence of any 'bad' features in the agreement. These were: that the offender got off too lightly, 'He would come home from community work and talk about how cushy it was'; it did not reflect the victim's concerns; and in one case it made things worse, 'Him having to pay me a set amount of money (for medical expenses) fired it up and made things worse. Even fewer offenders (n=2) commented on 'bad' features in the agreement.' They said:

I felt pressured to agree. I felt if I didn't go along with it I knew what would happen. (male offender)

Offenders (13 or 93%) were more likely than victims (9 or 64%) to say that the agreement was 'about right'. Three female victims felt that it was 'too soft'; as did one female offender who commented 'I should have been put onto a programme to change my behaviour.' One male victim said it was 'too harsh' as 'By that stage [the incident] had all been forgotten.'

A small number of victims (n=1) and offenders (n=2) said that it was important that the agreement take account of their cultural needs.<sup>74</sup> They all reported that this had happened. Agreements had incorporated the appropriated cultural elements and both offenders had attended culturally-based violence prevention programmes for men.

## 5.6 Sentencing

Offenders were asked if they had been sentenced in the Court and if so, whether they thought the Judge had taken into account their participation in a restorative justice process. Thirteen offenders said they had been sentenced. The outcomes, in order of frequency, were: discharge without conviction (n=4), convicted and discharged (n=2), supervision (n=2), prison (n=2), ordered to come up for sentence if called upon (n=1), 200 hours community work (n=1) charges withdrawn (n=1). In another four cases the charges had been withdrawn as a result of completion of the restorative justice agreement.

Eleven out of the 13 offenders said that the Judge had referred positively to the reports from the restorative justice meetings provided by programme facilitators. Comments included:

Yes he definitely did, the whole process. There was a three page report and he was quite happy to see the report it helped him and he said he felt I'd made reparation. (male offender)

The Judge said it seems she's done everything possible. Judge felt because I'd done everything possible to make up for what I'd done that I'd done enough - nothing else was needed. (female offender)

He said if it wasn't for me doing it all there was no other alternative for me - than prison. (male offender)

## 5.7 After the meeting

Satisfaction with the restorative justice process

Victims and offenders were asked various questions about the aftermath of the meeting and what they thought about the use of restorative justice processes for dealing with family violence. Table 5.4 summarises their views.

Table 5.4 Victms' (n=19) and offenders' (n=19) views of restorative justice process: numbers saying 'yes'

Views	Victims saying 'yes'	Offenders saying 'yes'
Pleased took part in the RJ meeting	17	17 <sup>1</sup>
Recommend RJ process to others to deal with fv	16 <sup>2</sup>	19
Would you take part in RJ again for fv	14	NA
RJ meeting helped off to take responsibility for behaviour	NA	14 <sup>3</sup>
Feel better about incident/offending now	11	11
Offender still needs help to address abusive behaviour	11	3
Has violence/abuse stopped	6	NA

#### Notes:

- 1 One further offender said 'partly'.
- 2 One further victim said 'partly'.
- 3 Three further offenders said 'partly'.

Most victims and offenders said that they were pleased they had taken part in the restorative justice meeting. Those who disagreed were frustrated that there had been no formal outcome or had thought that the meeting would be more constructive.

I suppose, part of me felt it was good but part of me felt it was a waste of time. Apart from him apologising, that was pretty much all that happened. (female victim)

I came away feeling frustrated and furious. It was a waste of time and didn't resolve anything. It didn't look like it would right from the start. It wasn't what I expected. I thought there would be more control and that we were only looking at the circumstances around the issue at hand. We've been divorced for 15 years. (male victim)

All offenders and most victims said that they would recommend restorative justice processes to other victims and offenders to deal with family violence. In fact almost four fifths (79%) of victims said that they would take part in the process again should the need arise.<sup>75</sup> A further two said it would

The victim who was not present at the conference was also asked this question. He said that he would be willing to attend a restorative justice meeting if he was involved in such an incident in the future 'To see what happens and to say how I felt.'

'depend on the circumstances.' However, three victims, who were women, said that they would deal with things differently if there was a 'next time'.

I wouldn't waste my bloody time. I'd be straight to Court and get it over and done with.

I wouldn't even bother calling the cops I would walk out the door and go. I did it to set boundaries. I don't want to see him in jail it just makes things worse.

Identical numbers of victims (n=11) and offenders (n=11) said that as a result of participating in the meeting they now felt 'better' about the incident/offending. It had enabled them to deal with the issues and get closure. Eight offenders said they felt 'no different', they still felt bad about what had happened.

I've felt horrendous since the moment it happened - my actions are still wrong. (male offender)

Never feel better about what you've done - it's still this awful thing. (female offender)

Similarly five victims said that they also felt 'no different' about the incident, they were still not happy about what had happened. Two said they felt 'worse'. In one case the meeting had broken down and in the other the victim had not been happy with the outcome of the meeting.

#### Taking responsibility for the offending

76

Victims (n=11) were more likely to think that offenders still needed help to address their violent behaviour, than the offenders (n=3) themselves were. Most (n=16) offenders said that they had got help or were involved in the process of getting it. Yet clearly victims still thought this was an issue. In fact, only two victims said they felt *'more safe'* as a result of taking part in the restorative justice meeting. However, most said that this was not an issue for them. Either because they no longer had a relationship with the offender (n=7) or that they had *'never felt unsafe'* (n=6). Others (n=4) said that they felt *'no different'*, they were wary of the offender.

Offenders were asked whether taking part in the restorative justice process had helped them take responsibility for their offending. The majority said that it had. Interestingly, two said that this was not an issue for them. They did not comment further.

Victims were asked whether the violence or abusive behaviour had stopped since the restorative justice meeting. Six said that it had. Another seven said that the violence had 'partly' stopped. They usually mentioned that it had changed in nature. That it was no longer physical but there was 'Lots of psychological stuff.' Five could not comment as they no longer had any contact with the offender. And another finished the relationship because the violence did not stop.

It didn't stop. The only way I can make it stop is to finish the relationship and I did that. (female victim)

In addition, offenders were asked what they had learnt about the effect of their behaviour as a result of participating in the meeting. Some said that they had learnt nothing new. The following comments sum up their views:

In addition, the victim who did not attend the conference said that he now felt 'more safe' as a result of the offender taking part in the process.

I didn't learn anything that I hadn't figured out. That I'd scared the hell out of them all. I can come up with a million reasons but no justification for what I did. (male offender)

Nothing new. The kids were there and I felt horrid about that. I did have to bow my head. (female offender)

However, others admitted that being confronted with their actions had had an effect.

I still feel like an idiot for losing it. She was really scared and I was really angry. I felt sick about it. (male offender)

Yes - it's hurt her. I've got a lot of making up to do. It gets thrown in my face a lot. Anything will bring it to the surface. (male offender)

Eleven (or 58%) offenders said that what they had learnt about their behaviour had affected their views of the use of violence or abuse. In most cases it was 'A wake-up call, 'Something that I know is not right', 'I needed to change.' Two referred to lives characterised by violence.

I feel like violence has been part of my life for a long time. I've been sexually and physically abused. I grew up believing violence was a part of life - that's how you gain respect, but really you're only gaining fear. (female offender)

I've never liked it. I saw it as I was growing up and I hated it. (male offender)

Appropriateness of restorative justice for family violence cases

Finally, both victims and offenders were asked if they thought restorative justice meetings were a good way to deal with family violence. Their responses were mainly in the affirmative. All victims thought that such processes worked in the case of family violence although some qualified this.

Yeah except for if there are people who are more violent there should be more security. Someone who can contain the offender or it could get ugly. (female victim)

I think they can be to a degree. It's more appropriate if you're actually in the relationship at the time or just shortly out of it. (male victim)

Yeah. There are a lot of people who are afraid to seek help because of high consequences of mistakes. There are consequences but you need to learn how to seek help. (female victim)

If that's what they want - if families want to be helped it (restorative justice) can help a lot. (female victim)

Offenders held similar views.

Yes for the first one, it wouldn't work if it was an ongoing thing. Couples need to be given the opportunity if they want to stay together. (male offender)

The person has really got to want to change. You can go through the motions and no-one will be any the wiser. (male offender)

It's one of the better ideas I've witnessed to sort things out. You go to Court [and] they get the information on what happened. It's cut and dried but in reality it's not like that. (male offender)

Yeah I reckon. It showed me not to do things like that again. They said I was a caring mother. I didn't see that at the time. (female offender)

In addition, two fifths (n=8) of victims said that they would have liked more information or support after the meeting. This was mainly in relation to support and/or counselling (individual or relationship). The following comments are illustrative:

There's a need for more male counselling ... if he could find someone he really liked and respected and trusted. There's no free male counselling for men. There's all the support in the world for women. This sort of behaviour won't stop until society starts to support men - they're a lost bunch! (female victim)

I would have liked to have someone to talk to and counsel me. But I had no-one. I had to battle on my own. (male victim)

Three offenders responded in a similar manner. One commented:

I needed a programme to prevent it happening again. There should be one for women. I would have been happy to go. (female offender)

#### Offenders' behaviour subsequently

77

Victims were asked how the offender had taken responsibility for his/her violent behaviour since the meeting. Approximately two thirds had positive comments to make in this regard.<sup>77</sup> The following are illustrative of what they said.

He paid for his own counselling for eight sessions. He is trying - there's no more denials, he's more honest about things. He can't pretend there's not a problem and he's working through it. It was his whole upbringing - counselling started to open a can of worms - he's trying to heal himself. This thing brought it out into the open. (female victim)

Living without violence' and parenting course and he was asked about it by family and he was honest. He said, "I fucked up". (female victim)

Possibly a little bit more relaxed. Not so wound up. Relationship counselling was not long enough. (male victim)

Five victims did not think the offender had taken responsibility for the behaviour 'Not fully, no', 'Hasn't, not really', 'Nothing, I made it clear I don't want anything to do with her.'

Offenders were also asked how they had taken responsibility for their abusive or violent behaviour since the meeting. More than three quarters reported strategies they now used. 'Anger management', 'walking away' and 'time out' were mentioned.

Walking away. Time-out phase. I made an obligation not to be physically violent. (male offender)

I guess thinking about relationships in general and if they don't work out control your behaviour and not to over react when things are beyond your control. (male offender)

This included the victim who did not attend the meeting. He said '[Offender] did something because of the meeting. She started doing things differently. She put a chart on the wall and if we did things we got stars. Before she used to yell and shout at us.'

Nature of relationship between victims and offenders subsequently

Victims were asked how their relationship with the offender had been since the restorative justice meeting. Ten (or 53%) said that the relationship had improved. 'Really awesome'. We just get along so much better.' However, one woman qualified this opinion, 'It's been alright but we both put lots of effort in.' Five (or 26%) said that it was 'Up and down. Some brilliant moments and some disastrous moments.' Three (or 16%) said that they no longer had a relationship with the offender. One victim said she 'didn't know'.

Similarly, offenders were asked how their relationship with the victim had been since the restorative justice meeting. Twelve (or 63%) said that the relationship had improved. Although two offenders qualified this: 'It got worse, then better.' 'It's functional now.' Four (or 21%) said that there was no longer a relationship. One remarked 'It determined where our paths were heading and we both realised they were heading in different directions.' Others (n=2) said that it was 'too soon' to tell. However, one offender thought that the victim was still angry with her.

#### 5.8 Views of matched victims and offenders interviewed

Victims and offenders commonly view family violence offending differently, particularly in relation to apologies given and remorse shown. They may also have different reasons for taking part in restorative justice processes in relation to the offending. This section explores the responses made by the 11 'matched' victims and offenders interviewed to some of the questions they were asked about the restorative justice (RJ) process and outcomes.<sup>78</sup>

Table 5.5 summarises the degree of congruence in the views of victims and 'their' offenders.

Table 5.5 Matched victms' and offenders' views (n=11) of restorative justice process: numbers

Views	Both saying 'yes'	Both saying 'no'	Different views
Recommend RJ process to others to deal with fv	11	0	0
Pleased took part in the RJ meeting	8	0	3
Any information or support needed	1	8	2
Victim accepted offender's apology	7	0	4
Offender understood how victim felt	5	1	5
Offender made up for what did	4	2	5
Feel better about incident/offending now	4	1	6
Offender still needs help to address abusive behaviour	1	2	8

All victims and offenders agreed that they would recommend the restorative justice process to others to deal with family violence. However, one victim qualified this as 'Only to couples wanting to work on their issues.' Approximately three quarters (n=8) were pleased that they had taken part in the restorative justice process. Where there were different views, victims (n=2) were more likely to have

78 That is, both the offender and victim involved in the same family violence incident.

reservations (i.e. respond 'partly') than offenders (n=1). One male victim said that he was not pleased that he had taken part.

Offenders and victims also mostly agreed on the extent to which they needed, or did not need, information and support after the meeting. Where there was disagreement the victims rather than the offenders said they would have liked more. One would have liked information on relationship counselling and the other wanted to be kept informed about whether the offender had completed the tasks in the agreement.

Although almost two thirds (n=7) of victims and offenders agreed that the offender's apology had been accepted there was some disparity in the views of the remaining third. Most response varied in the degree of acceptance (i.e. 'partly' versus 'yes'). However, in one case the victim disagreed with the offender who thought that his apology had been 'partly' accepted.

There was less agreement between victims and offenders as to whether the offender had understood the effect of the offending on the victim. However, in one case both the victim and offender agreed that s/he had not understood. Intuitively one would imagine that the offenders said that they understood the victims' feelings and that the victims would disagree with this, and this was so in three cases. However in another two it was quite the reverse. That is, the victims thought the offenders understood how the offending had affected them when in fact the offenders reported the opposite.

Similarly there was some disparity in the views of victims and offenders in relation to whether or not the offender had been able to make amends for the offending. Victims (n=4) were more likely to say that this had not happened. Whereas offenders were more likely to say that they had made amends to some degree (n=3) or that it was not an issue (n=2).

In addition more than one half of victims and offenders had somewhat different views on how they felt about 'the incident/offending' now. However, there was no clear pattern. For example, in two cases although the victims said that they felt 'better' the offenders said that they still felt bad about what had happened. Whereas in a further two cases the victims (one male and one female) felt 'worse' and the offenders 'no different' (this does not imply that they were not aware of the effects of the offending).

The area where there was the least agreement between victim and offender was whether or not the offender still needed help to address his/her abusive or violent behaviour. Victims were more likely than offenders to say that the offender still needed help. Seven out of the eight victims who disagreed with 'their' offender reported this. Offenders reported that they had dealt with the problem.

Most (9 out of 11) victims and their offenders agreed about the nature of their relationship since the restorative justice meeting. Six victims and offenders thought that the relationship had improved, one victim and offender said that *'it was a work in progress'* and two victims and offenders concurred that there was no relationship between them now. There was no pattern with the other two victims and offenders. In one case the victim thought that the relationship was better, whereas the offender considered the victim still to be angry at her. Conversely, in the other case the offender said that the relationship had improved and the victim said there was no relationship between them now.

## 5.9 Summary

Most victims and offenders said that they took part in restorative justice meetings voluntarily for a variety of reasons which included to:

- sort out the issues that had led to the offending;
- be able to discuss their differences in a safe environment; and
- get closure if they were no longer in a relationship.

In particular, victims wanted their offender to get help to change his/her behaviour and offenders wanted to take responsibility for their actions and make amends to the victim.

Offenders were more likely than victims (83% compared with 63%) to report that they had been adequately prepared for the meeting.

Most victims and offenders found the restorative justice meeting was a positive experience. Victims and offenders:

- generally understood what was happening and felt involved;
- felt that they had been treated with respect;
- had the opportunity to express their views; and
- were not too scared to say what they wanted.

Where participants had support at the meeting (50% of offenders and 37% of victims) they found this useful.

The good features of the restorative justice meeting most commonly reported by victims were the open dialogue and the healing process that occurred in a non-judgemental environment. They also appreciated being able to meet the offender in a safe supportive environment. Good features of the meeting most often referred to by offenders included: being able to put things right, feeling supported and being treated with respect. Few victims or offenders had anything bad to say about the meeting. Offenders (89%) were slightly more likely than victims (74%) to be satisfied overall with the meeting.

A few victims (21%) and offenders (16%) felt that it was important that the restorative justice meeting met their cultural needs. All except one reported that this had happened.

Agreements reached at meetings usually included some form or counselling or programme attendance and sometimes community work or a fun social activity. Although most participants said that they were satisfied with the agreement reached, offenders were, again, slightly more likely to say this than victims (93% compared with 79%).

Although offenders (68%) were more likely than victims (32%) to say that they had understood how the victim felt about the offending, most victims (63%) felt that the offender had been made accountable for their actions. However, almost one half of victims felt that the offender had not been able to 'repair the harm' or make up for what they had done and a few felt that the offender had got off too lightly.

Most victims and offenders were pleased that they had taken part in the restorative justice meeting. All offenders and most victims (84%) said that they would recommend restorative justice processes to other victims and offenders to deal with family violence. In fact almost four fifths (79%) of victims said that they would take part in the process again should they need to.

However, victims were more likely to think that offenders still needed help to address their violent behaviour, than offenders themselves. Most (84%) offenders said that they had got help or were involved in the process of getting it. Yet clearly victims still thought this was an issue. In fact, only two victims said they felt safer as a result of taking part in the restorative justice meeting. However, most said that this was not an issue for them: they had either ended the relationship or they had never felt unsafe.

Victims' views were mixed on whether or not the violence or abusive behaviour had stopped since the meeting. One third said it had whereas another third said that the violence had just changed in nature: from physical to psychological violence. More than one half of offenders said that they had learnt something about their behaviour and its effects on others and that they needed to change.

Offenders were slightly more likely than victims to think that their relationship with the other party had improved since the restorative justice meeting (63% compared with 53%).

Victims and offenders generally agreed that restorative justice meetings were a good way to deal with family violence, although some qualified this by saying it depended on those involved and the circumstances relating to the violence.

The eleven victims and eleven offenders interviewed who had been involved in the same incident all agreed that they would recommend the restorative justice process to others to deal with family violence. Most were pleased that they had taken part in the process; where there were differences victims were more likely to have reservations about this than offenders. There was less agreement between victims and offenders as to whether the offender had understood the effect of the offending on the victim and whether or not the offender had been able to make amends for what he/she had done. Victims were more likely than offenders to respond in the negative.

The area where there was the least agreement between the eleven victims and offenders was whether or not the offender still needed help to address his/her abusive or violent behaviour. Victims were more likely than offenders to say that the offender still needed help. This followed the trend of what all victims and offenders interviewed had reported.

Generally speaking, research has consistently shown that restorative justice processes can more fully involve victims than conventional criminal justice processes. The participation and involvement of offenders is also more achievable through restorative justice processes. Both victims and offenders express satisfaction with their involvement in restorative justice process. However, in the case of family violence there is much debate about the effectiveness or indeed appropriateness of the use of restorative justice process. Roth victims and offenders express satisfaction with their involvement in restorative justice process. The participation and involvement of offenders express satisfaction with their involvement in restorative justice process. The participation and involvement of offenders express satisfaction with their involvement in restorative justice process.

<sup>79</sup> For example, see Crime and Justice Research Centre (2005) and Paulin et al (2005)

## Chapter 6 The views of key informants

### 6.1 Introduction

This Chapter presents the views of 24 key informants representing a diverse range of stakeholder groups across the five sites. We identified key informants at each site in consultation with programme providers. Those identified included: District Court Judges, Police prosecutors, Police Family Violence Coordinators, Police Iwi Liaison Officers, Court Victim Advisers, lawyers, family violence programme providers. Their opinions were canvassed on a range of issues relating to the use of restorative justice processes in family violence cases.

## 6.2 Key informants' perspectives

#### Views in general

In response to the opening question 'In general, what are your views on the use of restorative justice processes in family violence cases?' 17 (or 71%) key informants supported it and seven (or 29%) key informants were against the use of restorative justice processes in family violence cases. Key informants held a diverse range of views across the five restorative justice (RJ) programme sites and across the various stakeholder groups they represented.

Eight (or 33% of all informants) expressed unconditional support for the use of such processes. As the following quotes illustrate, key informants with these views came from across the range of stakeholder groups:

It is a good idea because it restores the mana of the family regardless of the outcomes. It is about the wider whānau<sup>81</sup> supporting victims and offenders. Māori have always had a version of RJ. (Police)

I'm totally positive. It helps position people for change. The approach is certainly wider [than conventional Court processes.] It looks at the impact on the victim and the family. Every scenario is different. It gives people options to make positive change. (Violence intervention programme provider)

I think it's a good idea. I'm really supportive. (DC Judge)

RJ is a useful and effective vehicle by which dysfunctional families can resolve issues. (Police prosecutor)

Nine (or 38% of all informants) expressed conditional support for the use of restorative justice processes in family violence cases. Informants with these sorts of views were not from any particular stakeholder group. They said, for example:

There is every chance restorative justice processes can be subverted by the offender to favour him and not bring the benefit the complainant is seeking, but if handled properly by facilitators there can be full and frank sharing of views as to what causes the violence. (DC Judge)

It is good but it requires a specific skill set and a different approach. Family violence has different power relationships so facilitators need to be skilled at recognizing power differentials. (RJ coordinator)

It can be quite effective. (Police prosecutor)

Restorative justice can work with family violence cases. ... It is imperative that clear guidelines are in place that consider the safety of the victim and any children involved in the process. This will require facilitation from people who are skilled in the area of family violence. (Violence intervention programme provider)

Restorative justice has a place in family violence cases with regards to first time offenders. It might kick start getting them into programmes. It might be useful in situations where there's a history of family violence and the family is wanting to work on things. (Victim adviser)

I'm 60% for it, not a strong advocate. (Lawyer)

Seven (or 29%) of the 24 key informants we interviewed expressed views against the use of restorative justice processes in family violence cases. Again, those with opposing views came from across the range of stakeholder groups we interviewed. For example, they expressed their opposition in the following ways:

Restorative justice has no place in family violence cases. ... Family violence is very complex. The offender can't be 'a little bit of an abuser.' (Service provider)<sup>82</sup>

I can't see what the benefits might be. (DC Judge)

My views are that it repeats victimisation for the victim or that accountability is minimised for the offenders. (Violence intervention programme provider)

I have reservations about the use of RJ and concerns about revictimisation. There is pressure on women to be part of trying to get lower sentences for offenders. If the victim is a stranger, you get an answer at the conference and get closure. In domestic violence, a one-off meeting doesn't work. Pressure is high on the victim (to participate). It's offender orientated. (Victim Adviser)

There are serious issues. ... The facilitator needs to be extremely skilled. ... Safety after the session is the main issue. If the man intimidated them [at the meeting] how is it going to be later? Family Court counselling keeps them apart. Separate sessions to keep her safe. [Feedback was that] the facilitator had not picked up on safety issues. The outcome has been that the agreement was not adhered to. (Police Family Violence Coordinator)

Offences/offenders for which it is particularly appropriate

82

Key informants identified a range of criteria they used to determine the appropriateness of restorative justice for family violence cases. Some identified the type of cases, other focussed on the offenders, while others considered the particular victim/offender relationships. .

Three key informants' responses fell into the 'needs to be determined on a case by case basis.'

Summary Offence Act assaults and threats up to male assaults female 'depending on the circumstances around the incident'; property offences, 'low level type offences', 'less serious

<sup>&#</sup>x27;Service providers' are those providing services/support to victims such as Women's Refuge or women's centres

offences', and 'low end violence' were the types of offences some key informants described as being particularly appropriate for the use of restorative justice processes in family violence cases.

First offenders, offenders with long periods of time between offending, offenders who 'showed genuine remorse and accepted responsibilities', offenders whose 'level of violence was not considered extreme'; and offenders for whom it was a 'one-off incident' were offender types that some key informants identified as being particularly appropriate.

Two key informants thought that victim/offender relationships of the type mother/son or between siblings were particularly appropriate for the use of restorative justice processes where violence had occurred.

Lastly, some key informants identified a range of circumstances as being particularly appropriate for the use of restorative justice in family violence cases. Their responses included: 'where healing is possible for the victim'; 'where a bond exists between victim and offender'; for 'couples who need to work through issues'; where 'families want to stay together'; 'where there's no physical violence'; 'if the victim wants to stay in relationship'; 'so long as there is agreement between the parties for RJ'; and 'where the victim and offender have children together.'

#### Offences/offenders for which it is particularly inappropriate

Some key informants identified offences which they considered particularly inappropriate for the use of restorative justice processes in family violence cases, while others described victims, and offenders for whom its use was deemed particularly inappropriate. Several described a mix of offences, victims, offenders, and circumstances.

Rape and other sexual offences, and serious inter-personal violence offences (such as grievous bodily harm, aggravated assault, assault with intent to injure) were considered by those key informants who offered a view as being too serious to be dealt with through the use of restorative justice processes. One key informant thought the more serious end of male assaults female offences would be inappropriate and another that breaches of protection orders would be inappropriate.

Some key informants were of the view that classic battered woman situations or co-dependent situations were inappropriate for the use of restorative justice processes as were any situations where 'the abused woman could be intimidated or coerced into going', or where 'it may be too much of a risk for the victim' to participate.

The magnitude of the offending, the harm caused and the severity of the attack were factors that one key informant thought pertinent to the determination of whether the use of restorative justice processes were appropriate or not.

Offenders who had done 'extreme violence' were thought inappropriate for restorative justice, as were repeat offenders, long term offenders, or those offenders who had already gone through a restorative justice process.

#### Their ratings of local programmes' processes

Key informants were asked to rate on a on a five point scale from 1=very poorly to 5=very well how well their local programme was working in dealing with family violence cases. Almost half (11 or 46%) of the key informants we interviewed were of the opinion that the programme worked 'well' or

'very well' (i.e. gave a rating of 4 or 5). One (or 4%) gave a rating of 3, ('neither well nor poorly'), and three (or 12%) gave a rating of 1 or 2 ('very poorly' or 'poorly'). Nine (or 38%) said they simply did not know.

#### Strengths

Key informants who supported their local restorative justice programme described its strengths as:

• Giving victims a voice, for example:

It gives victims a true voice, the ability to articulate what has happened to them in a true way, not watered down. (Police Family Violence Coordinator)

It gives victim an opportunity to say how they felt. They get a lot out of it. (Victim Adviser)

Giving offenders an opportunity to look at the consequences of their actions, for example:

It gets the perpetrator to look at how his violence has impacted on the victim and the children. (Service provider)

 Providing an opportunity, in a less formal environment than the Court environment, for victims and offenders to look at the broader context in which the violence occurs, for example:

It gets the victim and offender to address why they are there, and to have a good look at themselves. For the offender to look at why he is violent and why he responds to stress in that way, and for the victim to look at why she accepts the violence. You get ownership of the problem. (Police prosecutor)

It looks at whole scenario. It takes on board the victims' and offenders' needs and the dynamics of the family unit and significant others. It leads to a far more robust picture. (Violence intervention programme provider)

It's particularly successful when the matter is removed out of the more formal structure of Court to a less threatening environment in which people can more freely deal with issues. Court is a sanitised structured process. (Judge)

The victim and the offender have an opportunity to confront one another in an environment where the victim can feel safe. I feel strongly that most relationships want to continue and in the main the victim just wants the violence to stop. (Lawyer)

The skills and personal attributes of the facilitators, for example:

The facilitators are very good. They have good relationships, are easy to deal with, not bone heads. (Police prosecutor)

[Facilitator] sits out of the process, is not involved in decisions, is totally impartial, has a good overview of the whole case, has done her homework, has a beautiful way with people. (Police prosecutor)

[The facilitator] is a strength ... He has gone from strength to strength in training panel members and checking with family violence experts and being open with the restorative justice processes and being accountable to the community. (Violence intervention programme provider)

Getting the local community involved, for example:

You get genuine involvement from and by the community with community panel members there – a broad understanding of what the community wants. (Police Family Violence Coordinator)

#### Weaknesses

Some key informants thought their local programme exhibited no weaknesses. They appreciated that the conventional Court processes were available as a backup should 'the wheels fall off' or 'hit a brick wall' during the restorative justice processes. Of those who identified weaknesses, none mentioned safety as an issue.<sup>83</sup> Rather they described their local programmes' weaknesses as:

• The length of time it takes them to work their way through the restorative justice processes that needed to happen, for example:

Too many remands off. If victim is not available should restorative justice go ahead? ... The time delays need tightening up. (Victim adviser)

Maybe sometimes the amount of time some things take to resolve. There's roll-on a lot. The Courts get uppity. Sometimes [the roll-on] relates to reparation, course attendance ... (Police prosecutor)

If agree on tasks, things get remanded. They take too long. If the defendant is not playing ball, they will bring it back to Court. We want a report from RJ that it is working early on. I've met with couple of RJ people to discuss that. (Judge)

• The need for their restorative justice processes to be more transparent. One commented:

There needs to be greater transparency within the process to give greater confidence that the victims are taking a full role within the RJ process but at the same time are feeling safe throughout. There is a sense that this area could be improved. (Violence intervention programme provider)

- Restorative justice agreements that lacked 'innovative ideas for addressing offending';
- A reliance on the quality of the relationship the restorative justice provider has with other criminal justice professionals. A Judge and a Police representative observed in turn:

When the resident Judge has a good rapport [with the restorative justice provider], there's a better sense of direction. Now there are circuit Judges, some of whom are not in favour. In restorative justice, you need a Judge who will drive a project ... [The programme] seems afloat and drifting. (Judge)

This RJ model is not supported by the Police. In [X] they are telling victims 'Don't do it'. (Police)

 Restorative justice reports that were light on information and in need of better formatting prior to their presentation to the Court.

In addition, some informants took the opportunity to more generally criticize restorative justice processes. These included:

83

- Processes being offender driven, due to the pressures of fixed dates for Court sentencing.
- Restorative justice providers presenting the views of the victim in the Court rather than the victims doing this themselves. For example one said:

Victims are meant to have a voice but they are not heard that clearly in the court. [The restorative justice provider] is a bigger voice... (Victim Adviser)

- The 'one-off' nature of the restorative justice meeting not serving victims or offenders in complex situations well. One informant thought that resources could be better directed into counselling.
- Victims having to relive the event at the meeting.

#### Their ratings of local programmes' agreements

Key informants were asked to rate the agreements reached at restorative justice meetings in family violence cases on a five point scale from 1=too soft to 5=too harsh. Fourteen of the key informants were unable to rate the agreements reached. Of the ten who did, opinions were fairly evenly divided: one rated the agreements reached as 'too soft' ('In my opinion they're weak. Offenders actually should be made to do more.') two as 'soft', four as 'neither soft nor harsh' and three as 'harsh'.

Some expanded on their choice of rating. Examples from informants according to their ratings of agreements include:

In my opinion they're weak. Offenders actually should be made to do more. (too soft) (Service provider)

They don't all have to do a programme. That's a concern for me. ... There's a strong urge to get counselling in place for them to understand their behaviour. It doesn't necessarily change behaviour. (soft) (Violence intervention programme provider)

[Agreements] are considered on a case by case basis. (neither soft nor harsh) (Police prosecutor)

My impression is that victims are incredibly harsh, especially on first time offenders. There's lots of variation within an offence in terms of severity. (harsh) (Victim Adviser)

#### Effectiveness of the local programme

Next, key informants were asked to rate the effectiveness of the local restorative justice programme on a five point scale from 1=very ineffective to 5=very effective in relation to its processes in family violence cases. Where key informants were able to give a rating these were, for the most part, positive. However, a small number of ratings (between one and three in the group responding to each question) indicated a clear criticism of the programme's level of effectiveness. Their responses are summarised in Table 6.1.

Table 6.1 Key informants' (N=24) ratings on programme effectiveness in family violence cases; numbers

Dimension		No Rating		
	1-2 Ineffective <sup>1</sup>	3 Neither effective or ineffective	4-5 Effective <sup>2</sup>	
Involving the victim	3	1	11	9
Holding the offender accountable	3	3	8	10
Addressing offender's needs	3	3	8	10
Keeping victim safe during meeting	2	2	7	13
Addressing ongoing safety of victim	3	2	5	14
Repairing the harm caused by the offending	1	2	7	14
Reducing reoffending	1	0	3	20

#### Notes:

- 1 This includes those who gave a rating of '1=very ineffective' or '2=ineffective'.
- 2 This includes those who gave a rating of '4=effective' or '5-very effective'.

#### Involving the victim

Fifteen (62%) key informants rated and nine were unable to rate the programmes in terms of their effectiveness in involving the victim in their processes. Of the 15 who did almost three quarters (11 or 73%) gave a rating of 4 ('effective') or 5 ('very effective'). A smaller number (n=3) gave a rating of 1 ('very ineffective') or 2 ('ineffective').

#### Holding the offender accountable

Fourteen (58%) key informants rated and 10 were unable to rate the programmes in terms of their effectiveness in holding the offender accountable during their processes. Of the 14 who did, almost three fifths (57%) gave a positive rating of 4 ('effective') or 5 ('very effective'). Three gave a rating of 3 ('neither effective nor ineffective') and an identical number gave a rating of 1 ('very ineffective') or 2 ('ineffective').

#### Repairing the harm caused by the offending

Ten key informants rated and 14 were unable to rate the programme in terms of its effectiveness in repairing the harm caused by the offending. Of the ten who did, most (80%) rated the programme as 4 ('effective') or 5 ('very effective'), two as 3 ('neither effective nor ineffective') and one gave a rating of 1 ('very ineffective').

#### Addressing the needs of the offender

Fourteen key informants rated and 10 were unable to rate the local programme in terms of its effectiveness in addressing the needs of the offender through its processes. Of the 14 who did, more than half (57%) rated the programme as 4 ('effective') or 5 ('very effective'), three as 3 ('neither effective nor ineffective'), and a further three gave a rating of 1 ('very ineffective') or 2 ('ineffective').

#### Reducing re-offending

Four (17%) key informants rated and 20 were unable to rate the programmes in terms of their effectiveness in reducing re-offending. Three out of the four who did, rated the programme as being effective – that is, either 4 or 5, and one gave a rating of 2 ('ineffective').

#### Keeping the victim safe during the meeting

Eleven key informants rated and 13 were unable to rate their local programme in terms of its effectiveness in keeping the victim safe during the meeting. Of the 11 who did, seven gave a rating of 4 ('effective') or 5 ('very effective'), two gave a rating of 3 ('neither effective nor ineffective'), and another two gave a rating of 1 ('very ineffective') or 2 ('ineffective').

#### Addressing the ongoing safety of the victim

Ten key informants rated and 14 were unable to rate the programme in terms of its effectiveness in addressing the ongoing safety of the victim.<sup>84</sup> Of the 10 who did, one half rated the programme positively as 4 or 5, two as 3, and a further three as 1 or 2.

Effectiveness in meeting the needs of Judges, victims, offenders & the community

Key informants were then asked to rate the local restorative justice programme on a five point scale from 1=very ineffective to 5=very effective on the effectiveness of its processes in family violence cases in meeting the needs of Judges, victims, offenders and the community. Their responses are summarised in Table 6.2.

Table 6.2	Key informants' (N=24) ratings on programme effectiveness in family
	violence cases in meeting needs of stakeholders; numbers

Stakeholders				
	1-2 Ineffective <sup>1</sup>	3 Neither effective or ineffective	4-5 Effective <sup>2</sup>	No Rating
The community	1	4	11	8
Offenders	1	4	11	10
Victims	3	3	7	11
The Judge	2	1	9	12

#### Notes:

- 1 This includes those who gave a rating of '1=very ineffective' or '2=ineffective'.
- 2 This includes those who gave a rating of '4=effective' or '5-very effective'.

## Judges

Twelve key informants rated and twelve were unable to rate the programme in terms of meeting the needs of the Judges. Of the 12 who did, three quarters rated the local programme as 4 ('effective') or 5 ('very effective'), one as 3 ('neither effective nor ineffective'), and two as 2 ('ineffective').

One stakeholder was of the opinion that addressing the ongoing safety of the victim was not the role of the programme provider.

Of the four District Court Judges we interviewed, one rated the local programme as 'very effective', one as 'effective', one as 'ineffective' and one was unable to rate the programme in meeting their needs.

One violence intervention programme provider explained the unenviable position that restorative justice providers could sometimes find themselves in:

A lot of times the emphasis was on meeting the Judge's need which means offenders don't do a programme. The Judge is saying why is this process taking so long? So if [restorative justice processes] are done effectively, it doesn't meet Judge's needs.

#### **Victims**

Thirteen key informants rated and 11 were unable to rate the local restorative justice programme in terms of the effectiveness of its processes in family violence cases in meeting the needs of victims. Of the 13 who did, more than half (54%) rated the programmes positively as 4 ('effective') or 5 ('very effective'). Three gave a rating of 3 ('neither effective nor ineffective'), and a further three gave an unfavourable rating of 1 ('very ineffective') or 2 ('ineffective').

#### Offenders

Fourteen key informants rated and 10 were unable to rate the programme in terms of the effectiveness of its processes in family violence cases in meeting the needs of offenders. Of the 14 who did, 11 rated the programme favourably as 4 ('effective') or 5 ('very effective'), two as 3 ('neither effective nor ineffective'), and one as 2 ('ineffective').

#### The community

Sixteen key informants rated and eight were unable to rate the programme providers in terms of their effectiveness of their processes in family violence cases in meeting the needs of the community. Of the 16 who did, more than two thirds (69%) gave a rating of 4 ('effective') or 5 ('very effective'), four as 3 ('neither effective nor ineffective') and one gave a rating of 2 ('ineffective').

One Police Family Violence Coordinator was of the view:

The community is able to deal with the 'real side', the violence within the family, because we expose it in a more open way by having the issues discussed and the violence exposed. It wouldn't happen in a court with a guilty plea.

Effectiveness in meeting the needs of different cultural groups

Key informants were asked to rate the local restorative justice programme on a five point scale from 1=very ineffective to 5=very effective on the effectiveness of its processes in meeting the needs of those victims and offenders of family violence who were Māori, Pacific or from 'Other' ethnic groups.

#### Māori

Only six key informants rated and 18 were unable to rate the programme in terms of meeting the needs of Māori. Of the six who did, four rated the programme as 4 ('effective') or 5 ('very effective'), one as 'neither effective nor ineffective', and one as 1 or 'very ineffective'.

## Pacific Island peoples

Only five key informants rated and 19 were unable to rate the programme in terms of meeting the needs of Pacific Island peoples. Of the five who did, three rated the programme as 4 ('effective') or 5 ('very effective'), and two as 3 ('neither effective nor ineffective').

#### Other ethnic groups

Only three key informants rated and 21 were unable to rate the programme in terms of meeting the needs of 'Other Ethnic Groups'. Of the three who did, one rated the programme as 'very effective', one as 'effective', and one as 'neither effective nor ineffective'.

#### Improvements in programme processes

Key informants were asked whether they thought any improvements could be made to local programme processes, and if so, to describe what these might be. Thirteen answered in the affirmative. Improvements they identified were:

- facilitators and panel members trained in understanding the dynamics of family violence and dealing with it: 'Keeping fresh and keeping it real.' It was thought this would assist in ensuring victims are kept safe throughout the whole restorative justice process;
- a panel always including a member with particular expertise in dealing with family violence;
- making restorative justice agreements more meaningful for the participants: 'Practical help instead of token gestures.' (Service provider)
- better access to programmes in some locations (for example, appropriate courses for women perpetrators and anger management programmes in Timaru and support programmes in Levin)
- a tightness around the monitoring of offenders' progress with their agreement: 'If a defendant is dragging the chain with regards to attendance or completion of programmes or counselling I want them back in Court to have to explain to me why this is so.' (Judge)
- a shortening of timeframes in which restorative justice processes occur; and
- more and better structured information in restorative justice reports for the Court: 'Some tend to be "he said, she said" '. (Lawyer)

## Quality of relationships

Key informants were asked to rate on a five point scale where 1=very poor to 5=very good the quality of their stakeholder group's relationship with the local restorative justice provider. Eighteen key informants rated the quality of their group's relationship and six were unable to do so. Of the 18 who did, almost four-fifths (78%) rated the relationship favourably as 4 ('good') or 5 ('very good'), two gave a rating of 3 ('neither good nor poor'), and two a rating of 2 ('poor').

#### Impact on agency/service

Key informants were asked whether the use of restorative justice in family violence cases had had an impact on them or their agency or service. Thirteen responded in the affirmative, saying that it had had the effect of slightly reducing their workloads. They said, for example:

The RJ process has had minimal impact on the work of the organisation because of the limited number of referrals that have been made. (Violence intervention prevention provider)

A bit but not hugely. I get a lot of clients who want to defend. I tend to use other resources. I refer people to joint family counselling through the Family Court whatever the plea. (Lawyer)

#### Final comments

Key informants were given the opportunity to add any final comments. Those who commented usually returned to themes they had expressed early in the interviews.

Supporters of the use of restorative justice processes in family violence cases tended to be realistic as to what these processes might achieve. They did not view restorative justice as a cure all in all family violence cases, but as an opportunity for some victims and offenders of family violence to come together in a less pressured environment than the Court for victims to have a voice and for offenders to start to address their offending. They said, for example:

In the vast majority of cases the relationship is going to carry on. You get women repeatedly coming back to these violent relationships for whatever reason. I appreciate that there are problems with it [restorative justice] but if people are going to be in ongoing relationships don't we need input from the victim as to how she wants it put right? ... I'm realistic for a lot of people they're not going to change their behaviour. But RJ will provide an opportunity for change or for the victim to decide if she wants to move on. (Judge)

It is often an appropriate forum for confronting the offender and discussing frankly with the offender the need for his behaviour to change. It's more personal, more time is given. The Court is full on. (Judge)

It doesn't work in all areas or for everybody. That's where the facilitator/coordinator has to have a good understanding of the complexities of family violence to make it a success. It's not something you can teach people in a short amount of time. It's something learned with experience. Courts are often quite removed from what is happening in the community at a local level. (Family Violence Coordinator)

[It's appropriate for] certain offences and offenders. I've been quite impressed regarding victim and offender interactions. Offenders have demonstrated that they're wanting to make amends in certain cases. (Police prosecutor)

The victims don't get a say in Court. RJ could be held on a marae and both families present. I would use it post sentencing, as I think we need to have the Justice Department as a back-up. In [X] it is an under-utilised process. [X] have good facilitators and a good venue, but it is not being used. You need to get buy-in and ownership from agencies such as Police CYFS, MSD Work & Income for it to work properly. The Tenancy tribunal facilitates and good things come out of it. The same could be happening with family violence cases. It gives the opportunity for the victims to have a voice. (Police)

Opponents of the use of restorative justice processes in family violence cases returned to their view that a one-off meeting was of no use. To quote two informants:

I would like to see a different system than restorative justice used in family violence. It's far too complex for a one-off meeting. At sentencing, restorative justice is seen as a mitigating factor, always as a good thing, regardless of the offender's motivation. (Victim Adviser)

You need to marry up healing for the victim and the victim's family with strong consequences for the offender. A one-off meeting is of no use. Perhaps a women's education programme through which a woman might get a clearer understanding of the dynamics of family violence might be of use. (Service provider)

One opponent to the use of restorative justice for family violence cases was of the view that, because of the dynamics of family violence, all family violence offenders should be sentenced to supervision with special conditions.

## 6.3 Summary

In conclusion, the 24 key informants from stakeholder groups including violence intervention programme providers, Victim Advisers, Judges, Police and lawyers across the five sites, were divided in their views as to the appropriateness of using restorative justice processes with victims and offenders in family violence cases.

- 33% expressed unconditional support
- 38% expressed conditional support
- 29% were against the use of restorative justice for family violence cases

Key informants stated that the use of restorative justice for family violence cases was appropriate for less serious offences with offenders who showed remorse and accepted responsibility for their offending and in situations where there was likely to be an ongoing relationship between the victim and offender. They did not consider restorative justice appropriate for rape and other sexual offences, serious assaults, or in situations where the victim could be coerced by the offender.

Thirty-eight percent of key informants considered that they did not know enough about the restorative justice processes used by their local programme when dealing with family violence cases to be able to comment on their performance. However, key informants who were familiar with their local programme gave them the following ratings:

- 46% working well or very well
- 4% neither well nor poorly
- 12% poorly or very poorly

Opponents of the use of restorative justice processes in family violence cases continually referred to the futility of a one-off meeting. However, supporters of the use of restorative justice processes in family violence cases had a more realistic view of what these processes might achieve. Participation in restorative justice processes could provide some victims and offenders with an opportunity to come together in a less pressured environment than the Court. It could allow for victims to have a voice and for offenders to start to address their offending. This is particularly important if the relationship is ongoing.

## Chapter 7 Summary and discussion

## 7.1 Meeting the objectives of the review

Following extensive consultation process<sup>85</sup> the Ministry of Justice published a set of principles of best practice identifying how and when restorative justice processes should be used in criminal cases (Ministry of Justice, 2004). The principles are intended to be used as a resource for restorative justice providers to help 'protect the inherent flexibility of restorative justice processes [and to] provide clear guidance about the use of these processes in a safe and appropriate way' (Ministry of Justice, 2004; p 10). They are not prescriptive and thus accommodate a range of approaches to restorative justice in New Zealand. They are therefore, applicable to the varied approaches utilised by the five sites that were the focus of this review. At two of the sites, Levin and Nelson, the restorative justice process used a community panel pre-sentence model which involves two panel members and a coordinator/facilitator. At another two other sites, Nga Whare Waatea and Christchurch, the restorative justice process is a victim-offender meeting with a facilitator and a cofacilitator. At Nga Whare Waatea the process takes place either pre or post-sentencing. In the case of Christchurch the restorative justice process can take place pre or post sentencing or where there is no formal charge. At the fifth site, Timaru, the process is one of community panel pre-trial diversion and involves the Police, panel members and a facilitator.

Principle 8 states that: Restorative justice processes should only be undertaken in appropriate cases. In effect this principle prescribes that the use of a restorative justice process in a particular case must be carefully considered. And, very careful consideration should to given to the use of restorative justice processes in cases of family violence and sexual violence. The objectives of the review were to describe the nature and extent of the delivery of restorative justice in family violence cases including the following:

- how selection of cases occurs;
- assessment of appropriateness of each referral;
- how the consent of participants is obtained;
- detail of the restorative justice processes;
- safety of participants;
- the outcomes for individuals (both victim and offender) from the processes e.g. participation in therapeutic programmes, sentencing outcomes, etc); and
- the experience of individuals who have participated in restorative justice processes in family violence or domestic violence cases.

The views of individuals who have participated in restorative justice processes in family violence cases and the views of key informants are integrated throughout.

### Selection of cases

The selection of family violence cases for restorative justice processes occurs at several levels. At the referral level, Judges, lawyers, Police, and some other criminal justice representatives, may

suggest to the Judge that the offender might be a suitable candidate for restorative justice. At the restorative justice provider level, it is usually the coordinator who decides whether to proceed with the use of restorative processes by typically considering the types of offences the provider accepts, the suitability of the victim and the offender to participate (or consent to the meeting proceeding), the skills of the facilitator and the facilitator's ability to hold a meeting in a supportive and safe environment.

Victims and offenders reported that the suggestion of dealing with the offending through restorative justice processes came from a variety of sources. The suggestion often came from the offender's lawyer (ten offenders and four victims reported this) although in some cases the victim made the suggestion (two offenders and one victim reported this). Two victims said the Court Victim Adviser had suggested restorative justice. An identical number of offenders said that the restorative justice provider was at Court and had been the one to suggest that their case be dealt with in this fashion. A few (three victims and one offender) said that they thought that the Judge had suggested restorative justice.

It was usually the restorative justice provider who had first contact with offenders. Eighty-three percent of offenders reported this. The remainder (17%) said that their first point of contact was their lawyer. Court Victim Advisers usually had the first contact with victims. More than one half (or 58%) of the victims who took part in restorative justice panels/meetings reported this. The remainder (42%) said they had initially been contacted by the restorative justice provider.

Almost three-quarters (17 or 71%) of key informants interviewed were in favour of the use of restorative justice processes in family violence cases, although over half of this number added caveats to their support such as the importance of facilitators skilled in the dynamics of family violence relationships and the need for clear safety protocols/guidelines. Some identified a range of circumstances as being particularly appropriate for the use of restorative justice in family violence cases, these included:

- where both parties agree to the process;
- where healing is possible for the victim;
- if the victim wants to stay in relationship;
- where a bond exists between victim and offender;
- where couples need to work through issues;
- where families want to stay together;
- where the victim and offender have children together; and
- where there is no physical violence.

Key informants did not view restorative justice as a cure-all in all family violence cases, but as an opportunity for some victims and offenders of family violence to come together in a less pressured environment than the Court for victims to have a voice and for offenders to start to address their offending. This was particularly important because, as some key informants commented, in the vast majority of cases the relationship is going to carry on.

#### Assessment of appropriateness of each referral

The coordinator/facilitator assesses, on a case by case basis, the appropriateness of the offender and the victim to participate in restorative justice processes. They based their decision making on many factors relating to the victim, the offender, and the facilitator (see, especially section 3.2 suitability of referrals). Restorative justice meetings were more likely to be held for family violence cases in circumstances where the victim wanted to continue in the relationship with the offender.

Victims and offenders were asked why they had decided to take part in the restorative justice process. Some common themes emerged from what they said:

- victims and offenders wanted to sort out the issues that had led to the offending;
- victims and offenders wanted to be able to discuss their differences in a safe environment;
- some victims wanted the offender to get help to change;
- offenders wanted to take responsibility, help themselves and/or make amends to the victim;
   and
- those who were no longer in the relationship, wanted closure.

## Consent of participants

The offender's verbal consent to participate in restorative justice processes is usually obtained at a pre-meeting, as is that of the victim once the provider has explained their processes. Written consents are not usually obtained by coordinators/facilitators at the five sites.

Victims and offenders said that they were comfortable being contacted about their potential participation in the restorative justice process. Only a few (two victims and one offender) said they had felt indirectly pressured to take part in that they were unsure of what other options were available if they refused.

#### Detail of the restorative justice processes

At the five sites that were the main focus of this review, the restorative justice processes used by providers for family violence cases did not appear to differ significantly from the processes they would usually use to deal with other types of cases. At some sites, such as Nelson, the victim must agree to be present at the meeting whereas at others, such as Timaru, this is not a requirement. It was a common theme for offenders to offer and apology at the meeting, although they were not formally required to do so. Given the usual long history of the offending, providers were of the view that apologies would in any case likely to have been made from time to time by the offenders to the victims, and to require an apology by the offender in a one-off meeting might serve to minimise the offending or result in less than a sincere apology being given.

Most victims and offenders had expectations from participating in the restorative justice process. These included:

- to talk about what had happened incorporating both points of view;
- for the offender to realise the impact of what he had done and be made accountable;

- to come to some resolution over the offending repair the harm; and
- to help the offender and victim so that this would not happen again.

Victims and offenders generally understood what was happening and felt involved, they felt they had been treated with respect and had the opportunity to say what they wanted. Most offenders (89%) had been given the opportunity to explain their actions. Similarly the majority of victims (84%) had been able to make clear the effects of the offending on them

Victims and offenders had been able to meet in a safe, non judgemental environment and talk about the effects of the offending. In general, levels of satisfaction with the process and agreements reached were high, although offenders were consistently more likely than victims to report this

A few victims (n=4) and offenders (n=3) said that it was important that the restorative justice process meet their cultural needs. Only one offender said his/her needs had not been met.

Most participants found involvement in the restorative justice process a positive experience and were pleased that they had taken part.

Key informants who provided an opinion also generally felt that the local restorative justice programme was effective:

- in involving the victim (11 out of 15);
- in holding the offender accountable (8 out of 14);
- in addressing the needs of the offender in relation to his/her violent behaviour (8 out of 14);
   and
- in meeting the needs of the victim (7 out of 13) and offender (11 out of 14) during the restorative justice process.

Eleven out of 16 key informants also felt that the local restorative justice programme was effective in meeting the needs of the community when dealing with family violence cases.

#### Safety of participants

The safety of participants at the restorative justice meetings is of paramount concern to the coordinators/facilitators. The providers' consideration of safety issues begins with their assessment of the suitability to take part in restorative justice of the offenders and victims who cases are referred. If the coordinator/facilitator has any safety concerns the meeting will not go ahead. At the meeting, additional safety elements at Timaru include the presence of a Police representative, and at Nelson include the mandatory presence of support people for the victim and offender. In Christchurch, all participants are required to sign an agreement saying that they will abide by the rules set down for the meeting and there is a specific protocol setting out the areas to be discussed with family violence victims and offenders prior to the restorative justice process taking place (see Appendix B4). The facilitator's experience and ability is crucial when dealing with family violence cases. If the victim says she or he feels unsafe, the facilitator will stop the meeting, and the offender will be returned to the Court to be dealt with in the usual way.

In preparation for the restorative justice meeting, approximately a third of victims (n=6) and a small number of offenders (n=3) said that they had been asked by the coordinator/facilitator whether or not

they had any concerns for their safety if they participated. More than two-fifths of offenders said that they had been informed of the safety rules under which the meeting would be conducted - zero tolerance of abuse of any type. There was a clear indication that the meeting would stop if anyone breached these conditions. A small number of victims (n=3) said there had been a direct reference to how they would be kept safe during the meeting and how they could take a break or leave at any time if they felt things were getting out of hand. Not all victims appeared to be concerned about their safety – most assumed that they would be safe during the restorative justice process due to the general rules set down for the meeting.

Although most victims and offenders reported being apprehensive and feeling some anxiety when they first went into the meeting by the end they were feeling relieved and positive about the experience.

Few victims or offenders said that they were too scared to say what they really felt during the meeting. Where participants had support at the meeting (50% of offenders and 37% of victims) they found this useful.

Only two victims said they felt safer as a result of taking part in the restorative justice process. However, most said that this was not an issue for them: they had either ended the relationship or they had never felt unsafe.

Eleven key informants rated their local restorative justice programme in terms of its effectiveness in keeping the victim safe during the meeting. Almost two-thirds (64%) felt that the programme was managing to do this effectively. Similarly, of those key informants (n=10) who rated the programme in terms of its effectiveness in addressing the ongoing safety of the victim one half felt that the programme achieved this in an effective manner.

No key informants specifically mentioned having any concerns about the safety of participants when asked to identify the strengths and weaknesses of their local restorative justice programme.

#### Agreements reached at meetings

During 2005/2006 all five programmes used restorative justice processes for the offences of male assaults female and assault/common assault. The restorative justice meetings held for family violence cases in Horowhenua, Nelson and Timaru resulted in agreements which offenders are required to complete before the disposition of their case. In the case of Horowhenua and Nelson a report is provided for the sentencing Judge, who may choose to incorporate all or part of the agreement into the offender's sentence. In community panel diversion cases- ie Timaru – a report goes to Police Prosecutions who may withdraw charges if the agreement is fulfilled by the offender.

Typically, these agreements commonly included the offender's attendance at a violence intervention programme. Written apologies or community work featured less frequently. Although at Levin community work was invariably a feature of the agreement. An additional feature of Nelson agreements was that they usually included a fun element or social activity.

At some sites, offenders must pay for their programme attendance whereas at others they are subsidised or free of charge.

Although most participants interviewed said that they were satisfied with the agreement reached at their restorative justice meeting, offenders were more likely to say this than victims (93% compared with 79%).

Offenders (68%) were also more likely than victims (32%) to say that they had understood how the victim felt about the offending, yet most victims (63%) felt that the offender had been made accountable for their actions. Despite this almost one half of victims felt that the offender had not been able to 'repair the harm' or make up for what they had done and a few felt that the offender had got off too lightly.

Approximately two-fifths (or 42%) of key informants had an opinion on the nature of agreements reached at the meetings held in their area. These opinions were fairly evenly divided. Three thought that the agreements were soft; four thought they were about right and three thought they were harsh.

The principles of the Sentencing Act 2002 require the Judge to take into account any outcomes of restorative processes. Sentencing outcomes of offenders of family violence who have participated in restorative justice processes at the five sites and who have completed their agreements, have included s106 (Sentencing Act 2002) discharges without conviction, s108 (Sentencing Act 2002) convictions and discharges and s110 (Sentencing Act 2002) orders to come up for sentence if called upon.

Most of the offenders (11 out of 13) interviewed who had been sentenced said that the Judge had referred positively to the reports from the restorative justice meetings provided by programme facilitators.

Twelve key informants rated their local restorative justice programme in terms of meeting the needs of the Judges. Three quarters of them gave a rating of 4 ('effective') or 5 ('very effective'). Of the four District Court Judges interviewed, three gave ratings: one rated the local programme as 'very effective', one as 'effective' and one as 'ineffective' in meeting their needs.

#### Experiences of participants

The good features of the restorative justice meeting most commonly reported by victims were the open dialogue and the healing process that occurred in a non-judgemental environment. They also appreciated being able to meet the offender in a safe supportive environment. Good features of the meeting most often referred to by offenders included: being able to put things right, feeling supported and being treated with respect. Few victims or offenders had anything bad to say about the meeting. Offenders (89%) were more likely than victims (74%) to be satisfied overall with the meeting.

A few victims (21%) and offenders (16%) felt that it was important that the restorative justice meeting meet their cultural needs. All except one reported that this had happened.

Victims' views were mixed on whether or not the violence or abusive behaviour had stopped since the restorative justice meeting. Although a third said it had another third said that the violence had just changed in nature: from physical to psychological violence. However, more than one half of offenders said that they had learnt something about their behaviour and its effects on others and thought that they needed to change.

Nonetheless, after the meeting victims were more likely than offenders to think that the offenders still needed help to address their violent behaviour. Whereas most offenders said that they had got help or were involved in the process of getting it victims still thought this was an issue.

There was less agreement between victims and offenders as to whether the offender had understood the effect of the offending on the victim and whether or not the offender had been able to

make amends for what he/she had done. Victims were more likely than offenders to respond in the negative.

All offenders and most victims said that they would recommend restorative justice processes to other victims and offenders to deal with family violence. And approximately 80% of victims said that they would take part in the process again should they need to. All participants generally agreed that restorative justice meetings were a good way to deal with family violence, although some qualified this by saying it depended on those involved and the circumstances relating to the violence.

#### 7.2 In conclusion

86

Family violence offences span the continuum from less serious offences (such as wilful damage) to the most serious offences (such as rape and murder). Based on the findings of this review we think that the use of restorative justice processes in family violence cases is appropriate for the types of cases currently being accepted by providers at the five sites. That is, mostly offences in the less serious to medium seriousness range such as male assaults female offences that have not required the victim's admission to hospital.

The dynamics of family violence are complex and the nature of the close and often ongoing relationship of the victim and offender further complicates the issue. Family violence offences that are to be addressed through a restorative justice meeting are likely to be only a small proportion of the many offences the victim has endured at the hands of the offender. Opponents of the use of restorative justice in family cases point to a one-off meeting as being too simplistic to deal with such cases, and that victims have a high chance of being re-victimised and unsafe at these meetings. 80 Conversely, advocates argue that restorative justice meetings, do, however, (with careful selection, planning and skill by facilitators) provide a forum for addressing victims' needs (especially where they are present) in a safe environment. And that such processes provide 'a way of looking at the wrongdoing that emphasises repair of harm and accountability of the wrong doer' (Zehr, 2007), something conventional court processes are unable to do. The findings of this review concur, and indicate that restorative justice meetings for couples who want to continue their relationships, combined with offenders' participation in violence intervention programmes, may lead to positive change for some couples. Similarly, restorative justice meetings may enable couples who are no longer together to gain some resolution and end the relationship on amicable terms. This may be particularly relevant where there are children involved.

Only one of the five sites included in this review had specific guidelines on issues that should be discussed for the use of restorative justice in family violence cases. The other four providers felt that their generic restorative justice protocols based on the Ministry's principles of best practice were sufficient for them to accept referrals of family violence cases on a case by case basis. Particular care was taken by providers at all sites in the preparation of the victim and offender, the selection of appropriate panel members, the safety of participants and the development of appropriate agreements.

We consider that restorative justice processes, as one of a suite of potential responses to family violence, need to continue to be offered. Therefore, there may be value in the Ministry of Justice developing specific guidelines for the use of restorative justice processes in family violence cases to

See for example Daly (2002) who refers to family violence as a 'gendered harm'. Stubbs (2007) agrees and describes the pitfalls of using generic restorative justice processes to deal with family violence.

assist providers operating in an area characterised by patterns of violent behaviour.	complex relationships and often entrenched

## References

- Anae, M., Coxon, E., Mara, D., Wendt-Samu, T. and Finau, C. (2003). *Pasifika Education Research Guidelines*. www.minedu.govt.nz.
- Australian Bureau of Statistics. (2005). *Australian and New Zealand Standard Classification of Occupations Information Paper.* Canberra: Australian Bureau of Statistics.
- Braithwaite, J. (1989). Crime, Shame and Reintegration. Cambridge: Cambridge University Press.
- Braithwaite, J. and Daly, K. (1994). 'Masculinities, violence and communitarian control.' In, T. Newburn and E. Stanko (Eds.), *Just Boys Doing Business? Men, Masculinities and Crime*. London: Routledge.
- Busch, R. (2002). 'Family violence and restorative justice initiatives: who pays if we get it wrong? 'In, H. Strang and J. Braithwaite (Eds.), *Restorative Justice and Family Violence*. Cambridge: Cambridge University Press.
- Carbonatto, H. (1995). Expanding Intervention Options for Spousal Abuse: the use of restorative justice.

  Occasional Papers in Criminology. New Series No. 4. Wellington: Institute of Criminology, Victoria University of Wellington.
- Crime and Justice Research Centre. (2005). New Zealand Court-Referred Restorative Justice Pilot: Evaluation. Report prepared by the Crime and Justice Research Centre with Sue Triggs of the Ministry of Justice. Wellington: Ministry of Justice.
- Cunningham, C. (2000). 'A framework for addressing Māori knowledge in research, science and technology.' *Pacific Health Dialogue*, 7 (1), pp. 62-69
- Daly, K. (2006). 'Restorative Justice and Sexual Assault: an archival study of court and conference cases.' *British Journal of Criminology, 46* (2), pp. 334-356.
- Daly, K. (2002). 'Sexual Assault and Restorative Justice'. In H. Strang and J. Braithwaite (Eds), Restorative Justice and Family Violence. Cambridge: Cambridge University Press.
- Department for Courts. (2002). Facilitator Training Manual. Department for Courts Restorative Justice Pilot Facilitator Training. Wellington: Department for Courts.
- Hudson, B. (2003). 'Victims and offenders'. In, A. von Hirsch, J. Roberts and A. Bottoms (Eds.), Restorative Justice and Criminal Justice: competing or reconcilable paradigms? Oxford: Hart Publishing.
- Kingi, V and Poppelwell, E. (2005). *The Viewing Habits of Users of Sexually Explicit Movies: a Hawke's Bay sample.* Wellington: Office of Film & Literature Classification.
- Lewis, R., Dobash, R.E, Dobash, R.P. and Cavanagh, K. (2001). 'Law's progressive potential': the value of engagement with the law for family violence'. *Social and Legal Studies*, *10* (1), pp. 105-30.
- Maxwell, G., Kingi, V., Robertson, J., Morris, A., and Cunningham, C. (2004). *Achieving Effective Outcome in Youth Justice: Final Report*. Wellington: Ministry of Social Development.
- Maxwell G, Morris A, Anderson T. (1999) *Community Panel Adult Pre-Trial Diversion: Supplementary Evaluation.* Wellington: Crime Prevention Unit, Department of Prime Minister and Cabinet.
- Ministry of Justice. (2004). *Restorative Justice in New Zealand: Best Practice*. Wellington: Ministry of Justice.
- Morris, A. and Gelsthorpe, L. (2000). 'Re-visioning men's violence against female partners'. *Howard Journal of Criminal Justice*, 39, pp. 412-28.

- Oppenheim, A.N. (1992). Questionnaire Design, Interviewing and Attitude Measurement, New Edition (reprinted 1997). London: Pinter.
- Paulin, J., Kingi, V. and Huirama, T. (2005). *The Rotorua Second Chance Community-Managed Restorative Justice Programme: An Evaluation*. Wellington: Ministry of Justice.
- Strang, H. and Braithwaite, J. (Eds.) (2002). *Restorative Justice and Family Violence*. Cambridge: Cambridge University Press.
- Stubbs, J. (2007). Beyond apology? Domestic violence and critical questions for restorative justice. Criminology & Criminal Justice, 7 (2), pp. 169-87. Stubbs, J. (2004). Restorative Justice, Family Violence and Family Violence. Issues Paper No. 9. Australian Domestic and Family Violence Clearinghouse. Sydney: University of New South Wales.
- Stubbs, J. (2002). 'Family violence and women's safety: feminist challenges to restorative justice'. In, H. Strang and J. Braithwaite (Eds.), *Restorative Justice and Family Violence*. Cambridge: Cambridge University Press.
- Te Puni Kokiri, Ministry of Māori Development (1999). Evaluation for Māori: guidelines for government agencies. Wellington: Ministry of Māori Development.
- Wilde, V. and Baker, G. (2006). *Ministry of Justice Restorative Justice Programmes Stock take: CPU-funded community-based restorative justice programmes*. Report prepared for the Ministry of Justice. Wellington: Ministry of Justice.
- Zehr, Howard. (2007). Hearing Victims' Voices Is Restorative Justice Listening. Seminar given on 17<sup>th</sup> August 2007 at the Wellesley Club, Wellington.

# Appendix A Supplementary tables

Table A2.1: Some characteristics of restorative justice meetings involving family violence cases, and estimates of number of such meetings

RJ Provider	Location	Type of meeting	Percent of r j meetings where victims present	Number of rj meetings held over 6 months	Percent of rj referrals that involve family violence	Estimate of number of rj meetings that involve family violence over 6 months	Estimate of number of rj meetings that involve family violence over 3 month review period
Te Whānau Awhina	Waitakere	panel	6	43	11-25	8	4
Horowhenua RJ Programme	Levin	panel	50	13	51-75	8	4
Project Turnaround	Timaru	panel	67	33	11-25	6	3
Turnaround Ashburton	Ashburton	panel	49	29	11-25	5	3
Nelson RJ Services (Just Steps)	Nelson	panel (with victim)	63	27	11-25	5	3
Turnaround Marlborough	Blenheim	panel	100	7	51-75	4	2
Tauranga Moana RJ Trust	Tauranga	victim-offender	No data	9	26-50	3	2
RJ Services	Christchurch	victim-offender	83	8	11-25	1	1
Taranaki RJ	New Plymouth	victim-offender	100	18	1-10	1	1
Nga Whare Waatea Trust	Counties/Manukau	victim-offender	100	11	1-10	1	1
Hawkes Bay RJ Te Puna Wai Ora	Hastings/Napier	victim-offender	100	7	1-10	<1	<1
RJ Services	Wellington	victim-offender	100	5	1-10	<1	<1
Court -RJ Services Hauraki	Paeroa	victim-offender	No data	11	1-10	<1	<1
Waitaki Turnaround	Oamaru	both	35	38	1-10	2	1
West Coast RJ Programme	Greymouth	both	77	12	1-10	<1	<1

#### Notes:

- 1 RJ Services, Wellington, do not accept cases involving partners or cases involving sexual violence.
- 2 All groups bar Te Whānau Awhina require victim consent before the case goes to conference/panel.
- Numbers of rj meetings involving family violence were estimated by multiplying the number of rj meetings by the midpoint of the percentage of rj meetings involving family violence.

Appendix B Information sheets, consent forms, and interview schedules

#### B1 Information sheets and consent forms

# VICTORIA UNIVERSITY OF WELLINGTON Te Whare Wānanga o te Ūpoko o te Ika a Māui



### Research about restorative justice meetings

#### Greetings

(Restorative justice provider) hold restorative justice meetings in your area for victims and offenders of family violence. We have been asked by the Ministry of Justice to find out more about these.

We are inviting all victims and offenders who have been to one of these restorative justice meetings recently to take part in this research.

We would like to talk to you for about 30 minutes to see how you felt about the meeting and what happened there. We would also like to know how things have been for you since then.

You do not need to talk to us unless you want to. However, if you decide to do so, we will arrange a time and place to suit you. You are welcome to bring someone to support you if you would like. We will keep what you tell us confidential and only use this for research purposes. Your name will not appear in any research reports and nothing will be published that might identify you. If you don't want to answer some of our questions, you don't have to and you can end the interview at any time.

If you have any questions about the research, you can ring the researchers, Venezia Kingi or Judy Paulin on 0800 586 987.

Warmest regards from the research team

Venezia Kingi, Judy Paulin and Laurie Porima

Reply slip							
If you do <b>NOT</b> wish to take part in the research, please tick the box enclosed reply paid envelope. Otherwise, a researcher will get in to you. She/he will tell you more about the project then.							
I do <b>NOT</b> wish to take part	[ ]						
Signature:							
Name (please print):							

#### VICTORIA UNIVERSITY OF WELLINGTON

Te Whare Wānanga o te Ūpoko o te Ika a Māui



## Restorative Justice Research

### **Consent Form**

I,, agree to be interviewed for this research
study.
The researcher has explained to me the purpose of the research, and my right to not answer any question
I don't like or to stop the interview, without having to explain why.
Table time of the stop and more man, maneut naturing to explain imp.
I understand that all information will be kept confidential by the researchers, and will be used only for
research purposes. My name will not be used in any research reports and nothing will be published that
might identify me.
I understand that if the researchers thought that I or someone else was at risk of serious harm they may
have to break confidentiality. Should they need to they will discuss this with me first.
Signed Date

Ring us or write for more information Phone: 0800 586 987 – Venezia, Judy or Laurie Victoria University, PO Box 600, Wellington

# B1 Ministry of Justice survey

# QUESTIONNAIRE FOR RESTORATIVE JUSTICE PROVIDERS IN RELATION TO FAMILY VIOLENCE CASES.

Name of restorative justice provider
Address of provider
Do you accept referrals for restorative justice cases involving family violence? Yes/No If you do not deal with family violence cases please go to question 17.
Do you use a Victim-Offender process or a Panel process for Family Violence cases?  Please circle the correct answer  Victim-Offender Panel
What percentage of the cases referred to you involve family violence offences?  Please circle the correct answer  1 to 10
What types of family violence offending have been referred to your organisation (over the last year)? Tick all that apply Physical assaults - no medical attention needed Physical assaults - causing injury treated by doctor or nurse Physical assaults - required admission to hospital Threats Damage to property
Sexual assaults adult children Sexual violation (rape, unlawful sexual connection) adult children
Please list the agencies or organisations which refer family violence cases to you for restorative justice e.g. the Court, police, churches, lawyers, and community groups.
5. Are family violence referrals from the Court generally made before or after sentencing?  Please circle the correct answer  Before After Don't get referrals from court  6. What percentage of your family violence cases you get referred would result in a conference or panel meeting?  a) General family violence  b) Family violence involving sexual offending  """  """  """  """  """  """  """
7. Do you require victim consent before the case goes to conference/panel? Yes/No How do you obtain the victim's consent?

8. How do you assess the suitability of the family violence cases referred to you for restorative justice?
Victims
Offers days
Offenders
9. What mechanisms/processes do you have in place to ensure safety for?
Victims
Offenders
Offenders
10. What relevant specialist training/experience do your facilitators of family violence cases have? Please list
11. What skills and knowledge does your agency have to work appropriately cross culturally?
12. Is your organisation an accredited family violence service provider to any organisations?
Yes/No
If yes, please describe your accreditation and the services you provide
12. Do conference/namel plane/agreements usually recommend that the effender he referred to other
13. Do conference/panel plans/agreements usually recommend that the offender be referred to other programmes or services?Yes/No
14. Please list the programmes or services that are usually recommended in plans
15. Please describe how the plans/agreements are usually monitored and the estions taken if they are not
15. Please describe how the plans/agreements are usually monitored and the actions taken if they are not complied with.

16. How do you obtain feedback on the services you provide from conference/panel participants (and/or other stakeholders)?
If you use a questionnaire or evaluation sheet please attach a copy.
17. Do you intend in the future to accept referrals of family violence cases for restorative justice? Yes/No
Completed by:
Date
Thank you very much for taking the time to complete and return this questionnaire.
Julia Hennessy Senior Policy Advisor - Restorative Justice Crime Prevention Unit – Ministry of Justice

# **Restorative Justice in Family Violence Cases**

## Questionnaire: Victim attending conference

		e to ask you about what happ	•	
		(month)		
with	respect to			_ (offences)
The	aim is to find out how you	u felt about the whole process	S.	
	TE: Coding - Throughout question)	use: 8=not applicable; 9=don	't know; 99=missing data; 98=	edid not want to answe
Cho	oosing to have a confere	nce/meeting		
<b>1</b> (Co	Whose idea was it to hav	re the restorative justice confe oded only, not asked)	erence/meeting?	
	Victim's (self)			
	Offender's			
	Offender's lawyer			
	Police			
	Judge			
	Victim adviser			
	RJ provider			
	Other (specify)			
2	Who contacted you first justice conference/meeti	about the possibility of dealin ng? ( <i>Tick which</i> )	g with the offending at a rest	orative
	Victim adviser			
	RJ provider			
	Other (specify)			
3		peing contacted about dealing ease out issues around breach		
4		o to the conference/meeting? stion record the answer and the		
	To tell the offender what	the offence was like for me		
	To get an apology from t	the offender for what he/she	did	
	To have a say			
	To work out an agreeme	nt on how to deal with the off	ence	
	To have the offender pay	y reparation		
	I thought it was a good i	dea		
	To get closure			
	Friends/family thought it	would be a good idea		
	Cultural reasons			
	The Judge suggested it			
	I didn't feel like I could re	efuse to go		

	Oth	er <i>(elaborate)</i>									
5		you feel pressur es, by whom and						he c	conference? (Code 1=Yes or 2=No)		
6		On a scale from <b>1 to 7</b> – where 1=No impact and 7=Very high impact, rate the impact of the incident on you at the time it happened									
		(Circ	le a i	numb	er on	the f	ollowi	ng s	scale)		
		1	2	3	4	5	6	7			
		No · ·							Very		
		impact							high impact		
7	con	ference/meeting	g? <i>(</i> 7						ee or speak to (the offender) before whether pressured by offender to atte		1
	•	de 1=Yes or 2=\n ' <b>es</b> , what happer	,								l
Pre	para	tion									
8	a)	Who contacted	d you	ı abc	out the	e arra	ingen	nent	ts for the conference/meeting?		
9	<b>b)</b>	How did they one you asked ab		ict yo	ou?					(Code 1=Yes or 2	2=No)
9		•		1				L - L			
	a)	Where the cor	itere	nce/r	neeti	ng sn	iouia	be r	neia?		
	b)	When the conf									
	c)	Who should be	e at t	he c	onfer	ence/	meet	ing?	<b>)</b>		
	d)	_			any	conc	erns a	abou	ut your safety if you attended?		
	(On	ly ask e) if an IP	V cas	se)							
	e)	Whether or no	t the	re wa	as a d	currer	nt pro	tecti	ion order in place?		
	f)	Whether or no	t the	re we	ere ba	ail co	nditio	ns ir	n place?		
10	Wei	re you <b>told</b> :									
	a)	In plenty of tir	ne al	oout	wher	the o	confe	renc	ce/meeting was to be held?		
	b)	What would ha	appe	n at t	the co	onfere	ence/	mee	eting?		
	c)	What you wou	ld ha	ive to	o do a	at the	confe	eren	nce/meeting?		
	d)	Who else wou	ld be	at th	ne co	nfere	nce/n	neet	ting?		
	e)	How you would	d be	kept	safe	while	at th	ie co	onference/meeting?		
	f)	How you would	d be	kept	safe	arrivi	ng at	and	d leaving the conference/meeting?		
	W	hat were you to	ld?								
11	(Coa	nis stage did any de 1=Yes or 2=No 'es, who made t	o)		•	•			ns of the sorts of things that could g	o in the agreemen	t? ]

12							II prepar	ed and 7	'=Very	y well prep	ared - how prepai	red did	I
	you feel for the conference/meeting?  (Circle a number on the following scale)												
						_	_	•	_	<del>,</del>			
		1	-11	2	3	4	5	6	7				
		Not at								Very well			
		prepar (elabor		toose o	ut issues	around v	vhothor v	would hav	ıa lika	prepare			
	(elaborate – tease out issues around whether would have liked more information)												
The Restorative Justice Conference/Meeting													
13	Did	some	one g	go the co	onference	e/meeting	g with yo	ou to sup	port y	ou?			
		a) If	f Vac	s who w	as this (	snecify)					(Code 1=Yes, 2	2=No)	
	b) Was their support useful? (Code 1=Yes, 2=No or 3= Partly)												
		c)	Wou	ld you h	ave liked	l more su	upport at	t the con	ferenc	ce/meeting	?		
										(Code 1=	Yes, 2=No or 3= F	Partly)	
		If Yes	s, fro	m who?	<b>)</b>								
14	Wh	at do y	ou th	nink the	conferen	ice/meet	ing was	trying to	achie	ve?			
15	Hov	v did y	ou fe	eel wher	n you first	t went int	to the co	nference	e/mee	ting?			
	(Pr	ompt fo	or fee	lings are	ound first	meeting	the offer	nder' ther	e)				
16	Dur	ing the	e con	ıference	/meeting	did you:							
	a)	Feel			J	·				(Code 1=	Yes, 2=No or 3= F	Partly)	
	b)	Unde	rstar	nd what	was goin	ıq on?				(Code 1=	=Yes, 2=No or 3= F	Partly)	
	c)				to explaii		e offenc	o affecto	d vou	•	,	• •	
	c)	Tiave	uic	CHARICE	ιο εχριαιί	THOW the	e oneno	e anecie	u you		=Yes, 2=No or 3=F	Partly)	
										(0000 7	- 103, <u>2</u> -110 01 0-1	uruy)	
	d)	Have	the	opportu	nity to sa	y what y	ou want	ed to say	<i>i</i> ?	(Code 1=	Yes, 2=No or 3= F	Partly)	
	e)	Feel	too s	cared to	say wha	at you rea	ally felt?			(Code 1=	Yes, 2=No or 3= F	Partly)	
17	Do	you thi	nk aı	nyone s	aid too m	nuch duri	ng the c	onferenc	e/me	eting?	(Code 1=Yes, 2	2=No)	
	a)	If Yes	s, wh	10? <i>(Doi</i>	n't ask - tio	ck all that	t apply)						
	Р	anel m	nemb	ers									
	٧	ictim's	sup	port (e.g	g. family c	r whanau	u)						Ш
	٧	ictim's	repr	esentat	ive								
		ictim (	-										
		offende											
				upport (	e.g. famil	y or whai	nau)						
		acilitat		(f <sub>1</sub> , )									H
46		other (s	-										
18		-			e/meeting					(0 ) 1	V 0 M	<b>1</b>	
	a)	vvere	you	treated	with resp	pect?				(Code 1=	Yes, 2=No or 3= F	artiy)	

(Be alert to cultural issues)

	b) If No or P	artly to	a) above	- why d	o you thi	nk that wa	s?		
	c) Did you fe	el unsaf	e (either	physica	lly/emotion	onally) at a	any ti	ime?(Code 1=Yes, 2=No 3= Partly)	
	i) If Yes o	r partly,	did you	say any	thing abo	out this at t	the ti	me?	
	ii) If Yes, h	now were	e your sa	fety con	cerns de	alt with?			
	d) Did people	e in the	conference	e/meeti	ng speal	k up on yo	ur be	ehalf?	
								(Code 1=Yes, 2=No or 3= Partly)	
19	During the con	ference/	meeting o	did (the	offender,	)			
	a) Apologise	for wha	t s/he did	?				(Code 1=Yes, 2=No or 3= Partly)	
	b) If Yes did	you acc	ept (the o	offender	's) apolo	gy?		(Code 1=Yes, 2=No or 3= Partly)	
20	Do you think (to	he offen	<i>der</i> ) unde	erstood I	how you	felt?		(Code 1=Yes, 2=No or 3= Partly)	
21	Was (the offen	der):							
	a) Able to ma	ake up f	or what s	he did?	•			(Code 1=Yes, 2=No or 3= Partly)	
	b) Made acc	ountable	for his/h	er offen	ding?			(Code 1=Yes, 2=No or 3= Partly)	
22	How would you	ı describ	e (the off	ender's	) behavio	our during	the c	conference?	
The	e conference ag	reemen	t						
23	Was an agreer  a) If Not, wh		ched?					(Code 1=Yes, 2=No)	
24	Who decided of All of us Victim (self) Victim's support Victim's represe Offender Offender's supp Panel member Facilitator Other (specify)	ter/family entative	y	e agree	ment?			(Tick either option 1 <u>OR</u> all that a	pply)
25	Can you remer	nber wh	at was in	the agr	eement?				
								(Code 1=Yes, 2=No or 3= Partly)	Ш
26	Did you unders	stand wh	at was in	the agr	eement?			(Code 1=Yes, 2=No or 3= Partly)	
27	7=Very satisfied	d.	-				om <b>1</b>	to 7 – where 1=Very dissatisfied	and
		-	number o	n the fo	llowing so	cale)			
	1	2	3	4	5	6	7		
	Very							Very	
	dissatisfie	d						satisfied	

28	What was <b>good</b> about the agreement if anything? (Ask the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and then tick/code the following reasons as appropriate the open ended question record the answer and the open ended question record the answer and the open ended question record the open ended questi	oriate)
	Offender apologised	
	Offender agreeing to attend counselling or a programme to deal with the violence	
	Offender agreeing to do something for others (through work or similar)	
	Measures put in place to keep me and/or my children safe	
	Receiving reparation for damage offender caused	
	Offender didn't go to prison	
	Reaching an agreement	
	Nothing	
	Other (specify)	
29	What was <b>bad</b> about the agreement if anything? (Ask the open ended question record the answer and then tick/code the following reasons as appropriate the control of the c	oriate)
	Offender got off too lightly	
	Being pressured to agree	
	Didn't address keeping me and/or my children safe	
	Didn't reflect my concerns	
	Agreement too vague	
	Not reaching an agreement	
	Too little reparation or too long to pay	
	Nothing	
	Other (specify)	
30	Was the agreement:	
	1= Too harsh 2= About right 3= Too soft	
31	a) Who was to make sure that the agreement was completed?	
	All of us	
	Victim (self)	
	Victim's supporter/family	
	Victim's representative	
	Offender	
	Offender's supporter/family	
	Panel member	
	Facilitator	
	Other (specify- if 'no-one' enter here)	
	b) How was this decided at the conference/meeting?	
Afte	er the conference	
32	How did you feel at the end of the conference/meeting?	
	(Record feelings at the end of the conference/meeting- tease out issues around safety)	
33	How did feel you about (the offender) by the end of the conference/meeting?	

(Record feelings at the end of the conference/meeting)

34		erall, what what sk the open							g if any? the following reasons as approp	riate)
	It wa	as well orga	anised							
	I felt	supported								
		safe								
	I wa	s able to m	eet with	the offen	der					
	I wa	sn't made	to feel tha	at the offe	ending w	as my	fault			
		s treated w			· ·	,				
		agreemen	-							
	It enabled me to deal with the offending									
	Nothing									
	Othe	er (specify)								
35		erall, what what sk the open							f any? the following reasons as approp	riate)
	It wa	asn't well o	rganised							
		n't feel sup	-							
		unsafe								
	l did	n't like me	eting with	the offe	nder					
		s made to	_			fault				
	I felt	bullied/inti	midated		-					
	l wa	sn't treated	with res	pect						
	The	agreemen	t							
	It dic	dn't achieve	e anythin	g/waste	of time					
	Noth	ning								
	Othe	er (specify)								
36	Hov whe	w satisfied ere 1=Very	were you dissatisf	with the	conferer =Very sa	nce/me atisfied	eting over	all? Rate	e on a scale from 1 to 7 –	
			(Circle a	number	on the foli	lowing	scale)			
		1	2	3	4	5	6	7		
		Very							Very	
		dissatisfie	ed					sa	atisfied	
37	We	re you give	n a copy	of the co	onference	e agree	ement? (i.e	. a sumr	mary of what is said in the	
	cor	nference/me	eeting tha	t is sent t	to the Jud	lge)			(Code 1=Yes, 2=No)	
	a)	If Yes do	you thin	k it accur	ately refl	ected v	what was a	greed to	o?	
								(Co	ode 1=Yes, 2=No or 3= Partly)	
38		you know v ference/m		or not (th	e offende	er) com	pleted the	tasks th	at s/he agreed to do at the	
									(Code 1=Yes, 2=No)	
	a)	If Yes ho	w did yo	ı find out	t?					
	b)	If No wou	uld you h	ave liked	to be ke	pt infoi	med?			
	<u>(</u> If	yes to Q38	3 ask Q39	)						

39	a)	Did (the offer	nder) complete the tasks in the agreemer	nt?
				(Code 1=Yes, 2=No or 3= Partly)
	b)	If not comple	eted or completed in part, do you know th	ne reason for this?
				(Code 1=Yes, 2=No)
The	pres	sent		_
40	Are	you pleased y	ou took part in the conference/meeting?	(Code 1=Yes, 2=No or 3= Partly)
41	As a	a result of takir	ng part in the conference/meeting how do	o you feel about the incident now? Do you
	feel		2. No different	2 Battan
	1=	= Worse	2= No different	3= Better
42	As a	a result of the	conference/meeting how safe do you (an	nd your children) feel now?
	1=	: Less safe	2= No different	3= More safe
43	Sin	ce the confere	nce/meeting has the violence and or abu	usive behaviour stopped?
44	Hov	w has (the offe	nder) taken responsibility for his/her beha	(Code 1=Yes, 2=No or 3= Partly) aviour since then?
45		,	(the offender) still needs help with his/he	
		,	( с) с несте негр напис	
				(Code 1=Yes, 2=No or 3= Partly)
46		•	information or support that you would hat conference/meeting, but didn't receive?	ave liked to have got after the
				(Code 1=Yes/2=No)
	a)	If Yes, what	type of information or support was that?	
47		you think resto	prative justice conferences/meetings are	a good way of dealing with family
48			is happened to you again, would you go	to another rj conference/meeting?  (Code 1=Yes/2=No)
				,
49		uld you recomı ence?	mend Restorative Justice conferences/m	neeting to other victims of family
				(Code 1=Yes, 2=No, 3=Partly)
Bac	karo	ound informat	ion	
	_		ask a few questions about you.	
50			ip (s) do you belong to?	
		-		(Tick all that apply,
		European		H
	Mād			H
		noan		H
		ok Island Māori	I	H
		ngan		$\vdash$
	Niu			H
		nese		H
	Indi		T. J.	H
	Oth	er (such as Du	tch, Japanese, Tokelauan) (specify)	

51	How important was it for the conference/meeting to take account of your cultural needs and expectations? Rate on a scale from 1 to 7 – where 1=Not at all important and 7=Very important.									
	CAPCOLATIONS		a numbe				iot at an	important and r-very import	arit.	
	1	2	3	4	5	6	7			
	Not at	all						Very		
	impor	tant			important					
(If t	he response to	Q51 is 4-7	7 ask Q52	)						
<u>52</u>	As a (ethnicit			,						
		feel that the titions? (F					nt of yo	ur cultural needs and		
	<b>b) If No</b> , wh	nat would	have be	en bettei	r?			(Code 1=Yes, 2=No, 3= Partly	<i>(</i> )	
53								cultural needs and important and 7=Very importa	ınt.	
			a number					, , , , , , , , , , , , , , , , , , , ,		
	1	2	3	4	5	6	7			
	Not at	all	-			-		Very		
	impor	tant					imp	ortant		
(If :	the response to	053 is 4-	7 ask 054	1)						
54	As a (		do you f	•						
	•		•		k accou	int of you	r cultura	al needs and expectations? (I	Probe for	
	why and	record co	omments)						. $\square$	
								(Code 1=Yes, 2=No, 3= Partly	/) []	
	b) If No, wh	nat would	have be	en bettei	r? ( <i>Prok</i>	e for why	and red	cord comments)		
55	Do you have a	any childr	en currer	ntly living	y with yo	u?		(O-d-d-)(0.N	, $\Box$	
	1841							(Code 1=Yes or 2=No	<i>′</i>	
56	What is your appropriate)		hip with t	ne perso	on who v	vas violei	nt or abi	usive towards you? (to be ticke	ed as	
	Partner - cur	rent								
	Partner – ex									
	Family – mo									
	Family –dau	_								
	Family –siste Family –aun									
	Family – ath		v)						H	
	Flatmate	o. (opco)	,,							
57	Were you livin	g with (th	ne offende	er) at the	time the	e inciden	t occurre	ed? (Code 1=Yes or 2=No)		
58	What contact	have you	had with	her/him	since th	ne restora	itive just	tice conference/meeting?		
	Living togeth	ner								
	Living separa	ately, but	seeing o	ne anoth	ner					
	Living separa	ately, con	tact only	over arra	angeme	nts for ch	ildren			
	Living separa	•	contact							
	Other (speci	fy)								

59	Did you have a prof	tection order in plac	e at the time of the conference/m	eeting?	
				(Code 1=Yes or 2=No)	
60	Do you have a prot	ection order in place	e currently?		
				(Code 1=Yes or 2=No)	
61	How has your rela	ationship with (the or	ffender) been since the conference	e/meeting?	
62	Which age group of	do you belong to:			
	Under 20 years				
	20-29 years				
	30-39 years				
	40-49 years				
	50-59 years				
	60 or over				
63	a) Are you in pai	id work?		(Code 1=Yes, 2=No)	
	i) If Yes, wha	t is your occupation	?		
	ii) Is this full ti	me or part time? (C	ircle which)		
	b) If you are no	t in paid work, wha	at category best describes you?		
	Beneficiary (e	e.g. sickness or DPE	3)		
	Unemployed				
	Home duties				
	Retired				
	Student				
	Other (elabora	ate)			
Tha	ank you				
	ank you for your part the interview?	icipation in the resea	arch, do you have any comments	or questions about the r	esearch
Fin	nally				
Wo	ould you like a summ	ary of the research?	(Code 1=Yes, 2=No)		
	s will be in approxim dress or to a family m		e – where would be the best plac	e to send this – to your	
		Name	Address		
You	u:				
Far	mily member(s):				
Frie	end(s):				

## **Restorative Justice in Family Violence Cases**

# Questionnaire: Victim not attending conference

held di	uring			(mont	th)		(year) fo	or	(offender)
with re								(offence	
							in your views.		
(NOTE		hroughou	t use:	8=not ap	oplicable	e; 9=don't	know; 99=missi	ing data; 98=did not wa	ant to ansv
Choos	sing to have	a confer	ence/	meeting					
	hose idea wa 1=Yes or 2=N					ce confer	ence/meeting?		
V	/ictim's (self)								
C	Offender's								
C	Offender's lav	vyer							
F	Police								
J	udge								
V	/ictim adviser								
F	RJ provider								
C	Other (specify)	)							
	Vho contacted stice confere					of dealing	with the offend	ing at a restorative	
V	/ictim adviser	•							
F	RJ provider								
	Other (specify)	)							
								ng at a restorative just gs about offender etc)	ice
(Only	ask Q4 if the	offence w	as IPV	<i>'</i> )					
4 W	ere you aske	d if there	was a	a current	protecti	on order i	n place?		
(0	Code 1=Yes o	r 2=No)							
	n a scale fro n5you at the	time it ha	appen				ry high impact,	rate the impact of the	incident
		1 2	3	4 5	6	7			
	ı	No				Very			
	im	pact				high ir	npact		
<b>c</b> \\\			nd tha	aanfaran	/		•		
6 W	ere you invite	ed to atte	na tne	conterer	ice/mee	eting?		(Codo 1-Voo or 2-A	(a)
a)	<b>If Yes,</b> can	ase tel	I me why	you dic	ln't want t	o attend? (Pron	Code 1=Yes or 2=N) npt for safety concerns)	<i>'</i> —	
b)	<b>If No,</b> woul	d you like	e to ha	ave been	invited?	?		(Code 1=Yes or 2=	No)
7 W	ere you aske	d whethe	er the o	conferenc	ce/meet	ing could	go ahead witho	•	(a)
li	f Yes, how did	d vou feel	about	going ahe	ead with	out vou?		(Code 1=Yes or 2=N	10)

8	Following the incident (the offending) did you see or speak to (the offender) before the conference/meeting? (Tease out issues around whether pressured by offender to attend or not to attend)									
			(Code 1=Yes or 2=No	p)						
	If Yes, what happened?									
9	Did someone represent you	or your views at the co	onference/meeting?							
			(Code 1=Yes or 2=No	o)						
	If Yes, who and were you h	appy with this?								
10	Were you told how the conf	erence/meeting went?								
			(Code 1=Yes or 2=No	p)						
14	What do you think the con-	ference/meeting was tr	rying to achieve?							
Re	ad out the contents of the	agreement								
12	Do you know this was wha	t was decided in the ag	greement at the restorative justice meeting?							
			(Code 1=Yes or 2=No	p)						
	a) If Yes who told you?									
	b) If Yes how were you t	told? (specify)								
13	7=Very satisfied.		scale from <b>1 to 7</b> – where 1=Very dissatisfie	d and						
	(Circle a nur	mber on the following sca	ale)							
	1 2 3	4 5	6 7							
	Very		Very							
	dissatisfied		satisfied							
14	What do you feel was good	about the agreement	if anything? (elaborate)							
15	What do you feel was bad a	about the agreement if	anything? (elaborate)							
15 16	What do you feel was <b>bad</b> a Was the agreement:	about the agreement if	anything? (elaborate)							
	•	about the agreement if  2= About right	anything? (elaborate)  3= Too soft							
	Was the agreement:	2= About right	3= Too soft							
16	Was the agreement:  1= Too harsh  Do you know whether or no	2= About right ot (the offender) comple	3= Too soft	) [						
16	Was the agreement:  1= Too harsh  Do you know whether or not  a) If Yes who told you?	2= About right ot (the offender) comple (specify)	3= Too soft eted the agreement?	) <u> </u>						
16	Was the agreement:  1= Too harsh  Do you know whether or no	2= About right ot (the offender) comple (specify)	3= Too soft eted the agreement?	) <u> </u>						
16	Was the agreement:  1= Too harsh  Do you know whether or not  a) If Yes who told you?  b) If Yes how were you to	2= About right ot (the offender) comple (specify) told? (specify)	3= Too soft eted the agreement?	,						
17	Was the agreement:  1= Too harsh  Do you know whether or not  a) If Yes who told you?  b) If Yes how were you to  Compared with how you fee	2= About right ot (the offender) comple (specify) told? (specify)	3= Too soft eted the agreement? (Code 1=Yes or 2=No	,						
17	Was the agreement:  1= Too harsh  Do you know whether or not  a) If Yes who told you?  b) If Yes how were you to Compared with how you feel now?	2= About right ot (the offender) complet (specify) told? (specify) elt about the offending a	3= Too soft eted the agreement?  (Code 1=Yes or 2=Note around the time of the meeting how do you for	,						
16 17 18	Was the agreement:  1= Too harsh  Do you know whether or not  a) If Yes who told you?  b) If Yes how were you to the compared with how you feel now?  1= Worse	2= About right ot (the offender) complet (specify) told? (specify) elt about the offending a	3= Too soft eted the agreement?  (Code 1=Yes or 2=Note around the time of the meeting how do you for	,						
16 17 18	Was the agreement:  1= Too harsh  Do you know whether or not a) If Yes who told you?  b) If Yes how were you to Compared with how you fe now?  1= Worse  How safe do you (and your 1= Less safe	2= About right ot (the offender) complet (specify) told? (specify) elt about the offending at 2= No different or children) feel now? 2= No different	3= Too soft eted the agreement?  (Code 1=Yes or 2=Note around the time of the meeting how do you for	,						
16 17 18	Was the agreement:  1= Too harsh  Do you know whether or not a) If Yes who told you?  b) If Yes how were you to Compared with how you fe now?  1= Worse  How safe do you (and your 1= Less safe)  Is that as a result of (the off)	2= About right ot (the offender) complet (specify) told? (specify) elt about the offending at 2= No different or children) feel now? 2= No different fender) attending the confender	3= Too soft eted the agreement?  (Code 1=Yes or 2=Note around the time of the meeting how do you for 3= Better  3= More safe conference/meeting or something else?	,						
16 17 18 19	Was the agreement:  1= Too harsh  Do you know whether or not a) If Yes who told you?  b) If Yes how were you to Compared with how you fe now?  1= Worse  How safe do you (and your 1= Less safe)  Is that as a result of (the off)	2= About right ot (the offender) complet (specify) told? (specify) elt about the offending at 2= No different or children) feel now? 2= No different fender) attending the confender	3= Too soft eted the agreement?  (Code 1=Yes or 2=Note around the time of the meeting how do you for 3= Better  3= More safe	eel about it						

#### **Background information**

Now we would like to ask a few questions about you.

23	Which ethnic group (s) do you belong to?	( <del>T</del> ')
	M7 European	(Tick all that apply)
	NZ European Māori	H
	Samoan	
	Cook Island Māori	H
	Tongan	H
	Niuean	H
	Chinese	H
	Indian	H
		H
	Other (such as Dutch, Japanese, Tokelauan) (specify)	
24	Do you have any children currently living with you?	(Code 1=Yes or 2=No)
25	What is your relationship with the person who was violent or abusive t appropriate)	owards you? (to be ticked as
	Partner - current	
	Partner – ex	
	Family – mother/father	
	Family –daughter/son	
	Family –sister/brother	
	Family –aunt/uncle	
	Family – other (specify)	
	Flatmate	
26	Were you living with (the offender) at the time the incident occurred?	(Code 1=Yes or 2=No)
27	What contact have you had with her/him since the restorative justice co	nference/meeting?
	Living together	
	Living separately, but seeing one another	
	Living separately, contact only over arrangements for children	
	Living separately no contact	
	Other (specify)	
28	Did you have a protection order in place at the time of the conference/n	neetina?
		(Code 1=Yes or 2=No)
29	Do you have a protection order in place currently?	
	. ,	(Code 1=Yes or 2=No)
30	How has your relationship with (the offender) been since the conferen	ce/meeting?

31	Which age group of	do you belong to:			
	Under 20 years 20-29 years 30-39 years 40-49 years 50-59 years 60 or over				
32	a) Are you in pa	id work?		(Code 1=Yes, 2=No)	
	i) If Yes, wha	t is your occupation	?		
	ii) Is this full t	ime or part time? (C	Circle which)		
	b) If you are no	t in paid work, wha	at category best desc	ribes you?	
	Beneficiary (e	e.g. sickness or DPE	3)		
	Unemployed				
	Home duties				
	Retired				
	Student				
	Other (elabora	ate)			
Tha	nk you				
	nk you for your part ne interview?	icipation in the rese	arch, do you have an	y comments or questions about the research	
Fina	ally				
Wou	ıld you like a summ	ary of the research?	? (Code 1=Yes, 2=No)		
	will be in approxim		ne – where would be t	the best place to send this – to your	
		Name	Address		
You	:				
Fan	nily member(s):				
	nd(s):				
	` '				

# **Restorative Justice in Family Violence Cases**

Questionnaire: Offender

In th	nis interview we would like to	ask you about what ha	appened at the restorative justice conference	ence/meeting				
held	d during	(month)	(year) for you with respect to					
			(offe	nces)				
The	aim is to find out how you felt	about the whole proce	ess.					
	TE: Coding - Throughout use: question)	8=not applicable; 9=d	on't know; 99=missing data; 98=did not wa	ant to answer				
Cho	oosing to have a conference	/meeting						
<b>1</b> (Co	Whose idea was it to have the de 1=Yes or 2=No - to be coded		nference/meeting?					
	Victim's (self)							
	Offender's							
	Offender's lawyer							
	Police							
	Judge							
	Victim adviser							
	RJ provider							
	Other (specify)							
2	Who contacted you first about the possibility of dealing with the offending at a restorative justice conference/meeting? (Tick which)							
	Victim adviser							
	RJ provider							
	Other (specify)							
3	Why did you decide to go to the conference/meeting? (Possible reasons to be ticked only, not asked) (Ask the open ended question record the answer and then tick/code the following reasons as appropriate)							
	To take responsibility for my	offending						
	To meet with/hear from the v	ictim						
	To be able to apologize for w	hat I did						
	To work out an agreement or	n how to deal with the	offence					
	To help myself and/or the vic	tim (circle which)						
	To make amends to the victi	m (by e.g. paying them	money or doing some work)					
	To have more of a say about	what would happen						
	To get closure							
	Friends/family thought it wou	ld be a good idea						
	Cultural reasons							
	My lawyer thought it would b	e a good idea						
	The Judge suggested it							
	I didn't feel like I could refuse	to go						
	Other (elaborate)							

4	Following the incident (the offending) did you see or speak to (the victim) before the conference/meeting? (Tease out issues around whether pressured by offender to attend or not to attend)										
	(Cod	de 1=Yes o es, what ha	r 2=No)				·		·		
Pre	parat	tion									
5	a)		tacted yo	u about	the arra	ngement	s for the	conferer	nce/meeting?		
	b)	How did 1	they cont	act you	?						
6	Wer	e you ask	ed about:	:						(Code 1=Ye	es or 2=No)
	a)	Where th	e confere	ence/me	eting sh	ould be h	eld?				
	b)	When the	confere	nce/me	eting sho	ould be he	eld?				
	c)	Who sho	uld be at	the con	ference/	meeting?					
	d) Whether or not you had any concerns about your safety if you attended?										
	-	ly ask e) if a		•					_		
	e)	Whether	or not the	ere was	a curren	t protecti	on order	in place	9?		
	f)	Whether	or not the	ere were	e bail cor	nditions in	place?				
7	Wer	e you <b>told</b>	l:								
	a)	In plenty	of time a	about wh	nen the o	conferenc	e/meetin	g was to	be held?		
	b)	What wo	uld happe	en at the	e confere	ence/mee	ting?				
	c)	What you	ı would h	ave to c	lo at the	conferen	ce/meeti	ng?			
	d)	Who else	would b	e at the	confere	nce/meeti	ing?				
	e)	If there w	ere any s	safety 'g	round ru	les' for th	e confer	ence/me	eeting?		
	f)	If there w	ere any s	safety 'g	round ru	les' for a	rriving at	and leav	ving the confe	rence/meeting?	
	W	What were you told?									
8	At th	is stage di	d anyone	e give yo	ou any si	uggestion	s of the	sorts of t	things that cou	uld go in the agre	ement?
	•	e 1=Yes or es, who m		n and w	hat were	they?			-		
9		a scale of feel for the				all prepare	ed and 7	=Very w	ell prepared -	how prepared di	d
			(Circle a	number	on the fo	ollowing s	cale)				
		1	2	3	4	5	6	7			
	I	Not at all						Ve	ery well		
	ı	prepared						р	repared		
		(elaborate -	- tease ou	ut issues	around	whether w	ould hav	e liked m	nore informatio	n)	

# The Restorative Justice Conference/Meeting 10. Did someone go the conference/meeting w

10	Did someone go the conference/meeting with you to support you?											
			(Code 1=Yes, 2=Ne	o)								
	á	a) If Yes, who was this (specify)										
	k	b) Was their support useful?	(Code 1=Yes, 2=No or 3= Partl	y)								
	(	Would you have liked more support at the conference	ce/meeting?									
		Was framula 0	(Code 1=Yes, 2=No or 3= Partl	y)								
		f Yes, from who?										
11	What	do you think the conference/meeting was trying to achie	eve?									
12	a) W	as (the victim or victim representative) at the conference/n	-									
	b) I	<b>f No,</b> would you have liked (the victim or representative) to	(Code 1=Yes, 2=No	o)								
13		did you feel when you first went into the conference/mee	eting?									
14	Durin	g the conference/meeting did you:										
	a) [	Feel involved?	(Code 1=Yes, 2=No or 3= Partl	y)								
	<b>b)</b> (	Jnderstand what was going on?	(Code 1=Yes, 2=No or 3= Partl	y)								
	c) F	eel you had the chance to explain why the offence happ	pened?									
			(Code 1=Yes, 2=No or 3=Partl	y)								
	d) l	Have the opportunity to say what you wanted to say?	(Code 1=Yes, 2=No or 3= Partl	y) 🗌								
	<b>e)</b> F	Feel too scared to say what you really felt?	(Code 1=Yes, 2=No or 3= Partl	y)								
15	Do yo	ou think anyone said too much during the conference/me	eting? (Code 1=Yes, 2=Ne	o)								
16	a) I Pal Vice Vice Vice Off Off Durir a) \	f Yes, who? (Don't ask - tick all that apply) nel members tim's support (e.g. family or whanau) tim's representative	(Code 1=Yes, 2=No or 3= Partl									
	b) I	f No or Partly to above - why do you think that was?										
	<b>c)</b> \	Nere you treated fairly?	(Code 1=Yes, 2=No or 3= Partl	y)								
	d) l	f No or Partly to above - why do you think that was?										

17	During the conference/meeting did you						
	a) Apologise for what you did?	(Code 1=Yes, 2=No or 3= Partly)					
	b) If Yes was the apology accepted?	(Code 1=Yes, 2=No or 3= Partly)					
18	Did you understood how (the victim) felt?	(Code 1=Yes, 2=No or 3= Partly)					
19	Did you feel ashamed of what you had done?	(Code 1=Yes, 2=No or 3= Partly)					
20	Were you able to make up for your offending?	(Code 1=Yes, 2=No or 3= Partly)					
21	During the conference/meeting did:  a) The way you were dealt with make you feel like (Be alert to possible cultural difference her and not						
	b) Did people in the conference/meeting speak up	(Code 1=Yes, 2=No or 3= Partly) on your behalf?					
(Code 1=Yes, 2=No or 3= Partilem 22)  (Code 1=Yes, 2=No or 3= Partilem 23)  (Code 1=Yes, 2=No or 3= Partilem 24)  (Code 1=Yes, 2=No or 3= Partilem 25)  (Code 1=Yes, 2=No or 3= Partilem 25)  (Code 1=Yes, 2=No or 3= Partilem 25)							
23	What did you learn about your behaviour and ifts effermeeting/conference?	ect on (the victim and/or family) during the					
24	Did this affect your views on the use of violence or a	abuse? (Code 1=Yes, 2=No or 3= Partly)					
The	e conference agreement						
25	Was an agreement reached?  a) If Not, why not?	(Code 1=Yes, 2=No)					
26	Who decided on the details in the agreement? All of us Victim Victim's supporter/family Victim's representative Offender (self) Offender's supporter/family Panel member Facilitator Other (specify)	(Tick either option 1 <u>OR</u> all that apply)					
27	Can you remember what was in the agreement?						
		(Code 1=Yes, 2=No or 3= Partly)					
28	Did you understand what was in the agreement?	(Code 1=Yes, 2=No or 3= Partly)					
29	Was the agreement what you expected?	(Code 1=Yes, 2=No or 3= Partly)					

30	Were you satisfied with the agreement? Rate on a scale from <b>1 to 7</b> – where 1=Very dissatisfied and 7=Very satisfied.										
	·	(Circle a	number o	n the fo	llowing	scale)					
	1	2	3	4	5	6	7				
	Very							Very			
	dissatisfi	ed					;	satisfied			
31	What was <b>go</b> (Ask the oper						ck/cod	le the following reasons as a	appropriate)		
	Being able to a	apologise									
	Agreeing to at	tend cour	nselling or	a progi	ramme	to deal witl	h the v	riolence			
	It allowed me	to have c	ontact with	n my pa	ırtner aı	nd/or childr	en (cir	rcle which)			
	Agreeing to do work or something similar to help others										
	Being able to	make rep	aration for	damag	ge caus	ed					
	Not going to p	rison									
	Reaching an a	greemen	t								
	Nothing										
	Other (specify)										
32	What was <b>ba</b> (Ask the oper						ck/cod	e the following reasons as a	appropriate)		
	Community wo	ork <i>(numb</i>	er of hours	s too hig	nh etc)						
	Being pressure	ed to agre	ee								
	Agreement too	vague									
	It stopped me		_	y partne	er and/o	or children	(circle	which)			
	Not reaching a	-									
	Too little repar	ation or t	oo little tin	ne to pa	ay				$\vdash$		
	Nothing Other (anality)										
	Other (specify)										
33	Was the agre	ement:									
	1= Too hars	sh	2=	About	right		3=	Too soft			
34	a) Who was	s to make	sure that	the ag	reemen	t was com	pleted	?			
	All of us										
	Victim										
	Victim's suppo	rter/famil	у								
	Victim's repres	sentative									
	Offender (self)										
	Offender's sup	porter/fa	mily								
	Panel member	ſ									
	Facilitator										
	Other (specify-	· if 'no-one	e' enter hei	re)							
	<b>b)</b> How was	s this dec	ided at the	e confe	rence/m	neeting?					

Aft	er the	e conferen	ice							
35	Ho	w did you f	eel at the	e end of the	ne confe	erence/r	neeting?			
	(Re	ecord feelin	gs at the	end of the	e confere	ence/me	eting- teas	e out is	ssues around safety)	
36		w did feel y ecord feelin		•				ference	e/meeting?	
37		erall, what sk the open							ng if any? le the following reasons as approp	riate)
	It wa	as well orga	anised							
	I felt	supported								
	I wa	s able to m	neet with	the victim	1					
	I wa	s treated w	ith respe	ect						
	The	agreemen	t was god	od						
	It en	abled me t	o put thir	ngs right						
	Noth	ning								
	Othe	er (specify)								
38		erall, what sk the open							g if any? le the following reasons as appropi	riate)
	It wasn't well organised									
	I didn't feel supported									
	I didn't like meeting with the victim									
	I was made to feel bad									
	I wasn't treated with respect									Ш
	The agreement									
	It didn't achieve anything/waste of time									
	Noth	ning								
	Othe	er (specify)								
39		w satisfied ere 1=Very						all? R	ate on a scale from 1 to 7 –	
			(Circle a	number	on the fo	ollowing	scale)			
		1	2	3	4	5	6	7		
		Very							Very	
		dissatisfie	ed					:	satisfied	
40	We	re you give	en a copy	of the co	onferenc	ce agree	ement? (i.e	e. a sui	mmary of what is said in the	
	coi	nference/m	eeting tha	t is sent t	o the Ju	dge)			(Code 1=Yes, 2=No)	
	a)	If Yes do	you thin	k it accur	ately re	flected v	what was a	agreed	to?	
								(	(Code 1=Yes, 2=No or 3= Partly)	
41	a)	Have you	u complet	ted the ta	sks tha	t you ha	d to do un	der the	e plan/contract?	
							-		and code 1=Yes, 2=No, 3=Partly))	
	b)	b) If not completed or completed in part, what was the reason for this?								

(Only ask Q42 if the plan/contract included: Com Work, D&A/Anger assessment or counseling of any kind, attendance at a course etc)

42	a)	Your plan	include	u (			.) ala sc	meon	e arrange u	nis ior y	/ou?		
	b)	Did anyor	ne check	that you	u did wh	at you had	d agree	d to to	?				
										(Code	e 1=Yes,	2=No)	
		If Yes who	?							•		,	
		(If Yes to b											
	c)			ork for y	ou? Ra	ite on a so	ale froi	n 1 to	7 – where	1=Very	dissatis	fied and	d 7=Very
			(Circle a	number	on the f	ollowing s	cale)						
		1	2	3	4	5	6	7	,				
		Not at all						Very	well				
The								10.5					
ine	pres	sent											
43	Are	you pleas	ed you to	ook part	in the co	onference	/meetin	g?	(Code 1=)	Yes, 2=1	No or 3=	Partly)	
44	As a	a result of t	aking pa	irt in the	confere	nce/meeti	ng how	do yo	u feel your	offendi	ng now?	Do yo	u feel:
	1=	Worse		2:	= No dif	ferent		3=	= Better				
45	Wo	uld you like	e more s	upport to	help yo	ou deal wi	th your	violent	t or abusive	behav	iour?		
									(Code 1=)	Yes, 2=1	No or 3=	Partly)	
46	ls ti	here any of	her infor	mation o	or suppo	ort that you	ı would	have	liked to hav	e got a	fter the		
	rest	orative jus	tice conf	erence/n	neeting,	but didn't	receive	?					
										(Coo	le 1=Yes/	/2=No)	
	a)	If Yes, W	hat type	of inform	mation o	r eunnort	wae tha	at2					
	aj	11 165, W	riat type	OI IIIIOII	nauon c	n support	was tric	at :					
47		you think re ence?	estorativ	e justice	confere	nces/mee	tings ar	e a go	ood way of o	dealing	with fam	nily	
48	Wo	uld you rec	ommeno	d Restora	ative Jus	stice confe	erences	/meeti	ng to other	victims	of family	y violen	ice?
									(Code 1	1-Vac	2=No, 3=	Dartly)	
									(Code I	1-163,	2-110, 3-	railly)	
49		-		_				-	u to take re t that you n	-		our/	
							(Code	1=Yes	, 2=No, 3= F	Partly or	=Not an	issue)	
										, .			<u> </u>
50		w have you ng different		esponsib	ollity for y	your beha	viour sii	nce the	e conferenc	ce/meet	ing? Wr	nat are	you
51		ve you com ither detect				the confe	ence/m	eeting	)?	(Code	e 1=Yes/	2=No)	
52	If Y	es, did this	sinvolve	any forn	n of fam	ily violend	e?			(Code	e 1=Yes/	2=No)	
Sen	tenc	ing											
53	Hav	e you bee	n senten	ced at co	ourt for t	the offend	ing that	was d	lealt with at	the co	nference	/meetin	ıg?
										(Code	e 1=Yes,	2=No)	
	a)	If Yes wh	at sente	nce did	you get?	?							
	b)	If Yes do when he/				took into	accoun	t the ta	asks you ha	ad to do	under th	ne agre	ement

#### **Background information**

Now we would like to ask a few questions about you.

54	Which ethnic	group (s)	do you l	pelong to	?				
									(Tick all that apply)  ☐☐
	NZ European								H
	Māori								H
	Samoan								H
	Cook Island N	∕lāori							
	Tongan								
	Niuean								
	Chinese								
	Indian								
	Other (such a	s Dutch, Ja	apanese	, Tokelaı	ıan) (spe	ecify)			
55		? Rate or	n a scale	from 1	<b>to 7</b> – w			t of your cultura important and	al needs and 7=Very important.
	1	2	3	4	5	6	7		
	Not at	all						Very	
	import	ant					imp	ortant	
	•						•		
	ne response to (		ask Q56	)					
56	As a (ethnicity	•		,					
		eel that th ions? <i>(Pr</i>					it of you	ur cultural need	is and
	- P			,		,		(Code 1=Yes.	2=No, 3= Partly)
	<b>b) If No</b> , wh	nat would	have be	en bette	r?			•	
57								cultural needs a important and 7	and 7=Very important.
		(Circle a	number	on the fo	llowing	scale)			
	1	2	3	4	5	6	7		
	Not at	all						Very	
	import						imn	ortant	
	import	ant					шр	Ortant	
(If th	e response to (	Q73 is 4-7	ask Q58	<del>)</del> )					
58	As a (	•	-	eel that:					
		eel that th record cor	_	ment too	k accou	ınt of your	cultura	ıl needs and ex	pectations? (Probe for
								(Code 1=Yes,	2=No, 3= Partly)
	b) If No, wh	nat would	have be	en bette	r? (Proi	be for why	and rec	cord comments)	
59	What is your appropriate)	relationsh	nip with t	the perso	on who	was violen	t or abı	usive towards y	ou? (to be ticked as
	Partner - cur	rent							
	Partner – ex								
	Family – mot	her/father							
	Family –daug	ghter/son							

	Family –sister/brother	
	Family –aunt/uncle	
	Family – other (specify)	
	Flatmate	
60	Were you living with (the victim) at the time the incident occurred? (Code 1=Yes or 2=No)	
61	What contact have you had with her/him since the restorative justice conference/meeting?	
	Living together	
	Living separately, but seeing one another	H
	Living separately, but seeing one another  Living separately, contact only over arrangements for children	H
	Living separately no contact	H
		H
	Other (specify)	
62	Had (the victim) taken out a protection order in place at the time of the conference/meeting?	
	(Code 1=Yes or 2=No)	
60	Is there a protection order against you in place currently?	
	(Code 1=Yes or 2=No)	
61	How has your relationship with (the victim) been since the conference/meeting?	
62	Which age group do you belong to:	
02		
	Under 20 years	
	20-29 years	
	30-39 years	
	40-49 years	
	50-59 years	
	60 or over	
63	a) Are you in paid work? (Code 1=Yes, 2=No)	
	i) If Yes, what is your occupation?	
	ii) Is this full time or part time? (Circle which)	
	b) If you are not in paid work, what category best describes you?	
	Beneficiary (e.g. sickness or DPB)	
	Unemployed	
	Home duties	
	Retired	
	Student	
	Other (elaborate)	
Tha	ank you	
Tha	ank you for your participation in the research, do you have any comments or questions about the re the interview?	esearch
	nally	
	ould you like a summary of the research? (Code 1=Yes, 2=No)	
<b>-</b>		

This will be in approximately six months time – where would be the best place to send this – to your address or to a family member or friend?

	Name	Address	
You:			
Family member(s):			
Friend(s):			

# Restorative Justice in Family Violence Cases Key Informant Interview

Area:		Date completed:						
Nam	e:	Pos	sition: (if relevant)					
Orga	nisa	tion: (if relevant)						
	ļ., .		vetice processes in family violence access?					
1 2	Are	general, what are your views on the use of restorative je there any offences/offenders for which you feel the lence cases are particularly	·					
	a)	Appropriate? (Tease out reasons for response i.e. why a	are these appropriate)					
	b)	Inappropriate? (Tease out reasons for response i.e. why	y are these inappropriate)					
3	On	a five point scale where 1=very poorly to 5=very well	I please rate how well you think the					
	(Re	estorative justice) programme is working in dealing with	family violence cases					
	(en	ter the number in the appropriate box or code dk=don't kr	now)					
	-	ase out reasons for the rating)						
	-	What are its strengths, if any?						
	•	What are its weaknesses, if any? (Prompt for safety con	,					
4		a five point scale where <b>1=too soft to 5=too harsh</b> plestorative justice) programme meetings/conferences de						
	(en	ter the number in the appropriate box or code dk=don't kr	now)					
5		a five point scale where <b>1=very ineffective to 5=very</b> stice) programme in terms of its effectiveness in family						
	(en	ter the number in the appropriate box or code dk=don't kr	now)					
	a)	Involving the victim						
	b)	Holding the offender accountable						
	c)	Repairing the harm caused by the offending						
	d)	Addressing the needs of the offender						
	e)	Reducing reoffending						
	f)	Keeping the victim safe during the meeting						
	g)	Addressing the ongoing safety of the victim						
6	pro	a five point scale where 1=very ineffective to 5=very organized in terms of its effectiveness in family violence of the appropriate box or code dk=don't know)						
	a)	The Judge						
	b)	Victims						

	c)	Offenders			
	d)	The community			
7	Jus	a five point scale where <b>1=very ineffective to 5=very effective</b> please rate tice) programme in terms of its effectiveness in meeting the needs of victims a tily violence who are: (enter the number in the appropriate box or code dk=don't know)			
	a)	Māori			
	b)	Pacific Island peoples			
	c)	Other ethnic group(s) (specify which)			
8		there any ways in which the (Restorative Justice) programme's processes (tick who be improved for family violence cases.		Yes	No
	If <b>Y</b> e	es, in what way?:			
9		a five point scale where <b>1=very poor to 5=very good</b> please rate your (or yourd vice's) relationship with <i>(the restorative justice provider)</i> .	organis	ation's	s or
	(ent	ter the number in the appropriate box or code dk=don't know)			
10	Has	s the use of restorative justice in family violence cases had an impact on		Yes	No
	you	or your agency/department/service? (tick which)			
	If Yes	, in what way (i.e. positive or negative)?			
Jud	ges o	only			
J1	th	deciding to refer a particular family violence case to (the provider), can you please e importance of each of these factors in your decision making on a five point scale venot very important and 5=very important?			
	(e	nter the number in the appropriate box or code dk=don't know)		_	7
a)	The v	victims' feelings, views or wishes			
b)	The c	offender showing an interest in or requesting a rj conference/meeting			]
c)	The c	offender admitting guilt and showing remorse			]
d)	Belie	ving that a rj conference/meeting would help the victim and/or the offender			
e)	Belie	ving that feedback from a rj conference/meeting would enable you to impose a more	<b>;</b>		
	mean	ningful sentence			]
f)	All the	e parties were supportive of having a rj conference/meeting			]
J2		every 10 cases of family violence, how often (% or proportion) are you asked to make an rj provider (e.g. by police or defence counsel)	e a refe	erral	
J3		every 10 cases of family violence where you are asked to make a referral to an rj pro v many do you refer.	vider a	bout	
J4		entencing an offender who has attended a rj meeting/conference in a family violenc eement and related material (e.g. rj report) sufficiently detailed for your purposes?	e case	is the	rj
	-		Yes	No	)
	(tick	( which)			]

J5	When does sentencing take place relative to the restorative justice process? (e.g. after the meeting has taken place, on completion of the agreement?)
J6	In family violence cases, when deciding whether or not to accept the conference/meeting agreement at sentencing, can you please rate on a five point scale- where <b>1=never and 5=frequently</b> -how often you would take the following factors into account.
	(enter the number in the appropriate box or code dk=don't know)
a)	The appropriateness of the agreement for the offence
b)	The agreement's reasonableness and practicality
c)	The extent to which victim's needs were being met
d)	Other factors (please elaborate)
J7	Do you take the conference agreement and the extent of the offender's compliance with this into account at sentencing? (Tease out issues relating to situations where this might not take place)
J8	Out of every 10 cases of family violence dealt with by restorative justice processes in what proportion do you accept the conference agreement?
J9	If there is total compliance with the agreement by the offender what would be the likely sentencing outcome? (e.g. conviction and discharge?)
Gen	eric
11	Is there anything else you would like to add about the use of rj in family violence cases
	Thank you for your participation in the research
	ald you like to receive a summary of the research report when the research is finished?  s, tick this box)

# B4 Areas to discuss in family violence cases - Restorative Justice Services Christchurch

#### Areas to discuss with victims of domestic abuse:

- 1. What has offender done to take responsibility before this?
- 2. What has the victim said of the history of domestic violence with this offender?
- 3. Is abuse still going on? (If so, BEWARE)
- **4.** Why do you think this offence happened?
- **5.** How safe do you feel around this person/the offender?
- 6. What evidence do you have for thinking you would be safe at the time of the RJC and afterwards?
- 7. What is the safety plan following the RJC? e.g. victim leaves first; offender is detained until victim is well away.
- 8. What contact have you had with the offender in the last week? two weeks? month?
- **9.** Are there any things you need before? during? after the RJC?

#### Sources of support:

Women's Refuges: Otautahi, West; Stopping Violence Services; Relationship Services.

#### Areas to discuss with offenders in domestic abuse situations:

- 1. Why do you think an RJC would meet your needs?
- 2. How do you think an RJC might help your victim/s?
- 3. What do you hope to achieve from an RJC?
- 4. What do you understand by violence?
- 5. When did you start using violence? What level of violence?
- **6.** Why do you think you should not go to prison?
- 7. What have you done to take responsibility for your violence? (SIGHT evidence of this)